# PARTICIPATION AGREEMENT

among

THE UNITED STATES DEPARTMENT OF ENERGY

and

PLAINS AND EASTERN CLEAN LINE HOLDINGS LLC,
ARKANSAS CLEAN LINE LLC,

PLAINS AND EASTERN CLEAN LINE OKLAHOMA LLC

and certain of their Affiliates (as set forth herein)

dated as of March 25, 2016

# **Table of Contents**

Article I De	efined Terms and Definitions	2
1.1	Defined Terms	2
1.2	Rules of Interpretation	43
Article II P	roject Ownership Structure	44
2.1	Scope of the Clean Line Entities' Rights in Respect of the Project	44
2.2	Ownership of Project Facilities	45
2.3	Rights to Electrical Capacity	46
Article III A	Acquisition of Project Real Estate Rights	47
3.1	Generally	47
3.2	Clean Line Entities' Obligation to Acquire Project Real Estate Rights	47
3.3	DOE's Acquisition of Project Real Estate Rights	48
3.4	Cost Responsibility for Acquisition of Project Real Estate Rights	50
3.5	Amendments and Modifications to Routing and ROW Plan	50
Article IV I	Development, Construction, Operation and Maintenance of the Project	50
4.1	Development, Construction, Operation and Maintenance of the Project Generally	50
4.2	DOE Mitigation Action Plan	53
4.3	Amendments and Modifications to the Project Plans	54
4.4	Construction Contracts and Project Contracts	54
4.5	Interconnection Agreements	54
4.6	Operational Coordination with SPP, MISO and TVA	55
4.7	Maintenance of the Project Facilities	55
4.8	Capital Repairs and Reserve Account	55
4.9	NERC Standards	56
4.10	DOE Cooperation	56

Article V	Coordination Committee	56
5.	1 Coordination Committee	56
Article V	I Conditions Precedent	58
6.	1 Conditions Precedent to Effective Date	58
6.	2 Conditions Precedent to Voluntary Land Acquisitions	60
6.	3 Conditions Precedent to Acquisitions by Condemnation	64
6.	4 Conditions Precedent to Notice to Proceed	67
Article V	II Term, Termination, Events of Default and Remedies	70
7.	1 Term and Termination	70
7.	2 Acquisition Option	71
7.	3 Events of Default	71
7.	4 Remedies Upon Event of Default	74
7.	5 Winding-Up of the Project	76
7.	6 Wind-Up Reserve Account	76
7.	7 Event of Loss	77
7.	8 Survival of Obligations	77
Article V	III Covenants	77
8.	1 Recordkeeping	77
8.	2 DOE's Access Rights, Etc.	77
8.	3 Reporting Requirements	78
8.	4 Authorizations and Approvals	82
8.	5 Insurance	83
8.	6 Payment of Taxes and Other Amounts	83
8.	7 Maintenance of Existence and Property	84
8.	8 Compliance with Applicable Laws	84

	8.9	Diligent Construction of the Project	84
	8.10	Performance of Obligations	85
	8.11	Permitted Liens	85
	8.12	Merger; Bankruptcy; Dissolution; Transfer of Assets	85
	8.13	New Subsidiaries; Partnerships	85
	8.14	Subsidiaries of Holdings	86
	8.15	Other Transactions	86
	8.16	Testing	86
	8.17	Creation and Perfection of Security Interests; Additional Documents; Filings and Recordings	86
	8.18	Additional Project Subsidiaries and Subsidiary Guarantors	87
	8.19	Lobbying Disclosure Requirement	87
	8.20	Improper Use	87
	8.21	Hazardous Substance Management	88
	8.22	Safety Compliance	88
	8.23	Prohibited Persons	89
	8.24	Davis-Bacon Act	89
	8.25	AM Laws, Anti-Corruption Laws Etc	90
	8.26	ACL Indebtedness	90
	8.27	Renewable Energy Transmission	90
Articl	e IX Gu	arantee	90
	9.1	Guarantee of the Obligations	90
	9.2	Contribution by Subsidiary Guarantors	91
	9.3	Payment by Subsidiary Guarantors	91
	9.4	Guarantee Absolute	91
	9.5	Liability of Guarantors Absolute; Waivers	91

	9.6	Continuing Guarantee; Assignments	93
	9.7	Obligations and Rights of Subsidiary Guarantors	93
Article	X For	ce Majeure	93
	10.1	Force Majeure	93
Article	XI Co	st and Expense Funding, Advance Funding, Indemnity and Collateral	93
	11.1	Cost and Expense Funding	93
	11.2	Participation Amount	94
	11.3	Advance Cost Funding	94
	11.4	Indemnification	95
	11.5	Performance Support	98
	11.6	Collateral	99
	11.7	Intercreditor Agreement	99
	11.8	Limitation of Liability to DOE	100
	11.9	Consequential Damages	100
	11.10	Release Provision	101
Article	XII Re	epresentations and Warranties	102
	12.1	Representations and Warranties of the Clean Line Parties	102
	12.2	Survival	112
	12.3	Disclosure Schedule	112
Article	e XIII M	Miscellaneous Terms and Provisions	112
	13.1	Notices; Consents; Approvals	112
	13.2	Further Assurances	113
	13.3	Amendment; Waiver	114
	13.4	Lender-and Financing-Related Provisions	114
	13.5	Project Financing Document Provisions	114

13.6	Grant of Security Interest	114
13.7	DOE Review Standard	115
13.8	DOE Delegation	115
13.9	Assignments	115
13.10	Successors and Assigns	115
13.11	Joint and Several Obligations	115
13.12	Right to Intervene	115
13.13	Publication; Public Statements	115
13.14	Third Parties	116
13.15	Independent Contractor Status	116
13.16	TN and TX Facilities	116
13.17	Governing Law	116
13.18	Jurisdiction	116
13.19	Dispute Resolution	117
13.20	Waiver of Jury Trial	117
13.21	Negotiation and Documentation of this Agreement	117
13.22	Severability	117
13.23	Counterparts	118
13.24	Entire Agreement	118
13.25	Time is of the Essence	118
13.26	Confidentiality of Information	118
13 27	Non-Exclusivity	118

# LIST OF SCHEDULES

Schedule 1: Clean Line Entities Real Estate Rights Acquisition Procedures

Schedule 2: Provisions Required to be Incorporated into Project Financing Documents

Schedule 3: Existing Indebtedness

Schedule 4: Local Government Contribution Payments

Schedule 5: Capitalization

Schedule 6: Officer's Certificate Regarding DOE Delegated Real Estate Rights

Schedule 7: Officer's Certificate Regarding Acquisitions by Condemnation

Schedule 8: Reserved

Schedule 9: Litigation

Schedule 10: Environmental Matters

Schedule 11: Governmental Approvals

Schedule 12: Code of Conduct for Acquisitions of Project Real Estate Rights

Schedule 13: Permitted Project Investments Representations and Warranties and

Covenants

Schedule 14: Conditions in Acceptable Transmission Services Agreements and

Acceptable Permitted Project Investment Commitments

Schedule 15: Davis-Bacon Requirements

Schedule 16: Required Approvals

Schedule 17: Clean Line Uniform Act Execution Plan

#### PARTICIPATION AGREEMENT

This PARTICIPATION AGREEMENT (this "<u>Agreement</u>"), dated as of March 25, 2016, is entered into by and among the UNITED STATES DEPARTMENT OF ENERGY (the "<u>Department</u>" or "<u>DOE</u>"), PLAINS AND EASTERN CLEAN LINE HOLDINGS LLC, a limited liability company organized under the laws of the State of Delaware ("<u>Holdings</u>"), ARKANSAS CLEAN LINE LLC, a limited liability company organized under the laws of the State of Delaware ("<u>ACL</u>"), PLAINS AND EASTERN CLEAN LINE OKLAHOMA LLC, a limited liability company organized under the laws of the State of Oklahoma ("<u>PECL OK</u>"), OKLAHOMA LAND ACQUISITION COMPANY LLC ("<u>OLA</u>"), a limited liability company organized under the laws of the State of Delaware, and, solely to the extent that any of the provisions set forth herein apply to the Clean Line Parties (as opposed to the Clean Line Entities, Holdings or any of the Project Subsidiaries), PLAINS AND EASTERN CLEAN LINE LLC, a limited liability company organized under the laws of the State of Arkansas ("<u>PECL</u>"). Capitalized terms used herein shall have the meanings set forth for such term in Section 1.1 of this Agreement.

#### **RECITALS**

**WHEREAS**, pursuant to Section 1222 of the Energy Policy Act of 2005 ("Section 1222"), the Secretary of the Department, acting through the Southwestern Power Administration ("SWPA"), has the authority to design, develop, construct, operate, maintain, own or otherwise participate with other Persons in designing, developing, constructing, operating, maintaining or owning new electric power transmission facilities and related facilities within any state in which SWPA operates and to accept third party funding for these purposes.

**WHEREAS**, pursuant to its authorities under, and in reliance upon, Section 1222, the Department is participating with the Clean Line Entities in the design, development, construction, operation, maintenance and ownership, as applicable, of approximately 705 miles of +/-600 kilovolt overhead, high voltage direct current ("HVDC") electric transmission facilities and related facilities with the capacity to deliver approximately 4,000 megawatts ("MW") (net) from renewable energy generation facilities located in the Oklahoma Panhandle and Texas Panhandle regions to the eastern state-line of Arkansas near the Mississippi River (the "Arkansas Connection Point") (collectively, the "Project").

WHEREAS, the Project shall include: (a) an AC/DC converter station and related facilities located in Texas County, Oklahoma (the "Converter Station Facility"), (b) an AC collection system of up to six AC transmission lines located in the Oklahoma Panhandle (the "AC Collection System"), (c) an intermediate AC/DC converter station and related facilities located in Pope County, Arkansas (the "Intermediate Converter Station") and (d) all transmission lines (including all structures and wires and related components) running from the Converter Station Facility to the Arkansas Connection Point and the AC transmission lines interconnecting the Converter Station Facility to the transmission system under the operational control of the Southwest Power Pool, Inc. ("SPP") and the Intermediate Converter Station to the transmission system under the operational control of the Midcontinent Independent System Operator ("MISO") (collectively, the "Transmission Line Facilities" and together with the Converter Station Facility, the AC Collection System, the Intermediate Converter Station and related

facilities, the "<u>Project Facilities</u>"). The Project Facilities located in Oklahoma are hereinafter referred to as the "<u>OK Facilities</u>" and the Project Facilities located in Arkansas are hereinafter referred to as the "<u>AR Facilities</u>." The Project Facilities include all temporary and permanent structures, wires and related components in respect of the Project.

**WHEREAS**, the Clean Line Parties are separately developing transmission and related facilities in Tennessee that will interconnect with the Project at the Arkansas Connection Point and may develop facilities in the Texas Panhandle that would interconnect with the Project at the Texas-Oklahoma state-line.

WHEREAS, on June 10, 2010, the Department published a *Request for Proposals for New or Upgraded Transmission Line Projects Under Section 1222 of the Energy Policy Act of 2005* (75 Fed. Reg. 32940) (the "RFP") in the Federal Register. In July 2010, Clean Line Energy Partners LLC ("CLEP"), a limited liability company organized under the laws of the State of Delaware, submitted a proposal relating to the Project in response to the RFP and submitted a revised proposal in August 2011. In April 2012, the Department determined that CLEP's proposal was responsive to the RFP, and subsequently, on September 20, 2012, the Department and SWPA entered into an Advance Funding and Development Agreement (the "AFDA") with CLEP, PECL and PECL OK, which, among other things, provides for advance funding by CLEP, PECL and PECL OK to the Department and SWPA to commence necessary environmental reviews and other due diligence activities in order to assess whether the Project meets the criteria specified by Section 1222.

**NOW, THEREFORE**, in consideration of the mutual covenants contained herein and other valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, the Parties hereto, intending to be legally bound, do hereby agree as follows:

# ARTICLE I DEFINED TERMS AND DEFINITIONS

1.1 <u>Defined Terms</u>. For purposes of this Agreement, the following words and expressions when initially capitalized shall have the meaning assigned to them below.

"Abandonment" means that (a) prior to the occurrence of Project Completion, development activities and construction activities (if construction has begun) in respect of the Project shall have ceased for any reason, other than as a result of the occurrence of Force Majeure or a Governmental Order that requires cessation of activities until compliance with the Governmental Order is achieved and the Clean Line Entity subject to such order is diligently pursuing compliance, for a period of ninety (90) consecutive days, (b) from and after the occurrence of Project Completion, the Project shall have ceased to operate, other than as a result of the occurrence of Force Majeure, for a period of ninety (90) consecutive days, or (c) at any time any Clean Line Entity shall have publicly declared that it intends not to continue with the development, design, engineering, construction, financing, ownership, operation, maintenance and management of the Project for any reason other than the occurrence of Force Majeure.

"AC" means alternating current.

"AC Collection System" has the meaning set forth in the recitals.

"Acceptable Counterparty" means any Person that either (a) owns and operates a renewable power generating facility located in any, all or some of the Oklahoma and Texas Panhandle regions and the State of Arkansas or (b) is a purchaser of renewable energy that is being delivered by the Project.

"Acceptable Form" means, with respect to any letter of credit, Guarantee, commitment, undertaking or other Contractual Obligation, that such letter of credit, Guarantee, commitment, undertaking or other Contractual Obligation is in form and substance reasonably acceptable or satisfactory to DOE; provided that it would be unreasonable for DOE not to accept or be satisfied with any such letter of credit, Guarantee, commitment, undertaking or other Contractual Obligation for purposes of satisfying any requirement, condition or other matter set forth in this Agreement to the extent such letter of credit, Guarantee, commitment, undertaking or other Contractual Obligation (a) is on customary market terms (to be determined taking into account the purpose for which such letter of credit, Guarantee, commitment, undertaking or other Contractual Obligation is being delivered or required under the terms of this Agreement), (b) does not contain any conditions precedent to any Person's obligations thereunder that could not reasonably be anticipated to be satisfied on a timely basis, (c) in the case of any Project Equity Commitment or Project Financing Commitment, such commitment is not predicated on the satisfaction of any general due diligence condition precedent, and (d) does not contain any unusual termination provisions (whether taking the form of an event of default, termination event, an event that gives rise to a right of acceleration or in any other form that gives rise to any of the foregoing) that could reasonably be anticipated to give rise to a termination of such letter of credit, Guarantee, commitment, undertaking of other Contractual Obligation or an acceleration of any applicable Clean Line Parties' obligations thereunder (excluding a termination based on a reasonable sunset date or date certain that in either case is consistent with the Project Schedule).

"Acceptable Guarantee" means an unconditional, irrevocable, direct-pay guarantee (a) that (i) is denominated in Dollars, (ii) provides that DOE is the beneficiary thereof, (iii) is issued by an Acceptable Support Provider, (iv) requires the issuer thereof to have waived all rights to make any claim against any Clean Line Party, DOE or the Collateral, whether for costs of maintaining the guarantee or reimbursement of amounts paid under the guarantee, or otherwise, and none of the Clean Line Parties shall be required to pay any fee to such issuer in respect of the issuance of such guarantee, in each case, except out of cash available for equity distributions or dividends to holders of Equity Interests in Holdings, (v) entitles DOE to make a demand for payment thereunder as contemplated by this Agreement, including under the circumstances contemplated by Section 11.5(a) and (vi) is otherwise in an Acceptable Form (including with respect to representations, covenants and requirements relating to posting of collateral support in instances where the issuer thereof ceases to be an Acceptable Support Provider) and (b) as to which DOE has received (i) such financial statements in respect of the issuer thereof as requested by DOE, (ii) customary legal opinions with respect to capacity, authority and enforceability of such guarantee and as to such other matters as reasonably requested by DOE from legal counsel acceptable to DOE, and (iii) corporate documents, resolutions, copies of any necessary consents and approvals and customary certificates by and in respect of the issuer thereof as reasonably required by DOE.

"Acceptable Letter of Credit" means an unconditional, irrevocable, direct-pay letter of credit that (a) is denominated in Dollars, (b) is issued in favor of DOE by an Acceptable Support Provider, (c) meets each of the following requirements and (d) is otherwise in Acceptable Form:

- (i) the initial expiration date thereof shall be at least twelve (12) months beyond the date of issuance, and shall automatically renew upon its expiration (which renewal period shall be for at least twelve (12) months) unless, at least forty-five (45) days prior to any such expiration, the issuer thereof shall provide DOE with a notice of non-renewal of such letter of credit;
- (ii) upon any failure to renew such letter of credit at least thirty (30) days prior to such expiration date, or if the issuer of such letter of credit shall cease to be an Acceptable Support Provider, the entire face amount thereof shall be drawable by DOE;
- (iii) such letter of credit shall be drawable by DOE as contemplated by this Agreement, including under the circumstances contemplated by Section 11.5(a);
- (iv) no Contractual Obligation executed or delivered in connection with such letter of credit shall provide the issuer thereof or any other Person with any claim against any Clean Line Party, DOE, or the Collateral, whether for costs of maintenance, reimbursement of amounts drawn under such letter of credit or otherwise (except if such letter of credit is provided as part of the Project Financing pursuant to the Project Financing Documents);
- (v) such letter of credit shall be payable immediately, conditioned only on written presentment from DOE to the issuer thereof of a sight draft drawn on such letter of credit and a certificate stating that DOE has the right to draw under such letter of credit in the amount of the sight draft without the requirement to present the original letter of credit; and
- (vi) such letter of credit shall allow for multiple draws.

"Acceptable Permitted Project Investment Commitment" means a binding Contractual Obligation entered into by one or more of the Clean Line Entities with an Acceptable Counterparty pursuant to which such Acceptable Counterparty has committed to make a Permitted Project Investment for fair market value in order to have the right to use a portion of the Electrical Capacity for the transmission of power from renewable energy sources related to such Acceptable Counterparty's purchase or sale of renewable energy or to enter into Acceptable Transmission Services Agreements with other third parties; provided that the obligation of such Acceptable Counterparty to make the applicable Permitted Project Investment shall only be subject to the satisfaction of the conditions precedent set forth on Part A of Schedule 14 hereto or shall otherwise be in an Acceptable Form.

- "Acceptable Support Provider" means a Person that meets the following criteria:
- in the case of an Acceptable Letter of Credit or letter of credit provided in connection with any Project Equity Commitment, such Person (i) is either (A) a bank with a branch or representative office in New York, New York and is organized under or licensed as a branch or agency under the laws of the United States or any State thereof or (B) a corporation or limited liability company that is organized under the laws of the United States or any State thereof, (ii) has outstanding unguaranteed and unsecured long-term Indebtedness for Borrowed Money that is rated "A-" or better by S&P and "A3" or better by Moody's (with neither such rating being on negative watch) and (iii) has a combined capital and surplus of at least \$500,000,000;
- (b) in the case of an Acceptable Guarantee, such Person (i) is a corporation or limited liability company that is organized under the laws of the United States or any State thereof, (ii) has outstanding unguaranteed and unsecured long-term Indebtedness for Borrowed Money that is rated at least "BBB" by S&P and "Baa2" by Moody's (with neither such rating being on negative watch) and (iii) has a combined capital and surplus of at least \$250,000,000; and
- in the case of a Project Equity Commitment, such Person is a corporation, limited partnership or limited liability company that is organized under the laws of the United States or any State thereof and such Person (i) has outstanding unguaranteed and unsecured long-term Indebtedness for Borrowed Money that is rated at least "BBB" by S&P and "Baa2" by Moody's (with neither such rating being on negative watch) or (ii) has a combined capital and surplus of at least \$250,000,000.

"Acceptable Transmission Services Agreement" means a firm committed Transmission Services Agreement with an Acceptable Counterparty that (a) satisfies the following criteria; (i) the term of such Transmission Services Agreement is for not less than five years, (ii) the obligation of the Acceptable Counterparty thereto is only subject to the satisfaction of the conditions precedent set forth in Part A of Schedule 14 hereto, (iii) such Transmission Services Agreement contains only rights of termination on the part of the Acceptable Counterparty set forth in Part B of Schedule 14 hereto, and (iv) such Transmission Services Agreement cannot be terminated for convenience, whether by payment of a penalty or otherwise, or (b) is otherwise in an Acceptable Form.

"Account Collateral" has the meaning set forth in Section 11.6(a)(i).

"ACL" has the meaning set forth in the preamble.

"Acquisition by Condemnation" means any acquisition of Project Real Estate Rights by DOE through its powers of eminent domain or by condemnation.

"Acquisition Option" has the meaning set forth in Section 7.2(a).

"Action" means any (a) action, suit or proceeding of or before any Governmental Authority, (b) investigation by a Governmental Authority or (c) arbitral proceeding.

"<u>Active Participation</u>" means, in respect of any Key Person, that such Key Person (a) prior to the issuance of the Notice to Proceed, devotes substantially all of his business time and attention to the Clean Line Entities and CLEP (and its Subsidiaries) and the conduct of their business and (b) after the issuance of the Notice to Proceed and until Project Completion, such Key Person remains involved with the Clean Line Entities and the conduct of their business.

"Adverse DOE Impact" means a material adverse effect on (a) DOE's rights and remedies under the Transaction Documents, including each Covered Party's right to be indemnified for Covered Liabilities, (b) each Clean Line Party's ability to perform in a timely manner its obligations under this Agreement or any other Transaction Document, including such Clean Line Party's obligation to pay Covered Costs, (c) DOE's express third party beneficiary rights under any Material Project Contract, (d) the construction or operation of the Project, (e) DOE's obligations or liabilities in respect of any DOE Delegated Real Estate Rights or the AR Facilities, (f) the validity or enforceability of any material provision of this Agreement or any other Transaction Document, and (g) the validity or enforceability of the Performance Support or the validity, enforceability or priority of DOE's security interests in the Collateral and the continued effectiveness and enforceability of the Security Documents.

"Advance Funding Account" has the meaning set forth in Section 11.3(a).

"Advanced Funding Contingency Amount" means, as of any given date, a contingency amount equal to 10% of all Covered Costs estimated by DOE to be due and payable by DOE in the three (3) month period immediately succeeding such date of determination.

"AFDA" has the meaning set forth in the recitals.

"Affiliate" means with respect to any Person, any other Person that directly or indirectly Controls, or is under common Control with, or is Controlled by, such Person and, if such Person is an individual, any member of the immediate family of such individual and any trust whose principal beneficiary is such individual or one or more members of such immediate family and any Person who is Controlled by any such member or trust.

"Affiliated Lenders" means any Person that either (a) is an Affiliate of any Clean Line Party or (b) owns, directly or indirectly, more than five percent (5%) of the Equity Interests in any Clean Line Party.

"Aggregate Payments" means, with respect to a Contributing Subsidiary Guarantor as of any date of determination, an amount equal to (a) the aggregate amount of all payments and distributions made on or before such date by such Contributing Subsidiary Guarantor in respect of the Guarantee under Article IX, *minus* (b) the aggregate amount of all payments received on or before such date by such Contributing Subsidiary Guarantor from the other Contributing Subsidiary Guarantors as contributions under Section 9.2.

"Agreed Rate" means the "Prime rate" for the "U.S." as published in the "Money Rates" table of The Wall Street Journal from time to time.

"Agreement" has the meaning set forth in the preamble.

"AM Laws" means, with respect to any Person, all Applicable Laws concerning or relating to anti-money laundering.

"Anti-Corruption Laws" means, with respect to any Person, all Applicable Laws concerning or relating to bribery or corruption, including, the Foreign Corrupt Practices Act of 1977 (Pub. L. No. 95 213, §§101-104).

"Applicable Amount" means (a) from and after the Commencement Date but before DOE has issued the Notice to Proceed, \$5,000,000 and (b) from and after the issuance of the Notice to Proceed, \$50,000,000 *minus* the balance of any funds then on deposit in, or credited to, the Wind-Up Reserve Account; <u>provided</u> that the Applicable Amount, from and after the issuance of the Notice to Proceed, shall in no event be less than \$10,000,000 at any time.

"Applicable Laws" means, with respect to any Person, any constitution, statute, law, rule, regulation, code, ordinance, treaty, judgment, order or any published directive, guideline, requirement, other governmental rule or restriction or Governmental Order which has the force of law, by or from a court, arbitrator or other of a Governmental Authority having jurisdiction over such Person or any of its Properties, whether in effect as of the date hereof or as of any date hereafter and including any applicable Environmental Laws.

"APSC" means the Arkansas Public Service Commission.

"APSC 2011 Order" has the meaning set forth in Section 12.1(t).

"AR Facilities" has the meaning set forth in the recitals.

"Arkansas Connection Point" has the meaning set forth in the recitals.

"Authorized Officer" means, (a) with respect to any Person that is a corporation, the chairman, chief executive officer, president, vice president, assistant vice president, treasurer, assistant treasurer, or any other financial officer of such Person, (b) with respect to any Person that is a partnership, each general partner of such Person or the chairman, chief executive officer, president, a vice president, an assistant vice president, treasurer, an assistant treasurer or any other financial officer of a general partner of such Person or (c) with respect to any Person that is a limited liability company, the manager, managing partner or duly appointed officer of such Person, the individuals authorized to represent such Person pursuant to the Organizational Documents of such Person, or the chairman, chief executive officer, president, vice president, assistant vice president, treasurer, assistant treasurer or any other financial officer of the manager or managing member of such Person.

"Bankruptcy Code" means Title 11 of the United States Code entitled "Bankruptcy", as amended.

"<u>Bankruptcy Law</u>" means the Bankruptcy Code and any similar federal, state or foreign law for the relief of debtors, conservatorship, bankruptcy, general assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization or similar debtor

relief laws of the United States or other applicable jurisdictions from time to time in effect, and any similar federal, state or foreign law for the relief of debtors affecting the rights of creditors generally.

"Base Amount" means, as of any date of determination, the sum of: (a) the amount of all Covered Costs estimated by DOE to be due and payable by DOE in the three (3) month period immediately succeeding such date of determination *plus* (b) the sum of any future amounts payable from time to time by DOE pursuant to any Contractual Obligation entered into by DOE in connection with the Project (including any Real Estate Rights Agreements) (but subject in all cases to Section 11.3(g)) regardless of the time at which such amount is payable by DOE (for the avoidance of doubt, such amount shall include any amounts payable by DOE under any such Contractual Obligation for any future years occurring during the term of such Contractual Obligation).

"Base Case Projections" has the meaning set forth in Section 6.1(h).

"Base Contingency Amount" has the meaning set forth in the definition of "Contingency Amount".

"<u>Biological Opinion</u>" means the U.S. Fish and Wildlife's biological opinion issued on November 20, 2015 pursuant to Section 7 of the Endangered Species Act regarding the Project and Other Facilities, as amended or updated from time to time as required.

"Business Day" means any day other than a Saturday, Sunday or any other day on which DOE is not open for business.

"<u>Capital Expenditures</u>" means, with respect to any Person for any period, any expenditure in respect of the purchase or other acquisition of any fixed or capital asset (excluding normal replacements and maintenance which are properly charged to current operations).

"<u>Capital Lease</u>" means, for any Person, any lease of (or other agreement conveying the right to use) any Property of such Person that would be required, in accordance with GAAP, to be capitalized and accounted for as a capital lease on a balance sheet of such Person.

"Capital Lease Obligations" means the obligations of any Person under any Capital Lease.

"Capital Repairs" means (a) any and all work necessary, desirable or appropriate to repair, restore, refurbish or replace any equipment, structure or any other component of the Project Facilities (or any portion thereof) after Project Completion, including any such work necessitated by (i) any defect or deficiency, (ii) physical or functional obsolescence of the Project Facilities, as adjudged by Prudent Utility Practices, or (iii) modifications required by any Applicable Law or dictated by the observance of Prudent Utility Practices or (b) the substitution, replacement, enlargement or improvement of any structure, facility, equipment, Property, land or land rights constituting part of the Project Facilities.

"Capital Repairs Reserve Account" has the meaning set forth in Section 4.8(b).

### "Change of Control" means:

- (a) prior to the occurrence of a Qualified IPO in respect of the Clean Line Entities, the occurrence of any of the following events or circumstances:
  - (i) CLEP or the Clean Line Entities are no longer Controlled by a Permitted Holder, or
  - (ii) until Project Completion, any Key Person ceases Active Participation (other than as a result of sickness, death, incapacity or retirement) and within sixty (60) days either: (A) is not replaced by an individual that is acceptable to DOE, such consent not to be unreasonably withheld or delayed (it being understood and agreed that it would be unreasonable to withhold consent to the extent the replacement individual has qualifications and experience substantially similar to or better than the experience of the individual being replaced and such replacement individual is not a Prohibited Person) or (B) recommences Active Participation; or
- (b) from and after the occurrence of a Qualified IPO, the occurrence of any of the following events of circumstances:
  - (i) the Clean Line Entities are no longer Controlled by the IPO Entity, or
  - any "person" or "group" (as such terms are used in Sections 13(d) and (ii) 14(d) of the Securities Exchange Act of 1934, but excluding any employee benefit plan of such person or its subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) other than the Permitted Holders becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Securities Exchange Act of 1934, except that a person or group shall be deemed to have "beneficial ownership" of all securities that such person or group has the right to acquire, whether such right is exercisable immediately or only after the passage of time (such right, an "option right")), directly or indirectly, of 25% or more of the equity securities of the IPO Entity entitled to vote for members of the board of directors or equivalent governing body of the IPO Entity on a fully-diluted basis (and taking into account all such securities that such "person" or "group" has the right to acquire pursuant to any option right), or
  - (iii) during any period of twelve (12) consecutive months, a majority of the members of the board of directors or other equivalent governing body of the IPO Entity cease to be composed of individuals (A) who were members of that board or equivalent governing body on the first day of such period, (B) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in clause (A) above constituting at the time of such election or nomination at least a

majority of that board or equivalent governing body or (C) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in clauses (A) and (B) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body, or

- (iv) the passage of thirty (30) days from the date upon which any Person or two (2) or more Persons acting in concert shall have acquired by contract or otherwise, or shall have entered into a contract or arrangement that, upon consummation thereof, will result in its or their acquisition of the power to exercise, directly or indirectly, Control over the IPO Entity, or Control over the equity securities of the IPO Entity entitled to vote for members of the board of directors or equivalent governing body of the IPO Entity on a fully-diluted basis (and taking into account all such securities that such Person or Persons have the right to acquire pursuant to any option right) representing 25% or more of the combined voting power of such securities, or
- (v) any "change of control" or any comparable term under, and as defined in, any Contractual Obligation to which the IPO Entity is a party shall have occurred; or
- (c) at any time, Holdings shall cease to own, directly or indirectly, 100% of the beneficial or of record Equity Interests in each of PECL, ACL and PECL OK; or
- (d) at any time, CLEP shall fail to own, directly or indirectly, 100% of the beneficial or of record Equity Interests in Holdings.

"Change of Law" means any change in any Applicable Law or the application or requirements thereof of any Governmental Authority issued after the Effective Date.

"Claiming Party" has the meaning set forth in Section 10.1.

"Clean Line Document" means, at any given time, this Agreement, any other Transaction Document in effect at such time and any Material Project Contract in effect at such time.

"Clean Line Entity" means Holdings and each of its Subsidiaries (other than PECL and any PECL Subsidiary).

"Clean Line Guarantor" means any Person that is a guarantor under any Acceptable Guarantee.

"Clean Line Material Adverse Effect" means, as of any date of determination, a material and adverse effect on (a) the Project, (b) the ability of the Clean Line Parties, taken as a whole, to perform their material obligations in a timely manner under any Transaction Document, (c) the business, Properties, operations or financial condition of the Clean Line Parties, taken as a whole, (d) the validity or enforceability of any material provision of any Transaction Document, (e) any material right or remedy of DOE under the Transaction Documents or (f) the Lien of

DOE on any of the Collateral under any Security Document (except as contemplated under this Agreement).

"Clean Line Obligor" means each of the Clean Line Parties and any Clean Line Guarantor.

"Clean Line Party" means Holdings and each of its Subsidiaries (including PECL and any PECL Subsidiary).

"CLEP" has the meaning set forth in the recitals.

"Collateral" means any Equity Interests in ACL or any other Property (whether tangible or intangible) of the Clean Line Entities whether now existing or hereinafter acquired that are subject to or are intended to be or become subject to the Lien granted to DOE pursuant to the Security Documents as required under the terms of this Agreement.

"Commencement Date" means the first date on which the conditions precedent set forth in Section 6.2 shall have been satisfied.

"Completion Conditions" means the satisfaction of each of the following conditions:

- (a) the Project has commenced commercial operation and has satisfied the requirements for "substantial completion" (or term of similar import) as defined in and in accordance with all Construction Contracts and the initial Electrical Capacity (as specified in the definition thereof) of the Project has been certified by an Independent Engineer;
- (b) the Project has been safely and reliably energized and energy may be delivered across the Project Facilities to SPP's, MISO's and TVA's transmission systems in accordance with the Interconnection Agreements;
- (c) the Project has been constructed and become available for normal, safe and continuous operation in compliance in all material respects with the requirements and specifications of the Project Plans, the Material Project Contracts, Applicable Law, any Required Approvals and Prudent Utility Practices;
- (d) all payments required to be made to each Contractor under each Material Construction Contract have been paid in full in cash, other than any payments that are subject to Contest;
- (e) a final invoice has been issued by (or on behalf of) the Clean Line Entities to any applicable Contractor as to any liquidated damages payable under any Construction Contract;
- (f) the Clean Line Entities have delivered to DOE executed acknowledgments of payments and releases of Liens from the Contractors under each Material Construction Contract and each Major Subcontractor thereunder for all work, services and materials, including equipment and fixtures of any kind, done,

- previously performed or furnished for the construction of the Project, other than to the extent payment thereof is subject to Contest;
- (g) all Required Approvals that are necessary or required to have been obtained as of Project Completion under Applicable Law or any material Contractual Obligation applicable to, or binding on, any Clean Line Entity (i) have been obtained, (ii) are in full force and effect and (iii) all conditions precedent to the effectiveness of any such Required Approval have been satisfied;
- (h) all Required Insurance with respect to the operational phase of the Project is in place, in good standing and in full force and effect without default and all premiums due thereon have been paid in full, and DOE has received evidence thereof;
- (i) all Project Real Estate Rights, utilities and other services, means of transportation, facilities, equipment, other rights and materials or supplies necessary for the operation of the Project in accordance (in all material respects) with Prudent Utility Practices, Applicable Law, Required Approvals, the Transaction Documents and the Material Project Contracts and as otherwise contemplated by the Project Plans are available to the Clean Line Entities under the terms of the Material Project Contracts then in effect or otherwise available on commercially reasonable terms materially consistent with the then applicable Project Budget;
- (j) the Capital Repairs Reserve Account has been established and funded in full; and
- (k) no Default or Event of Default has occurred and is continuing.

"Construction Contract" means any design, construction, procurement, supply or other Contractual Obligation executed in connection with the construction, procurement, installation, or improvement of land, buildings, equipment, or facilities necessary or desirable for the Project.

"Construction Contractor" means any Contractor under any Construction Contract.

"Construction Costs" means any and all Project Costs anticipated to be incurred by either of DOE or any Clean Line Entity in connection with the design, development, engineering, construction, administration, management, operation, financing and ownership of the Project through the occurrence of Project Completion.

"Construction Progress Report" means a construction progress report prepared quarterly by the Clean Line Entities, which shall include: (a) a reasonably detailed assessment of the progress of construction to date in comparison with the Project Plans then in effect for such quarterly period (along with an explanation of material delays, if any) and the expected progress of construction; (b) contingencies used or reasonably expected to be used to pay Construction Costs; (c) any events that have occurred or are reasonably expected to occur that would materially affect the construction schedule; (d) a description and explanation of any Events of Loss that have occurred and (e) material disputes or Actions between any Clean Line Entity and any other Person.

"Contest" means, with respect to any matter, claim or Governmental Order involving any Person, that such Person is contesting or appealing such matter, claim or Governmental Order in good faith and by appropriate proceedings timely instituted; provided that the following conditions are satisfied: (a) such Person has established reasonably adequate reserves with respect to the contested items in accordance with GAAP and (b) such contest and any resultant failure to pay or discharge the claimed or assessed amount or to comply with the applicable Governmental Order does not, and could not reasonably (individually or in the aggregate) be expected to, result in a Clean Line Material Adverse Effect or an Adverse DOE Impact.

"Contingency Amount" means (a) as of any date occurring prior to the issuance of the Notice to Proceed, a contingency amount equal to (i) the lesser of (A) 15% and (B) such lower percentage as (1) prior to the occurrence of Project Financial Close, the Coordination Committee may agree based on input from the Independent Engineer and (2) from and after the occurrence of Project Financial Close, the Project Financing Parties determine to be acceptable for purposes of the financing the Construction Costs as part of the Project Financing multiplied by (ii) the amount of all Construction Costs reasonably anticipated by the Clean Line Entities as of such date of determination to be incurred from and after such date of determination in connection with achieving Project Completion (such contingency amount being the "Base Contingency Amount") and (b) on any date occurring after the issuance of the Notice to Proceed, (i) the Base Contingency Amount as of the date of the issuance of the Notice to Proceed minus (ii) the amount of any reduction in the Base Contingency Amount agreed by the Project Financing Parties minus (iii) any amounts of the Base Contingency Amount applied to the payment of Construction Costs since the issuance of the Notice to Proceed.

"Controlling Person" means, with respect to any Person, any other Person that, directly or indirectly Controls such Person.

"Contractor" means any Person with whom a Clean Line Party enters into any Project Contract or performs any part of the Work or provides any materials, equipment, hardware or supplies for any part of the Work and any Person with whom any Contractor has further subcontracted any part of the Work.

"Contractual Obligation" means, as to any Person at any given time, any contractual provision of any security issued by such Person or of any indenture, mortgage, deed of trust, contract, agreement, instrument or other undertaking to which such Person is a party at such time or by which it or any of its Property is bound at such time.

"Contributing Subsidiary Guarantors" has the meaning set forth in Section 9.2.

"Control" means (including, with its correlative meanings, "Controlled by" and "under common Control with") as used with respect to any Person, possession, directly or indirectly, of the power to direct or cause the direction of management or policies of such Person (whether through ownership of voting securities or partnership or other ownership interests, by contract or otherwise); provided, that in any event and for all purposes of the Transaction Documents, (a) with respect to any Investment Fund, such Investment Fund shall be deemed to be Controlled by its general partner and any Person that Controls such general partner and (b) in all other cases, any Person that owns directly or indirectly ten percent (10%) or more of Equity Interests having

ordinary voting powers for the election of directors or other applicable governing body of another Person (but excluding limited partnership or similar types of ownership interests and tax equity investors) shall be deemed to Control such other Person.

"Converter Station Facility" has the meaning set forth in the recitals.

"Converter Station Real Estate Rights Agreements" has the meaning set forth in Section 6.2(a)(v).

"Coordination Committee" has the meaning set forth in Section 5.1(a).

"Covered Cost" has the meaning set forth in Section 11.1.

"Covered Liability" means any and all liabilities (including, without, limitation, negligence, warranty, statutory, product, strict or absolute liability, liability in tort or otherwise), obligations, losses, settlements, damages, penalties, fines (including NERC fines), sanctions, Taxes, claims, actions, demands, suits, judgments or proceedings of any kind and nature, costs, payments, expenses and disbursements (including fees and expenses of consultants, advisors, external counsel and allocable fees and expenses of internal personnel and attorneys) of whatsoever kind and nature (whether or not any of the transactions contemplated by any of the Transaction Documents are consummated), imposed on, incurred or suffered by, or asserted against any Covered Party in any way relating to or arising out of:

- (a) the Project, the Project Facilities, any of the Project Real Estate Rights, or any portion or interest in any one or more of the foregoing;
- (b) the Transaction Documents, the Project Contracts, the Project Financing Documents or the transactions contemplated thereby or the enforcement of any of the rights, remedies or terms of any thereof;
- (c) the conduct of the business or affairs of any Clean Line Party and the Project;
- (d) the sale or providing of any transmission services or any non-delivery by any Clean Line Party of any transmission services in respect of the Project;
- (e) the design, condition, operation, use, non-use, ownership, lease, sublease, maintenance, repair, substitution, possession, rental, conversion, return, registration, re-registration, alteration, overhaul, modification, improvement, testing, removal, replacement, installation, storage, severance, transfer of title, decommissioning, abandonment, sale, resale or other application or disposition or use of any of the Project, the Project Facilities, or any Project Real Estate Rights or any portion of or interest in any one or more of the foregoing or any other Property in which any Clean Line Party or, solely to the extent such Property relates to the Project, any other Property in which DOE, has an interest, including any Covered Liability in any way relating to or arising out of (i) failure to comply with, or costs of compliance with any Environmental Claim or any Release of any Hazardous Substance, (ii) loss or damage to any Property or the environment or death or injury to any Person resulting therefrom, (iii) any patent, trademark or

- copyright infringement relating thereto and (iv) latent or other defects with respect to the Project Facilities, regardless of whether discoverable and including injury, death and Property damage to other with respect to the foregoing;
- (f) the nonperformance or breach by any Clean Line Obligor or any Project Participant of any of its covenants or obligations or the falsity of any representation or warranty obligations under any Project Contract, Project Financing Document, Transaction Document or any other Contractual Obligation to which it is a party under or in respect of any Required Approval or any act, or omission to act in breach of a legal duty to act with respect to, or in connection with the Project, the Project Facilities, the Project Real Estate Rights or any portion of or interest in any one or more of the foregoing or any other Property in which any Clean Line Party has an interest;
- (g) the offer, sale, delivery, refinancing, funding or syndication of the Project Financing;
- (h) the imposition of any Lien on or with respect to the Project or the Project Facilities in respect of which any Covered Party has an interest (other than Permitted Liens);
- (i) failure of any Clean Line Obligor or any Project Participant to comply with any Applicable Laws (including any Environmental Laws);
- (j) the environmental condition and impact of or from the Project, the Project Facilities or the Project Real Estate Rights or any other Property in which any Clean Line Party has any interest, including personal injuries and injuries to the Property of third parties;
- (k) any regulatory action under any Applicable Law pertaining directly or indirectly to the Project, the Project Facilities or the Project Real Estate Rights or any other Property in which any Clean Line Party has an interest;
- (l) any costs incurred by DOE in connection with its performance or undertaking of any non-delegable obligations or responsibilities under the DOE Mitigation Action Plan, any Cultural Resource Agreement with NHPA, the Endangered Species Act or any other Applicable Law to the extent relating to the Project; or
- (m) any of the foregoing Covered Liabilities set forth in clauses (i)-(k), to the extent in connection with, arising out of or in any way related to the design, development, construction, financing, ownership, operation, maintenance or management of the TN Facilities or the TX Facilities and any Real Estate Rights related thereto or any other business conducted by any Clean Line Party from time to time.

"Covered Party" means the United States of America and each of its agencies, departments (including the Department and SWPA), authorities and instrumentalities and each of their elected officials, board members, secretaries, officers, directors, employees, counsel, financial advisors, technical consultants, or agents of any thereof.

"Credit-Worthy Affiliate" means, as of any date of determination, any Affiliate of a Clean Line Entity that either (a) as of any date of determination, is an Acceptable Support Provider or (b) has its obligations in respect of its applicable Project Equity Commitment supported in full by an unconditional and irrevocable letter of credit issued by an Acceptable Support Provider or by an unconditional and irrevocable Guarantee issued by an Acceptable Support Provider, which letter of credit or Guarantee is in an Acceptable Form.

"<u>Cultural Resource Agreement</u>" means a programmatic agreement, memorandum of agreement, and/or binding commitment concerning historic properties for purposes of complying with Section 106 of the NHPA in the development, construction and operation of Project.

"Curative Party" has the meaning set forth in Schedule 1 hereto.

"<u>Davis-Bacon Act</u>" means Subchapter IV of Chapter 31 of Part A of Subtitle II of Title 40 of the United States Code, including, and as implemented by, the regulations set forth in Parts 1, 3 and 5 of title 29 of the Code of Federal Regulations.

"Davis-Bacon Requirements" means, to the extent that DOE (or the Department of Labor, as the case may be) has made a determination that the Davis-Bacon Act is applicable to this Agreement and/or the Project: (a) the Davis-Bacon Act and (b) as set forth in Schedule 15 hereto (as such Schedule is supplemented from time to time in accordance with Section 8.24(b)): (i) all regulations related to the Davis-Bacon Act, including those set forth in the Department of Labor regulations at 29 C.F.R. §§ 5.5(a)(1) to (10) and 5.5(b)(1) to (4) and (ii) applicable wage determinations containing locally prevailing wages as determined by the Secretary of Labor.

"DC" means direct current.

"<u>Deadlock</u>" has the meaning set forth in Section 5.1(f).

"<u>Debarment Regulations</u>" means (a) the Government-wide Debarment and Suspension (Non procurement) regulations (Common Rule), 53 Fed. Reg. 19204 (May 26, 1988), (b) Subpart 9.4 (Debarment, Suspension, and Ineligibility) of the Federal Acquisition Regulations, 48 C.F.R. §§ 9.400 – 9.409 and (c) the revised Government-wide Debarment and Suspension (Non-procurement) regulations (Common Rule), 60 Fed. Reg. 33037 (June 26, 1995).

"<u>Debt Collection Improvement Act</u>" means the Debt Collection Improvement Act of 1996, as amended from time to time.

"<u>Default</u>" means an event that, with the giving of notice or passage of time or both, would become an Event of Default.

"<u>Default Rate</u>" means the lesser of (a) four percent (4%) *per annum* above the Agreed Rate and (b) the maximum rate of interest permitted by Applicable Law.

"Department" has the meaning set forth in the preamble.

"<u>Disposition</u>" means, with respect to any Property, any direct or indirect sale, lease, license, assignment, exchange, conveyance or other transfer or disposition thereof (including the granting of any Lien or security interest in respect thereof), whether by agreement, operation of law or otherwise other than licenses of or similar arrangements for intellectual property rights (and the verb "Dispose" shall be construed accordingly).

"Documentation Package" has the meaning set forth in Schedule 1 hereto.

"DOE" has the meaning set forth in the preamble.

"<u>DOE Acquired Real Property</u>" means any Real Estate Rights acquired by DOE pursuant to the terms of this Agreement.

"DOE Approved Project Equity Commitment" means any Project Equity Commitment that is in an Acceptable Form.

"<u>DOE Approved Project Financing Commitment</u>" means any Project Financing Commitments that are in an Acceptable Form.

"DOE Delegated Real Estate Right" has the meaning set forth in Section 3.3(a).

"DOE Direct Agreement" means one or more consents to assignment or direct agreements to be entered into between DOE and the Project Financing Parties (or their applicable agent) providing for, among other terms to be agreed between DOE and the Project Financing Parties, certain step-in and lender cure rights in favor of the Project Financing Parties in respect of this Agreement and other Transaction Documents, in form and substance acceptable to DOE.

"DOE Instituted Disposition" has the meaning set forth in Section 3.3(f).

"DOE Mitigation Action Plan" has the meaning set forth in Section 4.2.

"<u>DOE Policies</u>" means such practices and policies as are generally applied by DOE or SWPA from time to time with respect to the ownership, operation and maintenance of its real property and transmission assets and as shall be notified by DOE to Holdings in writing from time to time, whether or not such practices or policies have the force of law.

"Dollars" or "\$" means the lawful currency of the United States of America.

"<u>Effective Date</u>" means the date on which each of the conditions precedent set forth in Section 6.1 are satisfied.

"<u>Electric Reliability Organization</u>" means an organization certified by FERC to adopt and enforce mandatory standards for the reliable operation and planning of the bulk power system throughout the United States of America.

"<u>Electrical Capacity</u>" means the electric transmission transfer capability of the Project Facilities (or the applicable portion thereof) expressed in MW, which initially shall be 4,355 MW (gross) or 4,000 MW (net) in the aggregate.

"Emergency" means, with respect to the Project, AR Facilities or the DOE Acquired Real Property, an unplanned event that (a) is an abnormal system condition that requires immediate action to prevent or limit loss of transmission facilities that could adversely affect the reliability of the Project or the AR Facilities; (b) presents an immediate or imminent threat to the long term integrity of any part of the AR Facilities or the Project; (c) presents an immediate or imminent threat of endangerment to life, human health, safety or the environment, including damage to adjacent Property; or (d) is recognized or declared by the Federal Emergency Management Administration (FEMA), the U.S. Department of Homeland Security or other Governmental Authority with authority to declare an emergency.

"Emergency Capital Expenditures" shall mean those Capital Expenditures required to be expended consistent with Prudent Utility Practice in order to prevent or mitigate an Emergency that, in the good faith judgment of the Clean Line Entities (as subsequently confirmed by an Independent Engineer), necessitates the taking of immediate measures to prevent or mitigate such Emergency; provided that such expenditures are (a) payable under an insurance policy (in an aggregate amount not to exceed \$5,000,000 in any 12-month period); (b) payable by insurance or a warranty provided under any Project Contract (in an aggregate amount not to exceed \$5,000,000 in any 12-month period); (c) in an amount that does not exceed \$2,000,000 in any 12-month period; or (d) otherwise reasonably necessary to prevent or mitigate an Emergency.

"Emergency Operating Expenses" shall mean those amounts required to be expended consistent with Prudent Utility Practice in order to prevent or mitigate an Emergency that, in the good faith judgment of the Clean Line Entities (as subsequently confirmed by an Independent Engineer), necessitates the taking of immediate measures to prevent or mitigate such Emergency; provided that such expenditures are (a) payable under an insurance policy (in an aggregate amount not to exceed \$5,000,000 in any 12-month period); (b) payable by insurance or a warranty provided under any Project Contract (in an aggregate amount not to exceed \$5,000,000 in any 12-month period); (c) in an amount that does not exceed \$2,000,000 in any 12-month period; or (d) otherwise reasonably necessary to prevent or mitigate an Emergency.

"Endangered Species Act" means the Endangered Species Act of 1973, as amended from time to time, and the regulations promulgated, and any applicable rulings issued, thereunder.

"Environmental Claim" means any and all obligations, liabilities, losses, administrative, regulatory or judicial actions, suits, demands, decrees, claims, Liens, judgments, notices of noncompliance or violation, investigations (excluding routine inspections), proceedings, clean-up, removal or remedial actions or orders, or damages (foreseeable and unforeseeable, including consequential and punitive damages), penalties, fees, out-of-pocket costs, expenses, disbursements, attorneys' or consultants' fees, relating in any way to any violation of Environmental Law or any violation of any Governmental Approval issued under any such Environmental Law, including (a) any and all indemnity claims by any Governmental Authority for enforcement, cleanup, removal, response, remedial or other actions or damages pursuant to any applicable Environmental Law and (b) any and all indemnity claims by any third party seeking damages, contributions, indemnification, cost recovery, compensation or injunctive relief resulting from Release of Hazardous Substances, the violation or alleged violation of any

Environmental Law or Governmental Approval issued thereunder, or arising from alleged injury or threat of injury to health, safety or the environment.

"Environmental Laws" means any Applicable Law regulating, relating to or imposing obligations, liability or other compliance requirements concerning (a) environmental impacts resulting from the use of the Project Site or environmental conditions present on, in or under the Project Site (including, without limitation, NEPA and NHPA); (b) pollution, protection of human or animal health or safety or the environment, including flora and fauna; (c) Releases or threatened Releases of pollutants, contaminants, chemicals, radiation or Hazardous Substances; (d) otherwise relating to the generation, manufacture, processing, distribution, use, treatment, storage, recycling, disposal, transport, or handling of pollutants, contaminants, chemicals or Hazardous Substances; or (e) any noise generated by or in connection with the Project, including any regulations or guidance relating in any way to the Noise Control Act, 42 USC Section § 4901, et seq. or any related state laws or requirements governing noise or noise mitigation.

"Equity Collateral" has the meaning set forth in Section 11.6(a)(ii).

"Equity Interests" means, with respect to any Person, any and all shares, interests, rights to purchase, warrants, options, participations or other equivalents of or interests in (however designated) the common or preferred equity or preference share capital of a Person, including partnership interests and limited liability company interests.

"Event of Default" has the meaning set forth in Section 7.3.

"Event of Loss" means any event that causes any material portion of the Project to be damaged, destroyed or rendered unfit for normal use for any reason whatsoever, including, through a failure of title, or any loss of Property, or a condemnation.

"Existing Indebtedness" means the Indebtedness of the Clean Line Parties outstanding as of the Effective Date, as set forth in Schedule 3 hereto (as such Schedule may be updated pursuant to Section 12.3).

"Fair Share" means, with respect to a Contributing Subsidiary Guarantor as of any date of determination, an amount equal to (a) the ratio of (i) the Fair Share Contribution Amount with respect to such Contributing Subsidiary Guarantor to (ii) the aggregate of the Fair Share Contribution Amounts with respect to all Contributing Subsidiary Guarantors *multiplied* by (b) the aggregate amount paid or distributed on or before such date by all Funding Subsidiary Guarantors in respect of the Guaranteed Obligations.

"Fair Share Contribution Amount" means, with respect to a Contributing Subsidiary Guarantor as of any date of determination, the maximum aggregate amount of the obligations of such Contributing Subsidiary Guarantor under the Guarantee in Article IX that would not render its obligations hereunder or thereunder subject to avoidance as a fraudulent transfer or conveyance under Section 548 of Title 11 of the United States Code or any comparable applicable provisions of state law; provided that, solely for purposes of calculating the "Fair Share Contribution Amount" with respect to any Contributing Subsidiary Guarantor for purposes of Section 9.2, any assets or liabilities of such Contributing Subsidiary Guarantor arising by virtue of any rights to subrogation, reimbursement or indemnification or any rights to or

obligations of contribution hereunder shall not be considered as assets or liabilities of such Contributing Subsidiary Guarantor.

"FERC" means the Federal Energy Regulatory Commission.

"Final Environmental Impact Statement" means the *Plains & Eastern Clean Line Transmission Line Project Final Environmental Impact Statement* and all Appendices, dated October 2015, Document No. DOE/EIS-0486.

"Financial Statements" means with respect to any Person and for any period, such Person's balance sheet as at the end of such period and the related statements of income, changes in shareholder's equity, and cash flows for such period, all in reasonable detail and prepared in accordance with GAAP, with (a) any such statements delivered in respect of any fiscal year of such Person including all notes thereto and (b) any such statements delivered in respect of any fiscal quarter of such Person being subject only to normal year end audit adjustments and the absence of footnotes.

#### "Financing Condition" means:

- (a) as of any time of determination occurring from and after Project Financial Close, (i) the sum of the unused commitments of the Project Financing Parties under the Project Financing Documents *plus* the sum of the unused Project Equity Commitments *plus* amounts on deposit in the Advance Funding Account is equal to not less than the Remaining Project Costs, (ii) such Project Equity Commitments are in full force and effect without any materially adverse amendment, waiver, supplement or modification thereof from the form that was originally delivered for purposes of satisfying the Financing Condition, and (iii) the Project Financing Documents are in full force and effect and an Authorized Officer of Holdings shall have certified to DOE that, subject to its Knowledge, there is no existing fact or circumstance that will prevent all of the Permitted Draw Conditions from being satisfied, unless the applicable Project Financing Parties have agreed in writing to waive such condition(s); or
- (b) as of any time of determination occurring prior to Project Financial Close, (i) Firm Project Equity Commitments shall be in effect (together with amounts on deposit in the Advance Funding Account) for an amount not less than 150% of the Remaining DOE Acquisition Costs as of such time of determination, (ii) the sum of DOE Approved Project Equity Commitments then in effect *plus* any Firm Project Equity Commitments then in effect *plus* amounts on deposit in the Advance Funding Account *plus* the commitments provided for under any DOE Approved Project Financing Commitment (if any) shall be not less than the Remaining Project Costs, (iii) all such DOE Approved Project Equity Commitments shall be in full force and effect without any materially adverse amendment, waiver, supplement or modification thereof from the form that was originally delivered for purposes of satisfying the Financing Condition, and (iv) if applicable, the DOE Approved Project Financing Commitments shall be in full force and effect without any materially adverse amendment, waiver, supplement

or modification thereof from the form that was originally delivered for purposes of satisfying the Financing Condition and an Authorized Officer of Holdings shall have certified to DOE that, subject to its Knowledge, there is no existing fact or circumstance that will prevent all of the conditions precedent to Project Financial Close in respect of any such DOE Approved Project Financing Commitments or DOE Approved Project Equity Commitments from being achieved prior to the termination of any such DOE Approved Project Financing Commitment or DOE Approved Project Equity Commitments, as applicable, or the Clean Line Entities' needing to draw on any funds in respect of the Project Financing in order to pay any Remaining Project Costs, unless the applicable Financing Party has agreed in writing to waive such condition precedent.

"<u>Financing Party</u>" means any Project Financing Party or any other Person that provides a Project Equity Commitment.

"<u>Firm Project Equity Commitment</u>" means either (a) an irrevocable and unconditional Project Equity Commitment that is subject only to a request of funding (whether in the form of a capital call notice, flow of funds memorandum or otherwise) by Holdings and is not subject to any other condition or right of termination by the Acceptable Support Provider or Credit-Worthy Affiliate providing such Project Equity Commitments or (b) any other Project Equity Commitment that is in an Acceptable Form.

"FPA" means the Federal Power Act, as amended, and FERC's regulations thereunder.

"Force Majeure" means the occurrence of any event or act that delays or prevents a Party's performance of its obligations under the Transaction Documents or any Project Contract, but only to the extent that (a) such event is not attributable to the fault or negligence on the part of such Party, (b) such event is caused by factors beyond such Party's reasonable control and (c) despite taking all reasonable technical and commercial precautions and measures to prevent, avoid, mitigate or overcome such event and the consequences thereof, such Party has been unable to prevent, avoid, mitigate or overcome any such event or consequence, including, but not limited to, acts of God, strikes, lockouts, acts of the public enemy, wars, blockades, riots, insurrections, epidemics, landslides, lightning, earthquakes, fires, storms, floods, washouts, arrests and restraints of rulers and peoples, interruptions by government or court orders or orders of any regulatory body having proper jurisdiction, civil disturbances, explosions, breakage or accident to machinery and any other cause whether of the kind herein enumerated or otherwise. Force Majeure shall not include: (i) economic hardship of a Party or (ii) if claimed by DOE, interruptions by any Governmental Authority or Governmental Order that are directly caused by actions of DOE that specifically are targeted at any of the Clean Line Entities or the Project (and not of a more generally applicable nature) or unless arising as a result of a violation of Applicable Law by any Clean Line Entity, or the occurrence of an Event of Default.

# "Fundamental Event of Default" means the occurrence of:

(a) any Event of Default of the type described in Section 7.3(f) that (i) has resulted in a Safety Event, (ii) arises as a result of a material breach by any Clean Line Entity of any material provision of the DOE Mitigation Action Plan, any Applicable

Laws or Required Approvals or (iii) has resulted in the failure of the Required Insurance to be in full force and effect; <u>provided</u> that to the extent any such Event of Default is capable of being cured, such Event of Default shall not constitute a "Fundamental Event of Default" until such time as DOE has given notice to Holdings of the occurrence of such Event of Default and has given the Clean Line Entities an additional sixty (60) day period to cure such Event of Default; or

- (i) any Event of Default of the type described in Section 7.3(e) that, except to the (b) extent that the representation or warranty giving rise to the occurrence of such Event of Default is itself qualified by "Adverse DOE Impact" or "Clean Line Material Adverse Effect", has resulted in, or could reasonably be expected to result in, a Clean Line Material Adverse Effect or an Adverse DOE Impact, (ii) any Event of Default of the type described in Section 7.3(f) that arises as a result of any Clean Line Entity's breach of any of Section 2.3(c), Section 3.2(d), Section 4.1(i), Section 8.2, Section 8.3, Section 8.7(a), Section 8.11, Section 8.12(a), Section 8.12(c), Section 8.12(e), Section 8.13, Section 8.14, Section 8.18, Section 8.19 or Section 8.27, (iii) any Event of Default of the type described in Section 7.3(g), (iv) any Event of Default of the type described in Section 7.3(h) or (v) any Event of Default of the type described in Section 7.3(l); provided that to the extent any such Event of Default is capable of being cured, such Event of Default shall not constitute a "Fundamental Event of Default" until such time as DOE has given notice to Holdings of the occurrence of such Event of Default and has given the Clean Line Entities an additional sixty (60) day period to cure such Event of Default; or
- (c) an Event of Default of the type described in Section 7.3(j); <u>provided</u> that to the extent such Event of Default arises as a result of an Insolvency Event of a Clean Line Guarantor, such Event of Default shall not constitute a "Fundamental Event of Default" if, within thirty (30) days of the occurrence thereof, the Clean Line Entities have provided a replacement Performance Support from an Acceptable Support Provider in accordance with Section 11.5(a); or
- (d) any Event of Default of the type described in Sections 7.3(a) through 7.3(d), Section 7.3(i), Section 7.3(k) and Sections 7.3(m) through 7.3(r).

"Funding Subsidiary Guarantor" has the meaning set forth in Section 9.2.

"GAAP" means generally accepted accounting principles in the United States in effect from time to time, applied on a consistent basis.

"Governmental Approval" means any approval, consent, authorization, license, permit, order, certificate, qualification, waiver, exemption, or variance, or any other action of a similar nature, of or by a Governmental Authority, including any of the foregoing that are or may be deemed given or withheld by failure to act within a specified time period.

"Governmental Authority" means any federal, state, county, municipal, or regional authority, or any other entity of a similar nature, exercising any executive, legislative, judicial

(including any court of competent jurisdiction), regulatory, or administrative function of government with statutory jurisdiction over any Clean Line Party or the Project.

"Governmental Order" means with respect to any Person, any judgment, order, decision, or decree, or any action of a similar nature, of or by a Governmental Authority having competent jurisdiction over such Person or any of its Properties; provided that the term "Governmental Order" shall not include any judgment, order, decision, or decree, or any action of a similar nature taken by DOE that is specifically targeted at the Clean Line Entities or the Project (and not of a more generally applicable nature) except such judgments, orders, decisions, decrees or actions as may arise based on a violation of Applicable Law by the Clean Line Entities or the occurrence of any Event of Default.

"Guarantee" means (a) any obligation, contingent or otherwise, of the guarantor guaranteeing or having the economic effect of guaranteeing any Indebtedness of any other Person (the "primary obligor") in any manner, whether directly or indirectly, and including any obligation of the guarantor, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness (whether arising by virtue of partnership arrangements, by agreement to keep well, to purchase assets, goods, securities or services, to take or pay or otherwise) or to purchase (or to advance or supply funds for the purchase of) any security for the payment of such Indebtedness, (ii) to purchase or lease Property, securities or services for the purpose of assuring the owner of such Indebtedness of the payment thereof, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity of the primary obligor so as to enable the primary obligor to pay such Indebtedness, (iv) entered into for the purpose of assuring in any other manner the holders of such Indebtedness of the payment thereof or to protect such holders against loss in respect thereof (in whole or in part) or (v) as an account party in respect of any letter of credit or letter of guarantee issued to support such Indebtedness, or (b) any Lien on any assets of the guarantor securing any Indebtedness (or any existing right, contingent or otherwise, of the holder of Indebtedness to be secured by such a Lien) of any other Person, whether or not such Indebtedness is assumed by the guarantor; provided, however, that the term "Guarantee" shall not include endorsements for collection or deposit, in either case in the ordinary course of business.

"Guaranteed Obligations" has the meaning set forth in Section 9.1(a).

"Hazardous Substances" means any hazardous or toxic substances, chemicals, materials, pollutants or wastes defined, listed, classified or regulated as such in or under any Environmental Laws, including (a) any petroleum or petroleum products (including gasoline, crude oil or any fraction thereof), flammable explosives, radioactive materials, asbestos in any form that is or could become friable, urea formaldehyde foam insulation and polychlorinated biphenyls, (b) any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "extremely hazardous wastes," "restricted hazardous wastes," "toxic substances," "toxic pollutants," "contaminants" or "pollutants," or words of similar import, under any applicable Environmental Law and (c) any other chemical, material or substance, import, storage, transport, use or disposal of, or exposure to or Release of which is prohibited, limited or otherwise regulated under, or for which liability is imposed pursuant to, any Environmental Law.

"<u>Hazardous Substances Measures</u>" means those measures adopted by the Clean Line Entities as part of implementing the DOE Mitigation Action Plan as further described in Section 8.21, including (a) procedures, practices and activities to address and comply with Environmental Laws and Governmental Approvals with respect to any Release of Hazardous Substances in connection with the Project or any Project Real Estate Right and (b) actions to be taken in the event that a Hazardous Substance is discovered on Property on which Project Facilities are located.

"Holdings" has the meaning set forth in the preamble.

"HVDC" has the meaning set forth in the recitals.

"Immaterial Obligor" means any Clean Line Obligor (other than Holdings, ACL, PECL OK and PECL) that (a) has Property with a fair market value of less than \$5,000,000, (b) does not own any rights to any of the Electrical Capacity, (c) does not own any Project Real Estate Rights, (d) is not a party to any Material Project Contract and (e) as to which the occurrence of any Insolvency Event in respect of such Clean Line Obligor could not reasonably be expected to have a Clean Line Material Adverse Effect or an Adverse DOE Effect.

"Indebtedness" of any Person shall mean, without duplication:

- (a) all Indebtedness for Borrowed Money;
- (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments;
- (c) all obligations of such Person under conditional sale or other title retention agreements relating to Property or assets purchased by such Person;
- (d) all Guarantees by such Person of Indebtedness of others;
- (e) all Capital Lease Obligations of such Person;
- (f) the principal component of all obligations, contingent or otherwise, of such Person (i) as an account party in respect of letters of credit and (ii) in respect of bankers' acceptances, bank guaranties, surety or performance bonds and similar instruments; and
- (g) all obligations of such Person to purchase, redeem, retire, defease any Equity Interests in such Person or any warrants, rights or options to acquire such Equity Interests, valued, in the case of redeemable preferred interests, at the greater of its voluntary or involuntary liquidation preference *plus* accrued and unpaid dividends.

"Indebtedness for Borrowed Money" means as to any Person, without duplication, (a) all indebtedness (including principal, interest, fees and charges) of such Person for borrowed money or for the deferred purchase price of property or services (other than any deferral (i) in connection with the provision of credit in the ordinary course of business by any trade creditor or

utility or (ii) of any amounts payable under the Project Contracts) or (b) the aggregate amount required to be capitalized under any Capital Lease under which such Person is the lessee.

"Independent Engineer" means an independent engineer selected by (a) Holdings or (b) following satisfaction of the Financing Condition, by or on behalf of the Project Financing Parties that have executed DOE Approved Project Financing Commitments and that, in either case, is reasonably acceptable to DOE.

"Information" has the meaning set forth in Section 12.1(o).

"<u>Insolvency Event</u>" means, with respect to any Person, the occurrence of any of the following events, conditions or circumstances:

- (a) such Person shall file a voluntary petition in bankruptcy or shall be adjudicated bankrupt or insolvent, or shall file any petition or answer or consent seeking any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief for itself under applicable Bankruptcy Law, or shall seek or consent to or acquiesce in the appointment of any trustee, receiver, conservator or liquidator of such Person or of all or any substantial part of its Properties (the term "acquiesce," as used in this definition, includes the failure to file in a timely manner a petition or motion to vacate or discharge any order, judgment or decree after entry of such order, judgment or decree);
- (b) an involuntary case or other proceeding shall be commenced against such Person seeking any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief with respect to such Person or its debts under applicable Bankruptcy Law, or seeking the appointment of a trustee, receiver, liquidator, custodian or other similar official of it or any substantial part of its Property, and such involuntary case or other proceeding shall remain undismissed or unstayed for a period of ninety (90) consecutive days;
- (c) a court of competent jurisdiction shall enter an order, judgment or decree approving a petition filed against such Person seeking a reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under the applicable Bankruptcy Law, and such Person shall acquiesce in the entry of such order, judgment or decree or such order, judgment or decree shall remain unvacated and unstayed for an aggregate of ninety (90) days (whether or not consecutive) from the date of entry thereof, or any trustee, receiver, conservator or liquidator of such Person or of all or any substantial part of its Property shall be appointed without the consent or acquiescence of such Person and such appointment shall remain unvacated and unstayed for an aggregate of ninety (90) days (whether or not consecutive);
- (d) such Person shall admit in writing its inability to pay its debts as they mature or shall generally not be paying its debts as they become due;
- (e) such Person shall make an assignment for the benefit of creditors or take any other similar action for the protection or benefit of creditors; or

(f) such Person shall take any corporate, limited liability company or partnership action for the purpose of effecting any of the foregoing.

"Insurance Agreement" has the meaning set forth in Section 5.1(e)(ii).

"Interconnection Agreement" means an interconnection agreement or interim interconnection agreement, filed with FERC as applicable, granting a definitive interconnection right for the Project, subject to the completion of Material Interconnection Studies and required transmission system upgrades identified within such interconnection agreement or interim interconnection agreement, which is entered into by any applicable Clean Line Party and the interconnecting transmission owner(s) and system operators for (a) interconnection of the Project with the SPP-controlled transmission system, (b) interconnection of the Project with the MISO-controlled transmission system and/or (c) interconnection of the TN Facilities with the TVA transmission system.

"Intercreditor Agreement" has the meaning set forth in Section 11.7.

"Intermediate Converter Station" has the meaning set forth in the recitals.

"Investment Fund" means any Person that is established as an investment fund and is either (a) registered with the Securities and Exchange Commission under the Investment Company Act of 1940 or (b) exempt from registration under the Investment Company Act of 1940 pursuant to Section 3(c)(1) or 3(c)(7) of such Act.

"IPO Entity" has the meaning set forth in the definition of Qualified IPO.

"Key Person" means each of Mario Hurtado and Michael Skelly.

"Knowledge" means (a) with respect to any Clean Line Party, the actual knowledge of any Principal Person of such Clean Line Party or any knowledge that should have been obtained by any such Principal Person upon reasonable investigation and inquiry and (b) with respect to DOE, the actual knowledge of any Principal Person or any knowledge that should have been obtained by any such Principal Person upon reasonable investigation and inquiry. "Knowing" and "Known" shall be construed accordingly.

"Landowner" has the meaning set forth in Schedule 1 hereto.

"Lien" means any lien (statutory or other), pledge, mortgage, charge, security interest, deed of trust, assignment, hypothecation, title retention, fiduciary transfer, deposit arrangement, easement, encumbrance or preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever in respect of an asset, whether or not filed, recorded or otherwise perfected or effective under Applicable Law, as well as the interest of a vendor or lessor under any conditional sale agreement, Capital Lease or other title retention agreement relating to such asset, (including any conditional sale or other title retention agreement, any Capital Lease having substantially the same economic effect as any of the foregoing, or any preferential arrangement having the practical effect of constituting a security interest with respect to the payment of any obligation with, or from the proceeds of, any asset or revenue of any kind).

"<u>Local Government Contribution Payments</u>" means all infrastructure payments, voluntary payments and other payments (which are not Taxes) to be made by any Clean Line Party to local and state governments in connection with the Project, including those set forth in Schedule 4 hereto.

"<u>Loss Proceeds</u>" means all proceeds (other than any proceeds of business interruption insurance and proceeds covering liability of the Clean Line Entities to third parties) resulting from an Event of Loss.

"Loss Threshold" shall have the meaning set forth in the Insurance Agreement.

"Major Subcontractor" means each subcontractor at any tier performing work under a Material Construction Contract with a subcontract having a value, individually or in the aggregate, of \$5,000,000 or more.

"Material Construction Contract" means each Construction Contract with a total value of more than \$5,000,000 that relates in any material respect to the development, design, engineering and construction of the AR Facilities or any Project Facilities located (or to be located) on any DOE Acquired Real Property. For the avoidance of doubt, any Construction Contract that solely relates to the Other Facilities or any Project Facilities to be located on any Project Real Estate Rights that are not DOE Acquired Real Property shall not constitute a "Material Construction Contract".

"<u>Material Interconnection Studies</u>" means the TVA Facilities Study, an updated Criteria 3.5 Study as described in Exhibit B to the SPP Interconnection Agreement, the TVA System Impact Study, the SPP Facilities Study, the SPP System Impact Study, the MISO Interconnection Feasibility Study, the MISO Interconnection Facilities Study and the MISO Interconnection System Impact Study (or such comparable studies (i) if renamed or modified by revisions to the interconnection procedures of TVA, SPP and MISO or (ii) as required by an Interconnection Agreement).

"<u>Material O&M Agreement</u>" means each O&M Agreement that relates in any material respect to the operation, management and/or maintenance of the AR Facilities or any Project Facilities located (or to be located) on any DOE Acquired Real Property. For the avoidance of doubt, any O&M Agreement that solely relates to the Other Facilities or any Project Facilities to be located on any Project Real Estate Rights that are not DOE Acquired Real Property shall not constitute a "Material O&M Agreement".

"<u>Material Project Contract</u>" means, at any given time, the Interconnection Agreements, the Transmission Services Agreements, the Acceptable Permitted Project Investment Commitments, the Material Construction Contracts and the Material O&M Agreements, in each case, in effect at such time.

"MISO" has the meaning set forth in the recitals.

"MISO Interconnection Facilities Study" means an "Interconnection Facilities Study," including Definitive Planning Phase studies, as applicable, as defined in the MISO OATT and applicable business practice manuals related thereto.

- "<u>MISO Interconnection Feasibility Study</u>" means an "Interconnection Feasibility Study" as defined in the MISO OATT and applicable business practice manuals related thereto.
- "<u>MISO Interconnection System Impact Study</u>" means an "Interconnection System Impact Study" as defined in the MISO OATT and applicable business practice manuals related thereto.
  - "MISO OATT" means the MISO OATT on file with FERC.
- "<u>Mitigation Rights</u>" means any real Property and conservation credits acquired by the Clean Line Entities associated with the avoidance, minimization, or mitigation of environmental impacts of the Project pursuant to Required Approvals.
  - "Moody's" means Moody's Investors Service, Inc., so long as it is a rating agency.
  - "MW" has the meaning set forth in the recitals.
- "NEPA" means the National Environmental Policy Act of 1969 of the United States, as amended from time to time, and the regulations promulgated, and any applicable rulings issued, thereunder.
  - "NERC" means the North American Electric Reliability Corporation.
  - "NERC Agreement" has the meaning set forth in Section 4.9.
  - "NHPA" means the National Historic Preservation Act of 1966, as amended.
- "Notice to Proceed" means a written notice issued by DOE to Holdings notifying Holdings that the conditions precedent set forth under Section 6.4 have been satisfied and that the Clean Line Entities may notify the Construction Contractors to commence performance of the work under the applicable Material Construction Contracts.
- "O&M Agreement" means any Contractual Obligation entered into for the operation and maintenance of the Project with an annual value of more than \$1,000,000.
- "OATT" means an Open Access Transmission Tariff as defined under FERC's open access transmission rules and policies.
  - "OCC" means the Corporation Commission of Oklahoma.
  - "OCC 2011 Order" has the meaning set forth in Section 12.1(t).
- "OFAC" means the Office of Foreign Assets Control, an agency of the U.S. Department of the Treasury under the auspices of the Under Secretary of the Treasury for Terrorism and Financial Intelligence.
- "OFAC-Listed Person" has the meaning set forth in clause (a) of the definition of Prohibited Person.

"OFAC Sanctions Program" means any economic or trade sanction that OFAC is responsible for administering and enforcing. A list of OFAC Sanctions Programs may be found at http://www.treasury.gov/resource-center/sanctions/Programs/Pages/Programs.aspx.

"OK Facilities" has the meaning set forth in the recitals.

"Oklahoma Panhandle" means the geographic area within the panhandle region of Oklahoma, including Cimarron, Texas, and Beaver Counties.

"OLA" has the meaning set forth in the preamble.

"Operating Agreement" has the meaning set forth in Section 4.6.

"Operational EOD" means an Event of Default that (a) is an Event of Default pursuant to Section 7.3(f), (b) has resulted in an Adverse DOE Impact, (c) in DOE's reasonable judgment has resulted from, or arisen out of, the Clean Line Entities' detrimental, harmful, negligent or incompetent management, monitoring, supervision or administration of the construction, operation or maintenance of the Project on a persistent basis and (d) is not otherwise a Fundamental Event of Default.

"Organizational Documents" means with respect to any Person, (a) to the extent such Person is a corporation, the certificate or articles of incorporation and the by-laws of such Person, (b) to the extent such Person is a limited liability company, the certificate of formation or articles of formation or organization and operating or limited liability company agreement of such Person and (c) to the extent such Person is a partnership, joint venture, trust or other form of business, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization or formation of such Person.

"Other Facilities" means, collectively, the TX Facilities and the TN Facilities.

"Participation Amount" has the meaning set forth in Section 11.2.

"Parties" means the Clean Line Parties and DOE.

"Patriot Act" means the USA PATRIOT Act of 2001 and all rules and regulations adopted thereunder, as amended.

"PECL" has the meaning set forth in the preamble.

"PECL OK" has the meaning set forth in the preamble.

"<u>PECL Subsidiary</u>" means (a) any direct or indirect Subsidiary of Holdings that owns any of the Equity Interests in PECL and (b) any direct or indirect Subsidiary of PECL.

"<u>Performance Support</u>" means any Acceptable Letter of Credit or Acceptable Guarantee issued in favor of DOE from time to time pursuant to the terms of this Agreement.

### "Permitted Disposition" means:

- (a) with respect to any Equity Interests in any Clean Line Entity or Property comprising the OK Facilities, any Permitted Lien or Permitted Project Investment;
- (b) with respect to any Electrical Capacity, any Permitted Lien or other Disposition permitted under Section 2.3(c);
- (c) with respect to any Project Real Estate Rights acquired by any Clean Line Entity, any Permitted Lien or other Disposition contemplated or permitted by Section 3.2;
- (d) with respect to any other Property owned by any Clean Line Entity from time to time:
  - (i) Dispositions of such Property in the ordinary course of business and having a fair market value not in excess of \$1,000,000 for a single transaction or \$5,000,000 in the aggregate for all such transfers or Dispositions; provided that such Property is not necessary to the performance of the Project or the transactions contemplated by the Transaction Documents:
  - (ii) Dispositions of such Property that is, (A) obsolete, (B) no longer used or useful in the operation of the Project or (C) is promptly replaced (if applicable) by new or refurbished Property of equal or greater value and utility or having the same function (including upgraded models);
  - (iii) Dispositions of investment property in the ordinary course or in accordance with the granting of Permitted Liens;
  - (iv) Dispositions in connection with Events of Loss;
  - (v) Dispositions of Property in connection with warranty claims or assignments of Project Contracts permitted by the Project Financing Documents;
  - (vi) Dispositions of Mitigation Rights; or
  - (vii) Dispositions otherwise approved in writing by DOE.

"<u>Permitted Draw Conditions</u>" means conditions requiring the satisfaction of the following:

(a) truthfulness in all material respects of representations and warranties contained in the Project Financing Documents;

- (b) the non-occurrence and continuance of any "default" or "event of default" or "material adverse effect" under the Project Financing Documents;
- (c) the delivery of a customary notice of borrowing; and
- (d) other usual and customary drawdown conditions applicable to construction financings for transmission projects.

### "Permitted Holder" means any of the following Persons:

- (a) National Grid plc;
- (b) solely with respect to the Control of the Clean Line Entities, CLEP (<u>provided</u> that CLEP is Controlled by one or more Persons specified in clause (a) or clauses (c) through (g) of this definition);
- (c) the Zilkha Family;
- (d) the Ziff Family;
- (e) Bluescape Resources;
- (f) any Qualified Owner; or
- (g) any other Person (other than any Clean Line Entity) Controlled by a combination of the foregoing Persons.

#### "Permitted Indebtedness" means:

- (a) Existing Indebtedness;
- (b) obligations or liabilities under any Project Contracts, the Transaction Documents, the Project Equity Commitments, the Project Debt Commitments, the Project Financing Documents;
- (c) other liabilities or obligations allowed under the Project Financing Documents, if applicable;
- (d) any inter-company receivables or payables among Affiliates of Holdings for obligations not constituting Indebtedness that either Holdings or CLEP have paid on behalf of such Affiliates;
- (e) Indebtedness in the nature of guaranties or letters of credit, surety bonds or performance bonds securing the performance of a Clean Line Party pursuant to a Project Contract;
- (f) Indebtedness in respect of netting services, overdraft protections and otherwise in connection with deposit accounts;

- (g) guarantees in the ordinary course of business of the obligations of suppliers, customers, franchisees and licensees of any Clean Line Party; and
- (h) other Indebtedness in an aggregate amount not to exceed \$5,000,000 at any time outstanding.

### "Permitted Liens" means:

- (a) any Liens provided in favor of DOE or any rights and interests of the Project Financing Parties as provided in the Project Financing Documents that could constitute a Lien;
- (b) Liens for any Tax, assessment or other governmental charge not yet due, or subject to Contest;
- (c) Liens in favor of materialmen, workers or repairmen, or other like Liens arising in the ordinary course of business or in connection with the construction, repair or improvement of the Project, either for amounts not yet due or for amounts subject to Contest;
- (d) such other defects, matters or records affecting or encumbering title to the Project Site, which do not and will not materially impair the use, development or operation of the Project, or materially interfere with the ordinary course of the business of the Clean Line Entities, or materially detract from the value of the Project Site;
- (e) zoning, entitlement, building and other land use regulations imposed by Governmental Authorities having jurisdiction over the Project Site that do not and will not materially impair the use, development or operation of the Project, or materially interfere with the ordinary course of the business of the Clean Line Entities, or materially detract from the usefulness of the Project Site for its intended purpose;
- (f) deposits to secure the performance of bids, trade contracts and leases (in each case, other than Indebtedness), statutory obligations, surety bonds (other than bonds related to judgments or litigation), performance bonds and other obligations of a like nature incurred in the ordinary course of business not in excess of \$5,000,000;
- (g) Liens arising out of the judgment of a Governmental Authority so long as enforcement of such Lien has been stayed and an appeal or proceeding for review is being prosecuted in good faith and for the payment of which adequate reserves or other security reasonably acceptable to DOE have been provided or are fully covered by insurance;
- (h) Liens to secure Capital Lease Obligations and purchase money Liens on Property purchased securing obligations not in excess of \$5,000,000;

- (i) Liens (not securing Indebtedness) of depository institutions and securities intermediaries (including rights of set-off or similar rights) with respect to one or more checking accounts or other banking accounts (other than Account Collateral) established by the Clean Line Entities to conduct their business;
- (j) Liens securing judgments for the payment of money not constituting an Event of Default or securing appeal or other surety bonds related to such judgments;
- (k) pledges or deposits or other Liens in the ordinary course of business in connection with worker's compensation, unemployment insurance, social security and other governmental rules or restrictions that have the force of law; and
- (l) Liens on Property of the Clean Line Entities not essential for the operation of the Project and having a fair market value of less than \$1,000,000 in the aggregate.

"Permitted Project Investments" has the meaning set forth in Section 2.3(c)(iii).

"Person" means any individual, entity, firm, corporation, company, voluntary association, partnership, limited liability company, joint venture, trust, unincorporated organization, Governmental Authority, committee, department, authority or any other body, incorporated or unincorporated, whether having distinct legal personality or not.

"Principal Person" means (a) with respect to any Clean Line Party, any officer, director, owner, key employee or other Person with primary management or supervisory responsibilities with respect to such Person or any other Person (whether or not an employee) who has critical influence on or substantive control over such Person and (b) with respect to DOE, the Person(s) holding primary management or supervisory responsibilities for DOE with respect to the Project.

"<u>Prohibited Person</u>" means any Person (or any Person that is an Affiliate of a Person) that is:

- (a) named, identified or described on the list of "Specially Designated Nationals and Blocked Persons" as published by OFAC (an "OFAC-Listed Person");
- (b) an agent, department or instrumentality of, or is otherwise beneficially owned by, Controlled by or acting on behalf of, directly or indirectly, (i) an OFAC-Listed Person or (ii) any Person, organization, foreign country or regime that is subject to any Sanctions;
- (c) debarred, suspended, proposed for debarment with a final determination still pending, declared ineligible or voluntarily excluded (as such terms are defined in the Debarment Regulations) from contracting with any United States federal government department or any agency or instrumentality thereof or otherwise participating in procurement or non-procurement transactions with any United States federal government or agency pursuant to any of the Debarment Regulations;

- (d) indicted, convicted or had a final and non-appealable Governmental Order rendered against it for any of the offenses listed in any of the Debarment Regulations; or
- (e) otherwise blocked, subject to sanctions under or engaged in any activity in violation of other United States economic sanctions, including but not limited to, the Trading with the Enemy Act, the International Emergency Economic Powers Act, the Comprehensive Iran Sanctions, Accountability and Divestment Act or any similar law or regulation with respect to Iran or any other country, the Sudan Accountability and Divestment Act, any OFAC Sanctions Program, or any economic sanctions regulations administered and enforced by the United States or any enabling legislation or executive order relating to any of the foregoing.

"Project" has the meaning set forth in the recitals.

"Project Budget" means a reasonably detailed description and a reasonably detailed budget of anticipated Construction Costs (which shall include a Contingency Amount).

"Project Completion" means the occurrence of each of the following: (a) if the Project Financing is then in effect, (i) the Project has commenced commercial operation and has satisfied the requirements for "substantial completion" (or term of similar import) as defined in and in accordance with all Construction Contracts and the initial Electrical Capacity (as specified in the definition thereof) of the Project has been certified by an Independent Engineer, (ii) the Project has safely and reliably energized and energy may be delivered across the Project Facilities to SPP's, MISO's and TVA's transmission systems in accordance with the Interconnection Agreements and (iii) the occurrence of "final completion," "project completion" or "term conversion" (or term of similar import) for purposes of the Project Financing (including the funding of any required financial reserves as a condition precedent to the occurrence thereof) or (b) if the Project Financing is not then in effect, satisfaction of the Completion Conditions.

# "Project Contracts" means each of the following:

- (a) any Construction Contract, including any such agreement that may be entered into from time to time in respect of any Capital Repairs or improvements relating to the Project;
- (b) each Transmission Services Agreement and each Contractual Obligation evidencing a Permitted Project Investment in effect from time to time;
- (c) each Operating Agreement and Interconnection Agreement in effect from time to time;
- (d) the NERC Agreement;
- (e) any O&M Agreement in effect from time to time;
- (f) each Real Estate Rights Agreement; and

(g) any other agreement entered into by the Clean Line Entities and/or DOE in respect of development, design, engineering, construction, ownership, operation, maintenance and management of the Project, including any management service agreements, intellectual property license agreements and any retainer agreements relating to any consultants.

"Project Costs" means all costs and expenses incurred or to be incurred by any Clean Line Entity or DOE in connection with the Project, including: (a) expenses of administering and maintaining the corporate existence of the Clean Line Entities, (b) amounts payable under any other Project Contract in effect from time to time, (c) amounts payable in respect of the Project Financing (including interest, premium, principal and fees), (d) costs to acquire title or use rights of any Project Real Estate Rights and the Project Site, (e) any network upgrade costs required to be paid pursuant to the terms of the Interconnection Agreements, (f) costs and expenses of legal, engineering, accounting, construction management and other advisors or consultants incurred in connection with the Project, (g) labor costs, (h) mobilization costs, (i) funding of any required reserves (including any debt service reserve or other similar reserve required in connection with the Project Financing), (j) maintenance and Capital Repair expenses, (k) costs associated with any Wind-Up Event, (l) costs and expenses incurred in connection with obtaining any Required Approval and Required Insurance, (m) any other Covered Cost and (n) any Covered Liability.

"Project Development Progress Report" means, as of any date, a development progress report in respect of the Project that (a) describes in detail the status of the developmental activities related to the Project completed as of such date and (b) provides a reasonably detailed schedule of the project development, design, engineering, financing and construction activities expected to be undertaken after such date in order to achieve Project Completion (the "Project Schedule").

"<u>Project Equity Commitments</u>" means one or more equity commitments (which may include a commitment to provide loans) provided by any Acceptable Support Provider or Credit-Worthy Affiliate to the Clean Line Entities or any of the Project Financing Parties in respect of the funding of Construction Costs.

"Project Facilities" has the meaning set forth in the recitals.

"Project Financial Close" means that the following conditions have been satisfied in full:

- (a) the Clean Line Entities shall have obtained Project Financing and Project Equity Commitments in an amount equal to 100% of the total Construction Costs as set forth in the then current Project Budget (including the Contingency Amount contemplated thereby) as confirmed, at the election of DOE, by an Independent Engineer;
- (b) all Project Financing Documents required in order to obtain the funding referred to in clause (a) above shall have been executed and delivered by all parties thereto and shall be in full force and effect, and DOE shall have received certified copies of all such Project Financing Documents;

- (c) all conditions precedent (other than the Permitted Draw Conditions) under the Project Financing Documents referred to in clause (b) above shall have been satisfied or permanently waived;
- (d) the first drawdown of loans (or issuance of debt securities, to the extent applicable) under the Project Financing shall have occurred; and
- (e) DOE shall have received satisfactory evidence demonstrating that each of the foregoing conditions has been satisfied.

"Project Financing" means, subject to Section 13.5, any debt securities or syndicated commercial bank or other syndicated credit facilities (including any working capital facilities and letter of credit facilities) issued or obtained by the Clean Line Entities from Persons other than Affiliated Lenders to finance the development and construction of the Project in an amount equal to not less than forty percent (40%) of the anticipated total Construction Costs on a limited recourse basis and any refinancing that takes a similar form.

"<u>Project Financing Commitments</u>" means one or more debt financing commitment letters provided by lenders or financial institutions (other than Affiliated Lenders) in respect of the Project Financing.

"<u>Project Financing Documents</u>" means all financing (including all security documentation) and equity contribution agreements entered into in respect of the Project Financing and the Project Equity Commitments.

"<u>Project Financing Parties</u>" means the Persons holding any debt securities or providing loans, other credit facilities or interest rate hedging facilities as part of the Project Financing (and including any agent or trustee thereof), but excluding any Affiliated Lender. Prior to Project Financial Close, the Project Financing Parties will be deemed to include any lenders or financial institutions providing Project Financing Commitments to the Clean Line Parties.

"<u>Project Participant</u>" means any Contractor or any other Person (other than a Clean Line Party or DOE) that is party to a Project Contract from time to time.

"Project Plans" means the reasonably detailed execution plans for the Project (which shall include a reasonably detailed description of the Project and the Project Budget) delivered by Holdings to DOE encompassing all development, design, engineering, construction, financing, operation, maintenance, management, replacement and decommissioning activities of the Project.

"<u>Project Real Estate Rights</u>" means any Real Estate Rights necessary for the Project, including access roads and temporary areas to be used for construction and maintenance activities in respect of the Project.

"<u>Project Schedule</u>" has the meaning set forth in the definition of Project Development Progress Report.

"<u>Project Site</u>" means all Real Estate Rights on which any of the Project Facilities are situated or are to be constructed, including, but not limited to, the areas and encroachments covered by the Project Real Estate Rights and any other land necessary for the Project.

"Project Subsidiary" means (a) any Subsidiary of Holdings that owns any Property or other rights relating to the Project, including each of ACL, PECL OK and OLA and (b) any Subsidiary of Holdings that, directly or indirectly, owns any Equity Interests of any such Subsidiary; provided that the term "Project Subsidiary" shall not include PECL or any PECL Subsidiary.

"<u>Project Work Agreement</u>" means an agreement with TVA pursuant to which TVA begins the necessary work to construct system upgrades necessary for the Project prior to the execution of an Interconnection Agreement with TVA.

"<u>Property</u>" means any right or interest in or to any asset or property of any kind whatsoever (including Equity Interests), whether real, personal or mixed and whether tangible or intangible.

"Prudent Utility Practices" means any of the acts, practices, methods, equipment, materials, specifications and standards engaged in or approved in connection with a significant portion of the electric utility industry in North America which, as applicable, in the exercise of professional judgment in light of the facts known at the time a decision was made, would have been expected to accomplish the desired result in a manner consistent with Applicable Laws, DOE Policies, any Electric Reliability Organization requirements, reliability, safety, dependability, environmental protection and expedition.

"PUHCA" means the Public Utility Holding Company Act of 2005, and FERC's regulations thereunder.

"Qualified IPO" means the issuance by any Clean Line Entity or any Person that Controls Clean Line (the "IPO Entity") of its common Equity Interests in an underwritten primary public offering (other than a public offering pursuant to a registration statement on Form S-8) pursuant to an effective registration statement filed with the Securities Exchange Commission in accordance with the Securities Act of 1933 (whether alone or in connection with a secondary public offering) which after giving effect thereto shall result in (a) no Person or group of Persons having Control over the IPO Entity or any of the Clean Line Entities (other than the IPO Entity) and any direct or indirect Person in the ownership chain between the IPO Entity and the Clean Line Entities (other than the IPO Entity) and (b) the common voting Equity Interests of such IPO Entity being traded on a regulated United States securities exchange.

"Qualified Owner" means any Person meeting all of the following requirements at the time of its acquisition of any direct or indirect Equity Interests in the Clean Line Entities, as applicable:

(a) (i) neither such Person nor any Person that, directly or indirectly, Controls such Person or any of their respective Principal Persons is a Prohibited Person and (ii) no event has occurred and no condition exists that is likely to result in such

- Person or any Person that, directly or indirectly Controls such Person or any of their respective Principal Persons becoming a Prohibited Person;
- (b) such Person does not owe any delinquent Indebtedness to any Governmental Authority of the United States, including Tax liabilities, except to the extent such delinquency has been resolved (or is in the process of being resolved) with the appropriate Governmental Authority in accordance with the standards of the Debt Collection Improvement Act;
- (c) (i) such Person, and each Person that, directly or indirectly, Controls such Person, and each of their respective Principal Persons, employees and agents have complied with OFAC, all other applicable Anti-Corruption Laws and all AM laws in obtaining any consents, licenses, approvals, authorizations, rights or privileges with respect to such Person's acquisition of any direct or indirect Equity Interests in the Clean Line Entities and (ii) the internal management and accounting practices and controls of such Person and each Person that, directly or indirectly, Controls such Person are adequate to ensure compliance with all applicable Anti-Corruption Laws, AM Laws and Sanctions;
- (d) such Person is organized under the laws of an Organization for Economic Cooperation and Development member country;
- (e) such Person has provided DOE all documentation and other information under applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act, that would customarily be provided or delivered to a financial institution in connection with a transaction involving an extension of credit to such Person at least thirty (30) days prior to its acquisition, directly or indirectly, of any Equity Interests in the Clean Line Entities and DOE shall not have notified Holdings of its objection to such Person's acquisition of such interests within such thirty (30) day period; and
- (f) all necessary Governmental Approvals arising as a result of such Person's acquisition of such Equity Interests shall have been obtained and in full force and effect, and, to the extent applicable, the Committee on Foreign Investment in the United States shall have approved such acquisition.

"Real Estate Rights" means any real property rights, including temporary property rights and access rights (whether in the form of fee simple, a leasehold, easement, sub-easement, right of way, license, permit, concession or otherwise).

"Real Estate Rights Agreement" means any agreement entered into from time to time by any Clean Line Entity and/or DOE in respect of the acquisition of any Project Real Estate Rights (including any easement or right of way).

"Release" means disposing, discharging, injecting, spilling, leaking, leaching, dumping, pumping, pouring, emitting, escaping, emptying, seeping, placing and the like, into or upon any land or water or air, or otherwise entering into the environment.

"Release Provision" has the meaning set forth in Section 11.10(a).

"Remaining DOE Acquisition Costs" means, as of any time of determination, the aggregate of all Covered Costs reasonably anticipated to be incurred in connection with the acquisition of any Project Real Estate Rights reasonably anticipated to be designated as DOE Acquired Real Property as of such time of determination, as determined from time to time by the Coordination Committee.

"Remaining Project Costs" means (a) as of the issuance of the Notice to Proceed, (i) the amount of all Construction Costs reasonably anticipated by the Clean Line Entities to be incurred after the issuance of the Notice to Proceed in connection with achieving Project Completion (based on the then applicable Project Plans and Project Schedule) as confirmed, at the election of DOE, by an Independent Engineer *plus* (ii) the Base Contingency Amount and (b) as of any time of determination after the issuance of the Notice to Proceed, the sum of (i) the amount of all Construction Costs reasonably anticipated by the Clean Line Entities as of such time of determination to be incurred from and after such time of determination in connection with achieving Project Completion (based on the then applicable Project Plans and Project Schedule) as confirmed, at the election of DOE, by an Independent Engineer *plus* (ii) the then applicable Contingency Amount.

"Representation Date" has the meaning set forth in Section 12.1.

"Required Amount" means, as of any date of determination, the amount equal, without duplication, to the sum of the Base Amount *plus* the Advanced Funding Contingency Amount.

"Required Approvals" means all material Governmental Approvals and other material consents and approvals of third parties necessary or required under Applicable Law, DOE Policies or any Contractual Obligation for the development, design, engineering, construction, financing, ownership, operation, maintenance, management, replacement and decommissioning of the Project and the sale and provision of transmission services over the Project Facilities.

"Required Insurance" means insurance coverage for the Project as required by the Insurance Agreement as in effect from time to time.

"RFP" has the meaning set forth in the recitals.

"Routing and ROW Plan" means a plan prepared by Holdings, and acceptable to the Coordination Committee, specifying the planned routing corridor for the Project Facilities, identifying all Project Real Estate Rights and including a reasonably detailed budget covering all costs, expenses and disbursements projected to be expended in connection with the acquisition of such Project Real Estate Rights.

"S&P" means Standard & Poor's Financial Services LLC or its successor, so long as it is a rating agency.

"Safety Compliance" means with respect to the Project Facilities any and all improvements, repair, reconstruction, rehabilitation, restoration, renewal, replacement and changes in configuration or procedures respecting the Project Facilities to correct a specific

Safety Event that DOE has reasonably determined to exist by investigation or analysis, <u>provided</u> that DOE's determination shall be consistent with Prudent Utility Practices and Applicable Laws.

"<u>Safety Compliance Order</u>" means a written order from DOE to Holdings to implement Safety Compliance.

"Safety Event" means with respect to the Project Facilities (a) a material hazard, danger or other material risk to public or worker health or safety, (b) a material structural deterioration of a material portion of the Project or (c) material damage to a third party's Property or equipment.

"Sanctions" means economic or financial sanctions or trade embargoes or restrictive measures enacted, imposed, administered or enforced from time to time by (a) the United States government, including those administered by OFAC, the U.S. Department of State or the U.S. Department of Commerce, (b) the United Nations Security Council, (c) the European Union or any of its member states or (d) any other applicable Governmental Authority and including, for the avoidance of doubt, the Trading with the Enemy Act, the International Emergency Economic Powers Act, the Comprehensive Iran Sanctions, Accountability and Divestment Act and the Sudan Accountability and Divestment Act.

"Second Lien Collateral" has the meaning set forth in Section 11.6(a)(iii).

"Security Documents" means the security agreements, pledge agreements, financing statements, account control agreements or other instruments and documents that creates or purports to create or perfect a Lien on the Collateral in favor of DOE and, if applicable, the Intercreditor Agreement.

"Section 1222" has the meaning set forth in the recitals.

"Section 1222 Decision" has the meaning set forth in Section 6.1(a).

"Solvent" and "Solvency" mean, with respect to any Person on a particular date, that on such date (a) the fair value of the Property of such Person is greater than the total amount of liabilities, including contingent liabilities, of such Person, (b) the present fair salable value of the assets of such Person is not less than the amount that will be required to pay the probable liability of such Person on its debts as they become absolute and matured, (c) such Person does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's ability to pay such debts and liabilities as they mature and (d) such Person is not engaged in business or a transaction, and is not about to engage in business or a transaction, for which such Person's Property would constitute an unreasonably small capital. The amount of contingent liabilities at any time shall be computed as the amount that, in the light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

"SPP" has the meaning set forth in the recitals.

"SPP Facilities Study" means a "Facilities Study" as defined in the SPP OATT and applicable criteria and business practice documents related thereto.

"SPP OATT" means the SPP OATT on file with FERC and applicable criteria and business practice documents related thereto.

"SPP System Impact Study" means a "System Impact Study" as defined in the SPP OATT.

"Subsidiary" of any Person, means any corporation, partnership, joint venture, limited liability company, trust or estate of which (or in which) more than fifty percent (50%) of (a) the issued and outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time capital stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (b) the interest in the capital or profits of such partnership, joint venture or limited liability company or (c) the beneficial interest in such trust or estate, in each case, is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries or by one or more of such Person's other Subsidiaries.

"Subsidiary Guarantor" has the meaning set forth in Section 9.1(a).

"SWPA" has the meaning set forth in the recitals.

"<u>Taxes</u>" means all taxes, levies, imposts, duties, deductions, charges or withholdings imposed by any Governmental Authority, including any interest, penalties or additions thereto imposed in respect thereof.

"<u>Termination Date</u>" means the date on which this Agreement is terminated in accordance with Sections 7.1(a) or 7.1(b).

"<u>Texas Panhandle</u>" means the geographic area within the panhandle region of Texas, including Sherman, Ochiltree, and Hansford Counties.

"Threatened or Endangered Species" means any species listed by the United States Fish and Wildlife Service as threatened or endangered pursuant to the Endangered Species Act, as amended, 16 U.S.C. § 1531, *et seq.*, or any species listed as threatened or endangered pursuant to a state endangered species act.

"Title Defect" has the meaning set forth in Schedule 1 hereto.

"Title Search" has the meaning set forth in Schedule 1 hereto.

"TN Facilities" means those facilities developed by the Clean Line Parties in Tennessee.

"TRA" means the Tennessee Regulatory Authority.

"TRA 2015 Order" has the meaning set forth in Section 12.1(t).

"Transaction Documents" means this Agreement, the NERC Agreement, the Insurance Agreement, the Security Documents, any Performance Support and any other Contractual

Obligation entered into between DOE and any Clean Line Obligor from time to time in respect of the Project.

"<u>Transmission Services Agreement</u>" means a transmission services agreement under which the Clean Line Entities have agreed to provide transmission services using the Electrical Capacity owned by the Clean Line Parties.

"<u>TSA Precedent Agreement</u>" means a transmission services precedent agreement pursuant to which the Clean Line Entities and the counterparty agree to negotiate and enter into a Transmission Services Agreement.

"TVA" means the Tennessee Valley Authority.

"TVA Facilities Study" means a "Facilities Study" as defined in TVA's Transmission Service Guidelines and other applicable procedure documents related thereto.

"<u>TVA System Impact Study</u>" means a "System Impact Study" as defined in TVA's Transmission Service Guidelines and other applicable procedure documents related thereto.

"TX Facilities" means those facilities developed by the Clean Line Parties in Texas.

"<u>Uncontested Acquisition</u>" means any Acquisition by Condemnation instituted as a result of a request by any Landowner or Curative Party holding the applicable Project Real Estate Rights that such Project Real Estate Rights be acquired through condemnation. Determinations as to whether any Acquisition by Condemnation meets this definition of Uncontested Acquisition are to be made by DOE in its sole discretion and the undertaking of any Uncontested Acquisition prior to the satisfaction of the conditions precedent in Section 6.3 shall be at the sole discretion of DOE. DOE's further pursuit of Uncontested Acquisitions that subsequently are contested by the Landowner or Curative Party in a court of law will become subject to the satisfaction of those conditions precedent applicable to Acquisitions by Condemnation in Section 6.3.

"Uniform Act" has the meaning set forth in Schedule 1 hereto.

"Voluntary Land Acquisition" means (a) an acquisition of Project Real Estate Rights by DOE through an arm's length third party negotiated transaction or (b) at the sole option of DOE prior to the satisfaction of the conditions precedent set forth in Section 6.3, an Uncontested Acquisition.

"Waiver Parcel" has the meaning set forth in Schedule 1 hereto.

"WAPA" means the Western Area Power Administration.

"Wind-Up Event" has the meaning set forth in Section 7.5(a).

"Wind-Up Reserve Account" has the meaning set forth in Section 7.6.

"Work" means the development, design, engineering, construction, financing, operation, maintenance (including any Capital Repairs) and management of the Project, except for any obligations expressly contemplated by this Agreement to be performed by DOE.

"Ziff Family" means, collectively, (a) Dirk Ziff, Robert D. Ziff and Daniel M, Ziff, and their children and other lineal descendants, (b) the spouses or former spouses, widows or widowers of any of the Persons referred to in clause (a), (c) any (i) estate of one or more of the Persons listed in clauses (a) and (b) above or (ii) trust having as its sole beneficiaries one or more of the Persons listed in clauses (a) and (b) above and (d) any Person (other than any Clean Line Entity) the voting power of the outstanding ownership interests of which is Controlled by one or more of the Persons referred to in clauses (a), (b) and (c) above.

"Zilkha Family" means, collectively, (a) Michael Zilkha and his children and other lineal descendants; (b) the spouses or former spouses, widows or widowers of any of the Persons referred to in clause (a); (c) any (i) estate of one or more of the Persons listed in clauses (a) and (b) above or (ii) trust having as its sole beneficiaries one or more of the Persons listed in clauses (a) and (b) above; and (d) any Person (other than any Clean Line Entity) the voting power of the outstanding ownership interests of which is Controlled by one or more of the Persons referred to in clauses (a), (b) and (c) above.

### 1.2 Rules of Interpretation. In this Agreement, unless otherwise indicated:

- (a) any reference to this Agreement or any other Contractual Obligation means such agreement and all schedules, exhibits and attachments thereto as the same may be amended, supplemented or otherwise modified and in effect from time to time, and shall include a reference to any document that amends, modifies or supplements it, or is entered into, made or given pursuant to or in accordance with its terms;
- (b) each reference to any Applicable Law or Environmental Law shall be deemed to refer to such Applicable Law or Environmental Law as the same may be amended, supplemented or otherwise modified from time to time;
- (c) any reference to a Person in any capacity includes a reference to its permitted successors and assigns in such capacity and, in the case of any Governmental Authority, any Person succeeding to any of its functions and capacities;
- (d) references to days shall refer to calendar days unless Business Days are specified;
- (e) references to weeks, months or years shall be to calendar weeks, months or years, respectively;
- (f) the table of contents and section headings and other captions therein are for the purpose of reference only and do not affect the interpretation of this Agreement;
- (g) Article, Section and Schedule references within this Agreement are in reference to Articles, Sections and Schedules of this Agreement unless the context requires otherwise;

- (h) in the event of any conflict or inconsistency between any provisions contained in the documents comprising this Agreement, the Articles and Sections of this Agreement, as modified by any amendments or other modifications from time to time, shall take precedence over the Schedules and any other attachments to this Agreement;
- (i) defined terms in the singular shall include the plural and vice versa, and the masculine, feminine or neuter gender shall include all genders;
- (j) the words "hereof", "herein" and "hereunder", and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement;
- (k) the words "include," "includes" and "including" are deemed to be followed by the phrase "without limitation" unless the context specifically indicates otherwise:
- (l) words not otherwise defined herein that have well-known and generally accepted technical or trade meanings are used herein in accordance with such recognized meanings;
- (m) where the terms of this Agreement require that the approval, opinion, consent or other input of any Party be obtained, such requirement shall be deemed satisfied only where the requisite approval, opinion, consent or other input is given by or on behalf of the relevant Party in writing; and
- (n) any reference to "recitals" shall be a reference to the paragraphs immediately following the header of "recitals."

# ARTICLE II PROJECT OWNERSHIP STRUCTURE

### 2.1 Scope of the Clean Line Entities' Rights in Respect of the Project.

- (a) Pursuant to the terms of this Agreement (and subject in all respects to the terms and conditions hereof), the Clean Line Entities and DOE shall, at the sole cost and expense of the Clean Line Parties, undertake the Project.
- (b) From and after the Commencement Date, the Clean Line Entities and their authorized representatives (including any Contractors) shall have the right (subject to the other terms and conditions of the Transaction Documents and the Real Estate Rights Agreements) to enter into and use any DOE Acquired Real Property and the AR Facilities for purposes of carrying out the Project. Prior to the Commencement Date, the Clean Line Entities shall be solely responsible for gaining any access needed to any Project Real Estate Rights. Absent agreement by the Parties as to a later date, the Clean Line Entities' rights to enter into and use DOE Acquired Real Property or the AR Facilities shall automatically terminate on the Termination Date, and, on and after the Termination Date, all Project Facilities shall be removed at the Clean Line Entities' expense if so requested by DOE.

- (c) Subject to the foregoing and the other provisions of this Agreement, the Clean Line Parties shall not have any fee title, leasehold estate, possessory interest, permit, easement or other Real Estate Right of any kind in or to any DOE Acquired Real Property or in any of the AR Facilities. With respect to the AR Facilities and any DOE Acquired Real Property, the Clean Line Entities' interests under this Agreement shall be limited to contract rights constituting intangible personal property (and not real estate interests).
- (d) The Clean Line Entities' rights under this Agreement shall be subject in all respects to (i) DOE's ownership of any DOE Acquired Real Property and of the AR Facilities and (ii) all of DOE's rights and remedies under the Transaction Documents.
- (e) The Clean Line Entities' rights under this Agreement shall be subject in all respects to, and each of the Clean Line Entities shall be responsible for compliance with, the provisions of all Contractual Obligations (including any Real Estate Rights Agreement), Governmental Approvals or Governmental Orders pursuant to which DOE acquires ownership of any DOE Acquired Real Property, including paying all rents, Taxes, charges and filings fees in connection therewith on behalf of DOE; provided that to the extent that pursuant to any Applicable Law DOE has any non-delegable obligations or duties in respect of any undertakings under any Contractual Obligation (including any Real Estate Rights Agreement), Governmental Approval or Governmental Order relating to the Project, the obligations of the Clean Line Entities under this clause (e) shall be limited to (i) payment of all Covered Costs of DOE incurred in connection with such performance, (ii) providing access to DOE with respect to the Project Site and Project Facilities and (iii) otherwise using all commercially reasonable efforts to support and cooperate with DOE in order to enable DOE to perform any such non-delegable obligations and duties.
- or liable for any Indebtedness, losses, obligations or duties of any Clean Line Party or any other Person with respect to the Project, the TN Facilities, the TX Facilities or any other business or undertaking entered into or conducted by or on behalf of any Clean Line Party or the Project. All obligations to pay Project Costs, Taxes, assessments, insurance premiums, and all other fees, costs and expenses arising from or in connection with the Project (including the acquisition of Project Real Estate Rights) and all obligations to perform under any Contractual Obligation relating to the Project (other than DOE's obligations as expressly provided in this Agreement and in any other Transaction Document) shall be the sole responsibility of the Clean Line Parties.
- 2.2 <u>Ownership of Project Facilities</u>. Each of the Parties hereby agrees and acknowledges that:
  - (a) DOE shall own 100% of the AR Facilities.
  - (b) PECL OK shall own 100% of the OK Facilities. Except to the extent constituting a Permitted Disposition, PECL OK shall not Dispose of any of its rights or interests in the OK Facilities without DOE's prior written consent.

## 2.3 Rights to Electrical Capacity.

- (a) Pursuant to this Agreement, Holdings and/or any Clean Line Entity designated or nominated by Holdings, collectively, own 100% of the Electrical Capacity and have the right to market, use and sell transmission services relating to such Electrical Capacity pursuant to a Transmission Services Agreement or as otherwise permitted pursuant to Section 2.3(c), subject to FERC's open access transmission rules and policies.
- (b) All transmission and related services provided by any of the Clean Line Entities using any of the Project Facilities shall be provided in accordance with Applicable Laws and Prudent Utility Practices.
- (c) No Clean Line Entity shall Dispose of any of its rights or interests in any of the Electrical Capacity except:
  - (i) marketing and sales of transmission services using Electrical Capacity pursuant to Transmission Services Agreements as contemplated in this Agreement;
  - (ii) pledges or assignments of the Clean Line Entities' rights and interests in the Electrical Capacity and rights under any Transmission Services Agreements to which any Clean Line Entity is a party to the Project Financing Parties as collateral security for its obligations in respect of the Project Financing or to DOE as contemplated by Section 11.6;
  - (iii) sales or transfers of Electrical Capacity to Persons making an equity investment in any Clean Line Entity in an aggregate amount not to exceed, unless consented to by DOE, the lesser of (A) 500 MW (net) and (B) 20% of the net Electrical Capacity ("Permitted Project Investments"); provided that such Persons: (1) agree that the use of such Electrical Capacity shall be in accordance with FERC's open access transmission rules and policies applicable to such Person, Prudent Utility Practices and Applicable Laws, (2) make the representations, warranties and covenants set forth in Schedule 13 hereto for DOE's benefit, (3) agree not to transfer such Electrical Capacity to Prohibited Persons and (4) agree to the Release Provision; provided further that after giving effect to the applicable equity investment, (x) no Change of Control shall have occurred and (y) DOE shall have a fully perfected security interest in the Equity Collateral; and
  - (iv) other Dispositions consented to in writing by DOE in its sole discretion.
- (d) Subject to DOE's compliance with its obligations under this Agreement, in no event shall DOE or any other Covered Party under any circumstances have any liability to any Clean Line Party or any other Person in respect of any unavailability of any Electrical Capacity.

# ARTICLE III ACQUISITION OF PROJECT REAL ESTATE RIGHTS

3.1 <u>Generally</u>. The Project Real Estate Rights shall be those set forth in the Routing and ROW Plan as in effect from time to time and shall be acquired by the Parties in accordance with the terms of this Agreement.

## 3.2 <u>Clean Line Entities' Obligation to Acquire Project Real Estate Rights.</u>

- (a) The Clean Line Entities have the primary responsibility for acquiring all Project Real Estate Rights. In connection therewith, the Clean Line Entities shall use all commercially reasonable and good faith efforts to acquire all Project Real Estate Rights in accordance with the terms and conditions set forth in Schedule 1 hereto.
- (b) To the extent that any Clean Line Entity acquires any Project Real Estate Rights in Arkansas, such Clean Line Entity will grant to DOE, at no cost to DOE, a lease, sub-easement, right of way or other appropriate property interest or right of use in respect of such Project Real Estate Rights for all purposes of the development, design, engineering, construction, ownership, operation, maintenance and management of the AR Facilities. To the extent that any Clean Line Entity acquires any title insurance for Project Real Estate Rights in Arkansas or in respect of any other DOE Acquired Real Property, the Clean Line Entities shall use commercially reasonable efforts to name DOE as an additional insured in respect of such title insurance.
- (c) Solely for purposes of any exercise by DOE of any of its remedies upon the occurrence of an Operational EOD, each of the Clean Line Entities hereby grants to DOE and its designated replacement operator(s), at no cost to DOE or such replacement operator(s), a right of access and use in respect of any Real Estate Rights acquired by any of the Clean Line Parties in Oklahoma and in respect of the OK Facilities. Similarly, DOE agrees, upon the occurrence of an Operational EOD and exercise of the remedy described in Section 7.4(a)(v), to grant to DOE's designated replacement operator(s) a right of access and use in respect of any DOE Acquired Real Property and the AR Facilities for purposes of carrying out the Project.
- (d) Except for the grant to DOE of an interest in any Project Real Estate Rights acquired by the Clean Line Entities pursuant to the foregoing, the Clean Line Entities shall not Dispose of any of their respective rights or interests in any Project Real Estate Rights acquired by any of the Clean Line Entities (which excludes, for the avoidance of doubt, any DOE Acquired Real Property) without DOE's prior written consent; provided that no such consent shall be required for (i) any pledge or assignment to the Project Financing Parties as collateral security for Clean Line Entities' obligations under the Project Financing or to DOE (as contemplated by Section 11.6) or (ii) Project Real Estate Rights that are not necessary or are not reasonably likely to be necessary (A) for the development, construction or operation of the Project in accordance with Prudent Utility Practices, the Routing and ROW Plan, any Clean Line Document and the Project Plans as in effect from time to time or (B) for any of the Clean Line Parties to

perform its obligations under any Clean Line Document to which it is a party from time to time (including any Transmission Services Agreements).

# 3.3 <u>DOE's Acquisition of Project Real Estate Rights.</u>

- (a) Subject to the terms and conditions set forth in Schedule 1 hereto, Holdings shall be entitled to designate a Project Real Estate Right as a Real Estate Right to be acquired by DOE in the following circumstances (each, a "DOE Delegated Real Estate Right"):
  - (i) the Clean Line Entities have been unable, after using all commercially reasonable efforts and in compliance with their obligations hereunder, to locate any Landowner or Curative Party whose consent (or action) is necessary for a conveyance to the Clean Line Entities of a Project Real Estate Right in respect of any underlying Real Estate Right;
  - (ii) a Title Defect exists with respect to the underlying Real Estate Right and, notwithstanding the Clean Line Entities' compliance with their obligations hereunder and in Schedule 1 hereto, the Clean Line Entities have been unable to obtain any consent from or other necessary action by any Curative Party to enable the applicable Landowner to grant or convey a Project Real Estate Right to the Clean Line Entities in respect of the underlying Real Estate Right; or
  - (iii) the Clean Line Entities have been unable, after having otherwise complied with all of their obligations specified under Schedule 1 hereto, to acquire the applicable Project Real Estate Right.
- (b) At the sole cost and expense of the Clean Line Entities (and subject to Section 11.1) and subject to the satisfaction of the conditions precedent set forth below under Sections 6.2 and 6.3, as applicable, DOE shall assume responsibility for acquiring and shall acquire any DOE Delegated Real Estate Rights through a Voluntary Land Acquisition, Acquisition by Condemnation or through any other manner available to it under Applicable Law as contemplated herein on a prompt and timely basis (taking into account the fact that DOE's ability to promptly and timely acquire any such DOE Delegated Real Estate Rights may be subject to the actions of other third party Persons or Governmental Authorities). Subject to the foregoing sentence and DOE's compliance with its other obligations under this Agreement, the Clean Line Parties bear the risk of any time and cost impacts to the Project and Other Facilities related to DOE's acquisition of the DOE Delegated Real Estate Rights.
- (c) DOE shall not be required to enter into any Real Estate Rights Agreement relating to the acquisition of any DOE Delegated Real Estate Right that requires ongoing scheduled or regular payments from DOE after the payment of the initial consideration relating to the acquisition of the DOE Delegated Real Estate Right (which shall be funded using funds on deposit in, or credited to, the Advance Funding Account) or otherwise exposes DOE to any additional or ongoing payment obligations which are otherwise not fully funded in advance by a Clean Line Entity (x) under this Agreement prior to or

simultaneous with DOE's agreement to undertake such payment obligation or (y) pursuant to any subsequent Transaction Document entered into between a Clean Line Entity and DOE.

- (d) Notwithstanding the satisfaction (or lack of satisfaction) of the conditions precedent set forth in Sections 6.2 and 6.3, DOE agrees that promptly upon the Effective Date, and subject to receipt of adequate funding by the Clean Line Entities of the costs and expenses related thereto, it shall commence mobilization of personnel necessary to enable it to acquire DOE Delegated Real Estate Rights and set up procedures and processes for such acquisition such that, upon the satisfaction of the relevant conditions precedent, DOE shall be able to promptly pursue the acquisition of any such DOE Delegated Real Estate Rights through either Voluntary Land Acquisitions or by Acquisition by Condemnation; provided, that in no event shall DOE be obligated to commence actual acquisition or condemnation activities with respect to any DOE Delegated Real Estate Rights until the relevant conditions precedent have been satisfied.
- (e) The United States of America, acting through the Secretary of the Department, shall hold title to any and all DOE Acquired Real Property and the AR Facilities.
- (f) Without prejudice to DOE's rights to Dispose of any DOE Acquired Real Property or the AR Facilities in connection with an exercise of remedies following and during the occurrence of an Event of Default or after the Termination Date, DOE shall have the right to Dispose of its interest and title to all or any DOE Acquired Real Property or the AR Facilities to any other Person without the consent of any of the Clean Line Parties or any other Person (any such Disposition being a "DOE Instituted Disposition"); provided that the Clean Line Entities will not be responsible for the costs associated with any DOE Instituted Disposition; and provided further that any DOE Instituted Disposition:
  - (i) shall not occur prior to the earlier to occur of December 31, 2024 and Project Completion;
  - (ii) shall be subject to the Clean Line Entities' continued right of use in respect of such Project Real Estate Rights and the AR Facilities as provided in Section 2.1;
  - (iii) shall be subject to the continued right of use of the Electrical Capacity by Holdings or any other Person that holds rights to use such Electrical Capacity;
  - (iv) shall not be prohibited under the terms of the DOE Direct Agreement and the Intercreditor Agreement or otherwise shall have been consented to by any applicable Project Financing Parties party thereto; and
  - (v) shall have no materially adverse impact on any Clean Line Entities' material rights and material benefits under this Agreement or any other Transaction Document (including the Clean Line Entities' ability to secure on a

- commercially reasonably basis any necessary waivers, approvals or consents from DOE as required under the terms of this Agreement).
- 3.4 <u>Cost Responsibility for Acquisition of Project Real Estate Rights</u>. The acquisition of all Project Real Estate Rights shall be at the sole cost and expense of the Clean Line Entities in accordance with Sections 11.1 and 11.3.
- Amendments and Modifications to Routing and ROW Plan. Holdings shall 3.5 promptly notify DOE of any material proposed amendments or material modifications to the Routing and ROW Plan (including any updates to the planned routing for the AC Collection System) and provide a description of the reasons underlying such material proposed amendments or material modifications along with such other information as DOE may request in respect of such material proposed amendment or material modification. Any amendment or modification to the Routing and ROW Plan that could reasonably be expected to (a) materially and adversely affect: (i) the ability of the Clean Line Parties' to perform their respective obligations under any Clean Line Document then in effect or (ii) the construction or operation of the Project in accordance with the terms of the Project Plans, the Clean Line Documents and the Project Financing Documents, in each case, as then in effect, (b) result in an Event of Default under this Agreement or the other Transaction Documents or (c) materially increase (i) the Project Real Estate Rights reasonably anticipated to be DOE Delegated Real Estate Rights or (ii) DOE's obligations or liabilities in respect of any DOE Delegated Real Estate Rights or the AR Facilities, shall require the consent of the Coordination Committee. Each amendment or modification to the Routing and ROW Plan shall be made in material compliance with all Applicable Laws, including all Environmental Laws, Cultural Resource Agreements and all measures adopted in the DOE Mitigation Action Plan.

# ARTICLE IV DEVELOPMENT, CONSTRUCTION, OPERATION AND MAINTENANCE OF THE PROJECT

### 4.1 Development, Construction, Operation and Maintenance of the Project Generally.

- (a) Subject to the oversight of the Coordination Committee and DOE's obligations with respect to the acquisition of Project Real Estate Rights pursuant to the terms of this Agreement, the Clean Line Entities have sole responsibility for the management of all aspects of the Project, including the day-to-day management of the Project, the administration of all Project Contracts and the performance of all of the Work.
- (b) The Clean Line Entities hereby agree to perform or cause to be performed, all development, design, engineering, construction, operation, maintenance and management activities appropriate for the development of the Project in accordance with the Clean Line Documents (as in effect from time to time), the Project Plans, the Required Approvals and Prudent Utility Practices. As between DOE and the Clean Line Entities, the Clean Line Entities bear the risk of (i) any incorrect or incomplete review, examination or investigation by the Clean Line Entities of any of the Project Real Estate Rights or the Project Site and surrounding locations and (ii) any incorrect or incomplete

information resulting from the development, design, engineering, construction, financing, operation, maintenance, management, replacement and decommissioning activities conducted by the Clean Line Entities or any other Person in connection with the Work, the Project and the Other Facilities.

- (i) DOE does not, and shall not be required to, make any warranty or representation as to any surveys, data, reports or other information provided by DOE or other Persons concerning surface conditions and subsurface conditions, including the presence of Hazardous Substances, contaminated groundwater, archeological, paleontological and cultural resources and Threatened or Endangered Species that might affect any of the Project Real Estate Rights or the Project Site; and (ii) as between DOE and the Clean Line Entities, the Clean Line Entities bear the risk of all conditions occurring on, under or at the Project Site or in connection with any of the Project Real Estate Rights, including: (A) physical conditions, (B) changes in surface topography, (C) variations in subsurface moisture content, (D) the presence or discovery of Hazardous Substances, including contaminated ground water, (E) the discovery at, near or on any of the Project Real Estate Rights of any archeological, paleontological or cultural resources and (F) the discovery at, near or on the Project Real Estate Rights of any Threatened or Endangered Species; provided that, subject to Section 2.1(e), the foregoing does not alter or excuse DOE's non-delegable obligations and responsibilities under the DOE Mitigation Action Plan, any Cultural Resource Agreement under NHPA or the Endangered Species Act or any other Applicable Law, which shall in all circumstances remain the obligation and responsibility of DOE.
- (d) All Material Construction Contracts and Material O&M Agreements shall provide that DOE is a third party beneficiary thereof. Subject to agreement of the applicable Project Participant (which the Clean Line Entities shall use all commercially reasonable efforts to secure), DOE may, at its election prior to the execution of the applicable Material Construction Contract or Material O&M Agreement, become a party thereto for purposes of obtaining the benefit of any applicable warranties, indemnities and relevant protections thereunder, without any liability thereunder except as expressly assumed by DOE. Holdings shall deliver to DOE, at least ten (10) Business Days prior to the execution of any such Material Construction Contract or Material O&M Agreement, a final draft of such Material Construction Contract or Material O&M Agreement, as the case may be, so as to permit DOE to exercise the option referenced in the preceding sentence (if available following exercise of commercially reasonable efforts by the Clean Line Entities). Each Material Project Contract shall be executed by at least one of the Clean Line Parties. No waiver, approval or change to a Material Project Contract that has, or could reasonably be expected to have, an Adverse DOE Impact, shall be made without the approval of the Coordination Committee.
- (e) The Clean Line Entities hereby agree to retain or cause to be retained only Contractors that are qualified, experienced and capable in the performance of the portion of the construction, operation or maintenance of the Project to be performed by such Contractor. Each of the Clean Line Entities shall contractually require that each such Contractor has, at the time of the execution of any Construction Contract or O&M Agreement, and maintains at all time during performance thereunder, all Governmental

Approvals required by Applicable Law. The retention of Contractors by any Clean Line Entity shall not relieve any such Clean Line Entity from any of its responsibilities under this Agreement or any other Transaction Document.

- (f) In the performance of its obligations under this Agreement and the other Transaction Documents, each of the Clean Line Entities shall at all times comply, and contractually require that all Contractors comply, with all Applicable Laws (including with respect to the applicable contracts for construction, as defined in Department of Labor regulations at 29 C.F.R. § 5.2(j), the Davis-Bacon Requirements to the extent that DOE (or the Department of Labor, as the case may be) has determined that the Davis-Bacon Act is applicable to this Agreement and/or the Project), and all other Applicable Laws relating to labor, occupational safety and health standards, rules, regulations and federal and state orders and Environmental Laws. DOE shall make a determination or request that the Department of Labor make a determination as to the applicability of the Davis-Bacon Act to this Agreement and/or the Project no later than April 30, 2016. If DOE requests review by the Department of Labor, DOE will make any such request by April 30, 2016, and DOE shall provide Clean Line with an update by May 31, 2016 on the status of such review by the Department of Labor.
- (g) Without prejudice to the Clean Line Parties' obligations in respect of the payment of Covered Costs and Covered Liabilities and subject to DOE's obligations in respect of Section 4.10, in undertaking their respective obligations as set forth in this Agreement to develop, construct, operate and maintain, as applicable, the Project, DOE and the Clean Line Entities agree to take all steps necessary to comply in all material respects with all commitments for compliance with all Applicable Laws (including Environmental Laws) and Cultural Resource Agreements, including performance of any required measures set forth in the DOE Mitigation Action Plan; provided that with respect to DOE, its undertakings under this clause (g) shall only apply to the extent of any non-delegable obligation or responsibility of DOE under the DOE Mitigation Action Plan, any Cultural Resource Agreement under NHPA or the Endangered Species Act or any other Applicable Law.
- (h) Each of the Clean Line Entities shall, at its own cost and expense, comply in all material respects with all conditions imposed by and undertake all actions required by and all actions necessary to maintain in full force and effect all Required Approvals, including performance of any required measures set forth in the DOE Mitigation Action Plan.
- (i) No Clean Line Party shall enter into any agreement with any Project Participant, Governmental Authority (excluding DOE), Landowner or any other third Person having regulatory jurisdiction over any aspect of the Project or having any Property interest affected by the Project that in any way purports to obligate DOE, or states or implies that DOE has an obligation, to such Person to carry out any installation, design, construction, maintenance, repair, operation, control, supervision, regulation or other activity related to the Project, unless DOE otherwise approves in writing in its sole discretion. No Clean Line Party has any power or authority to enter into a Contractual Obligation in the name of or on behalf of DOE unless expressly authorized by DOE.

DOE agrees and acknowledges that the Clean Line Parties shall enter into the Material Project Contracts from time to time for purposes of the design, development, engineering, ownership, operation, management and maintenance of the Project.

# 4.2 DOE Mitigation Action Plan.

- (a) Following issuance of the Section 1222 Decision, DOE shall prepare a plan pursuant to 10 C.F.R. § 1021.331 (the "<u>DOE Mitigation Action Plan</u>") that explains how the mitigation measures in the Section 1222 Decision will be planned and implemented and addresses the following:
  - (i) mitigation commitments concerning Environmental Laws and Cultural Resource Agreements identified in the Section 1222 Decision;
  - (ii) any environmental protection measures, species-specific protection measures and best management practices identified in the Final Environmental Impact Statement;
  - (iii) reasonable and prudent measures or implementing terms and conditions set forth in the Biological Opinion; and
  - (iv) any conditions and procedures included in any Cultural Resource Agreement.
- (b) The DOE Mitigation Action Plan shall be prepared before DOE takes any action directed by the Section 1222 Decision that is the subject of a mitigation commitment.
- (c) The DOE Mitigation Action Plan shall be as complete as possible and commensurate with the information available regarding the course of action directed by the Section 1222 Decision. DOE may revise the DOE Mitigation Action Plan as more specific and detailed information becomes available, including to address any modified terms and conditions issued in connection with any Environmental Laws and the Cultural Resource Agreement.
- (d) The DOE Mitigation Action Plan will be available on the DOE NEPA Website (http://www.energy.gove/nepa/) and on the Plains and Eastern EIS website (www.plainsandeasterneis.com). Pursuant to 10 C.F.R. § 1021.331, DOE shall make copies of the DOE Mitigation Action Plan available for inspection in the appropriate DOE public reading room(s). Copies of the DOE Mitigation Action Plans shall also be available upon written request.
- (e) Holdings shall promptly inform DOE when more specific and detailed information becomes available that should be incorporated into the DOE Mitigation Action Plan, including enforceable commitments included in any Government Approvals for construction, operation and maintenance of the Project.

4.3 Amendments and Modifications to the Project Plans. Holdings shall promptly notify DOE of any proposed material amendments or material modifications to the Project Plans and provide a description of the reasons underlying such proposed material amendments or material modifications along with such other information as DOE may request in respect of such proposed material amendment or material modification. Any amendment or modification to the Project Plans that results in the anticipated Electrical Capacity being reduced by more than 1,500 MW (gross) or 1,000 MW (net) in the aggregate shall require the prior approval of the Coordination Committee. Further, the Clean Line Parties shall take all actions required under the Interconnection Agreements prior to making material modifications to the Project, for which prior notice, consultation, review or approval may be required by the applicable Interconnection Agreement.

## 4.4 Construction Contracts and Project Contracts.

- Unless otherwise determined by Holdings in its reasonable judgment to be beneficial to the Project (as notified by Holdings to DOE in writing prior to the signing of any of the Material Construction Contracts), the Material Construction Contracts for the Project Facilities shall (i) be lump sum, fixed price contracts, (ii) be in the nature of "turnkey" contracts concerning the works covered thereunder, (iii) contain provisions relating to guaranteed performance levels, guaranteed completion dates and liquidated damages that are consistent with current market practice for the construction of transmission facilities in the United States and (iv) require each Construction Contractor to provide credit support for its obligations under the applicable Material Construction Contract in the form of a creditworthy parent guarantee, bond, retainage or letter of credit or some combination of the forgoing. Each Material Construction Contract with respect to the AR Facilities shall provide that title to all Works covered thereunder will be transferred to DOE as completed, and upon payment by the Clean Line Entities of all amounts due and payable under such Material Construction Contract, all such Contractors shall waive any claims thereto or Liens thereon (to the maximum extent permitted by Applicable Law) and the Clean Line Entities shall advance funds to DOE for all Taxes DOE must pay as a result thereof.
- (b) The Clean Line Entities bear sole responsibility to pay all fees, expenses, Taxes, assessments, insurance premiums, indemnification claims and other amounts under the Construction Contracts and other Project Contracts, and DOE shall not be in any way responsible or liable for any payments, losses, obligations or duties under or in respect of the Construction Contracts or any other Project Contracts.
- (c) Each Construction Contractor shall be required to maintain and pay for customary insurance policies for such Construction Contract, including (if not obtained by the Clean Line Entities) builder's all-risk, delayed start-up, general and automobile liability, employer's liability, workers' compensation and excess liability coverages, as applicable, and that are otherwise consistent with the Insurance Agreement unless otherwise approved by the Coordination Committee.
- 4.5 <u>Interconnection Agreements</u>. The Project Facilities shall be interconnected to the electric transmission systems operated by SPP and MISO and the TN Facilities will be

interconnected to the electrical transmission system operated by TVA, in each case pursuant to Interconnection Agreements that, among other things, provide for interconnection sufficient to allow the Clean Line Parties to safely and reliably deliver energy across the Project Facilities up to the Electrical Capacity and also satisfy their obligations under the Transmission Services Agreements.

- 4.6 Operational Coordination with SPP, MISO and TVA. At such time as required by Applicable Law, but in any event no later than Project Completion, the Clean Line Parties shall enter into one or more agreements with SPP, MISO and TVA, as applicable, regarding the coordinated operation of the Project with SPP, MISO and TVA, which shall include identification of the entity responsible for exercising operational control of the Project as well as any agreements with respect to any inter-balancing area interchange of energy or ancillary services between the Project and the neighboring control areas (each, an "Operating Agreement"). Holdings shall consult with and report to DOE on the development of such Operating Agreements. Holdings shall provide DOE with a final draft of any such Operating Agreement at least ten (10) Business Days prior to the execution thereof by the Clean Line Parties; provided that such draft may redact or exclude such data or other information the disclosure of which is prohibited by Applicable Law; provided that the Clean Line Entities shall use all commercially reasonable efforts to apply for any consent or exemption that may be available under Applicable Law or from any Person or Governmental Authority for purposes of providing any such data or other information to DOE and shall, promptly upon receipt of such consent or exemption, provide DOE with an unredacted copy of such Operating Agreement. The Clean Line Parties have sole responsibility in respect of the execution, delivery and performance of each Operating Agreement, which may include delegating performance responsibilities to qualified third parties consistent with Prudent Utility Practices.
- 4.7 <u>Maintenance of the Project Facilities</u>. The Clean Line Entities have sole responsibility for engaging experienced and responsible Contractors to operate, maintain and repair the Project Facilities to a standard not less than Prudent Utility Practices and in accordance with Applicable Law and Required Approvals, and if such standard is not met then DOE may, subject to the terms of the DOE Direct Agreement and the Intercreditor Agreement to the extent applicable, direct the Clean Line Entities to terminate any applicable O&M Agreement(s) of each applicable Contractor in accordance with its terms. Except with the consent of the Coordination Committee, each Material O&M Agreement shall not (a) require a payment of a bonus or a fee materially in excess of expected bonus or fee levels for comparable contracts payable on a third party arms' length basis or (b) cap the liability of such Contractor at less than all fees (excluding cost reimbursement) received by it under such Material O&M Agreement.

# 4.8 <u>Capital Repairs and Reserve Account.</u>

- (a) The Clean Line Entities shall perform, or engage Contractors to perform, all Capital Repairs necessary or advisable in accordance with Prudent Utility Practices in connection with the Project. The Clean Line Entities shall be solely responsible for the costs and expenses of any such Capital Repairs.
- (b) On and after Project Completion, Holdings shall establish and maintain at all times a capital repairs and maintenance reserve account (the "Capital Repairs Reserve

Account"), which Capital Repairs Reserve Account shall be funded at all times with an amount sufficient to cover all estimated Capital Repairs in respect of the Project, *plus* a reasonable contingency amount as determined by the Independent Engineer appointed by the Project Financing Parties (or if there is no Project Financing, the Coordination Committee in consultation with the Independent Engineer), which are required by Prudent Utility Practice or reasonably anticipated to be incurred in the upcoming twelve (12) months. If at any time the Clean Line Entities fail to utilize such funds in the Capital Repair Reserve Account to make Capital Repairs when required, then subject to prior notice to Holdings and a grace period of thirty (30) days, DOE may, at its option (but with no obligation to) draw (i) on the Performance Support or (ii) if agreed to by the Project Financing Parties, from the Capital Repairs Reserve Account, and, in each case, apply the proceeds thereof to the making of any such Capital Repairs.

- 4.9 NERC Standards. Prior to the issuance of the Notice to Proceed, the Clean Line Entities and DOE shall enter into an agreement (the "NERC Agreement") pursuant to which (a) the Clean Line Entities shall assume sole responsibility for compliance with all applicable or desirable reliability standards (including NERC reliability standards) related to the Project, including any related documentation obligations, audits, violations and mitigation obligations, (b) the Clean Line Parties shall be solely responsible for all liabilities or claims that arise in connection with the operation of the Project (or any portion thereof) as a result of the noncompliance of the Project (or any portion thereof) with NERC's reliability standards and (c) the Clean Line Parties shall indemnify DOE and each Covered Party for all Covered Liabilities in connection with the operation of the Project (or any portion thereof) as a result of the Project's non-compliance with all applicable reliability standards or regulations.
- 4.10 <u>DOE Cooperation</u>. To the extent reasonably requested by Holdings, DOE shall coordinate and cooperate in good faith with the Clean Line Entities on the Project, including providing information and assistance in the preparation of any application for any Required Approval; <u>provided</u> that such cooperation and coordination shall be at the Clean Line Entities sole cost and expense and shall not impose an unreasonable burden on DOE.

# ARTICLE V COORDINATION COMMITTEE

### 5.1 Coordination Committee.

- (a) Holdings and DOE will establish a coordination committee promptly after the date of this Agreement (the "Coordination Committee"), which shall be composed of two (2) representatives from Holdings and two (2) representatives from DOE. Each of Holdings and DOE may replace its respective representatives at any time by providing written notice to the other Person. The Coordination Committee shall coordinate and manage the efforts of the Clean Line Entities and DOE relating to the Project and provide a forum for updates, discussion and attempted resolution of any relevant issues with respect to the Transaction Documents and the Project.
- (b) Prior to the occurrence of Project Completion, the Coordination Committee shall meet not less than once a month, and from and after the occurrence of

Project Completion, the Coordination Committee shall meet not less than once a fiscal quarter, in each case at mutually convenient times, locations or means as the Coordination Committee shall determine. The Coordination Committee will have the authority to create sub-committees to consider specific issues whenever it deems appropriate. Each of Holdings and DOE shall have the right to call a special meeting of the Coordination Committee upon not less than five (5) Business Days' prior written notice to the other Person. One (1) of Holdings' representatives will be designated as the Chair of the Coordination Committee. Holdings and DOE may submit any item for inclusion on any agenda of any Coordination Committee meeting.

- (c) Subject to Section 7.4(a), meetings of the Coordination Committee shall require a quorum consisting of one representative of each of Holdings and DOE. If a quorum is not present at the commencement of any meeting, the Chair will reschedule the meeting to take place within the following ten (10) Business Days and will give notice of such scheduled meeting to the representatives on the Coordination Committee.
- (d) Other employees and/or agents of the Parties shall be entitled to attend meetings of the Coordination Committee. Meetings may be conducted in person, by telephone or video conference call or by such other means as which permits the Parties' representatives to be verified and to hear and be heard by the other Parties' representatives. Attendees who are not representatives of any Party shall be identified at the commencement of any meeting and shall have no power to vote on any matters but may participate in discussions in accordance with the Coordination Committee's rules of order, which may limit the amount of time that such other attendees may participate.
- (e) Notwithstanding the delegation of authority granted to the Parties pursuant to this Agreement, and subject to Section 7.4(a), the following actions shall require the affirmative approval of one (1) representative of each of Holdings and DOE on the Coordination Committee:
  - (i) the approval of any public announcements relating to DOE's involvement in the Project;
  - (ii) the adoption, implementation and/or material modification of an insurance agreement (the "<u>Insurance Agreement</u>") for the Project and the making of any material claim in respect of any insurance relating to the Project;
  - (iii) the estimation of costs required to be funded into the Wind-Up Reserve Account, any matters relating to the funding of the Wind-Up Events, the decision to commence the Wind-Up Events and the entry into of any Contractual Obligations or undertakings relating to the Wind-Up Events;
  - (iv) if no Project Financing is then currently in effect, the issuance of any completion or similar certificate or the acceptance of any performance tests under any Material Construction Contract;

- (v) the determination from time to time of the amount of any Remaining DOE Acquisition Costs and to the extent applicable, the Contingency Amount; and
- (vi) any express consents or approvals delegated to the Coordination Committee under this Agreement, including pursuant to Sections 3.5, 4.1(d), 4.3, 4.4(c), 4.7, 4.8(b), 7.3(m), 7.6 and 7.7.
- (f) If the representatives of Holdings and DOE participating in a meeting of the Coordination Committee are unable to reach an agreement on a matter before the Coordination Committee (a "<u>Deadlock</u>"), Holdings and DOE shall attempt to resolve such Deadlock through negotiations of the representatives. If such Deadlock is not resolved within seven (7) days, the Deadlock shall be referred to a panel consisting of a senior level executive of each of Holdings and DOE with the authority to resolve the matter causing such Deadlock, who shall attempt to resolve such Deadlock within seven (7) days. For construction-related, operational-related or other technical issues or for financial or accounting issues, Holdings and DOE shall have the right to appoint an independent technical or financial consultant to assist in resolving such Deadlock.
- (g) DOE shall have the right to retain one or more technical (including engineering, market, legal or financial) consultants with respect to its participation on the Coordination Committee at the sole cost and expense of the Clean Line Entities.
- (h) All costs and expenses incurred by the representatives of DOE in the Coordination Committee shall be borne by the Clean Line Entities.

# ARTICLE VI CONDITIONS PRECEDENT

- 6.1 <u>Conditions Precedent to Effective Date.</u> DOE's obligations hereunder shall become effective upon the satisfaction of the following conditions:
  - (a) the Secretary of the Department shall have issued a Record of Decision (the "Section 1222 Decision") authorizing the participation by DOE in the Project pursuant to the statutory authority granted under Section 1222 and addressing all required determinations necessary for purposes of the participation decision under NEPA, the Endangered Species Act, the NHPA, and any other Applicable Law;
    - (b) each of the Parties shall have duly executed and delivered this Agreement;
  - (c) Holdings shall have delivered to DOE (i) certified Organizational Documents of each of the Clean Line Parties, (ii) secretary's certificates, officer's certificates, resolutions and good standing certificates for each of the Clean Line Parties (including certificates certifying to such matters as DOE shall reasonably require) and (iii) legal opinions from counsel to the Clean Line Parties;

- (d) at least thirty (30) days shall have passed since the Environmental Protection Agency shall have published a notice of availability for the Final Environmental Impact Statement in respect of the Project;
- (e) Holdings shall have delivered to DOE certified copies of duly executed term sheets for TSA Precedent Agreements in respect of not less than 3,500 MW of the Electrical Capacity in the aggregate, each of which shall be in full force and effect;
- (f) the Clean Line Entities shall be in compliance with all funding obligations required under the AFDA;

## (g) [Reserved]

- (h) Holdings shall have delivered to DOE (i) an updated Project Budget and a reasonably detailed project budget for the development, design, engineering and construction of the Other Facilities and (ii) an updated base case model of projections of operating costs and results (including projections in respect of revenues, expenses, cash flow, debt service and sources and uses of revenues) for the Project and for the Other Facilities (the "Base Case Projections");
- (i) Holdings shall have delivered to DOE a copy of audited consolidated financial statements of CLEP for the calendar year ending December 31, 2014 and unaudited consolidated financial statements of CLEP for each of the four (4) fiscal quarters of the calendar year ending December 31, 2015;
- (j) Holdings shall have delivered to DOE a Project Development Progress Report as of the Effective Date;
- (k) Holdings shall have paid all costs and expenses (including costs and expenses of all consultants, advisors and counsel to DOE) accrued and invoiced;
- (l) all representations and warranties made by any of the Clean Line Obligors in this Agreement shall be true and correct as of the Effective Date;
  - (m) no Default or Event of Default shall have occurred and be continuing;
- (n) (i) no Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (A) DOE's performance under this Agreement or (B) DOE's participation in the Project and (ii) no other final and non-appealable Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (A) the execution or delivery of this Agreement or (B) any Clean Line Entity's performance under this Agreement; and
- (o) (i) ACL shall have been duly formed and organized, (ii) all Property or physical facilities held by or in the name of any of Holdings or any of its Subsidiaries to the AR Facilities or any Project Real Estate Rights located in Arkansas shall have been transferred to ACL and documentary evidence thereof shall have been delivered by

Holdings to DOE and (iii) neither PECL nor any PECL Subsidiary (to the extent then in existence) shall thereafter own directly or control real Property of the Project or any physical facilities of the Project Facilities (excluding rights under any Project Contracts where multiple Clean Line Parties are parties).

### 6.2 <u>Conditions Precedent to Voluntary Land Acquisitions.</u>

- (a) DOE's initial obligation to assist with the acquisition of Project Real Estate Rights by way of Voluntary Land Acquisitions shall commence upon the satisfaction of the following conditions precedent:
  - (i) the Effective Date shall have occurred;
  - (ii) the Clean Line Entities shall have complied with all of the requirements and procedures set forth in Schedule 1 hereto with respect to the DOE Delegated Real Estate Right to be acquired;
  - (iii) Holdings shall have delivered to DOE (A) the Routing and ROW Plan, (B) the Project Plans and (C) an updated Project Development Progress Report as of the Commencement Date and an updated Project Budget, which shall include a summary and explanation of any deviations from the Project Budget and the Project Schedule delivered as a condition to the occurrence of the Effective Date;
  - (iv) Holdings shall have delivered to DOE certified copies of duly executed and enforceable TSA Precedent Agreements, Acceptable Transmission Services Agreements or Acceptable Permitted Project Investment Commitments in respect of not less than 3,500 MW of the Electrical Capacity in the aggregate, each of which shall be in full force and effect; provided that no less than 1,500 MW of such Electrical Capacity in the aggregate (calculated as the sum of (A) with respect to Acceptable Permitted Project Investments, the sum of each portion of the Electrical Capacity transferred (and for which the Clean Line Entities have received payment or will receive payment within three (3) years after the date of such Permitted Project Investment) pursuant to each Acceptable Permitted Project Investment Commitment and (B) with respect to the Acceptable Transmission Services Agreements, the sum of the average Electrical Capacity committed in the initial five (5) years of the term for each such Acceptable Transmission Services Agreement) shall be committed pursuant to Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments;
  - (v) Holdings shall have delivered to DOE copies of duly executed purchase options or comparable site control agreements (collectively, the "Converter Station Real Estate Rights Agreements") that permit the Clean Line Entities to obtain all Real Estate Rights necessary for the construction of the Converter Station Facility and the Intermediate Converter Station, and such purchase options, if any, shall have exercise periods consistent with the Project Schedule, each of which shall be in full force and effect;

- (vi) the Clean Line Parties shall be diligently proceeding with obtaining all necessary interconnection rights for the Project, including completion of the following conditions:
  - (A) SPP the SPP Facilities Study and the SPP System Impact Study shall have been completed and the Clean Line Parties shall have executed an Interconnection Agreement for interconnection of the Project with the SPP-controlled transmission system;
  - (B) MISO the MISO Interconnection Feasibility Study, the MISO Interconnection Facilities Study and the MISO Interconnection System Impact Study shall have been completed and the Clean Line Parties shall have executed an Interconnection Agreement for interconnection of the Project with the MISO-controlled transmission system; and
  - (C) TVA the TVA Facilities Study and the TVA System Impact Study shall have been completed and the Clean Line Parties shall have executed a Project Work Agreement or Interconnection Agreement for interconnection of the TN Facilities with the TVA transmission system;
- (vii) Holdings shall have delivered to DOE certified copies of (x) Firm Project Equity Commitments that are then in full force and effect and that provide for commitments (together with amounts on deposit in the Advance Funding Account) in an amount equal to not less than 150% of the Remaining DOE Acquisition Costs, as of the Commencement Date and (y) Project Equity Commitments or letters of interest and/or Project Financing Commitments or letters of interest in respect of the Project Financing (together with amounts on deposit in the Advance Funding Account) in an aggregate amount sufficient to cover all other Remaining Project Costs;
- (viii) Holdings shall have delivered to DOE the required Performance Support in an amount not less than the Applicable Amount;
- (ix) Holdings shall have completed the following design, engineering and project management activities and delivered evidence thereof to DOE:
  - (A) obtained design criteria, structure geometrics, structure loading schedules and estimated weights from vendors;
  - (B) selected the insulator and hardware vendor and completed electrical testing specifications;
  - (C) completed LiDAR survey, structure spotting, preliminary access road layout and vegetation clearing assessment;

- (D) completed the conductor, metallic return conductor and optical ground wire/shield wire design;
  - (E) completed preliminary foundation design; and
- (F) prepared a reasonably detailed project execution and construction schedule.
- (x) DOE and Holdings shall have entered into the Insurance Agreement and Holdings shall have delivered to DOE evidence that all Required Insurance is in full force and effect (including written binding verification of such coverage);
- (xi) (A) no Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (1) DOE's performance under this Agreement or any other Transaction Document then in effect or (2) DOE's participation in the Project and (B) no other final and non-appealable Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, (1) sets aside, enjoins or legally prohibits any Clean Line Entity's performance under this Agreement or any other Transaction Document then in effect or any Clean Line Entity's performance in any material respect under any Material Project Contract to which it is a party or (2) affects in any material respect the legality, validity or enforceability of any of the Transaction Documents, any Project Equity Commitment or any of the Material Project Contracts then in effect;
- (xii) (A) all representations and warranties made by any Clean Line Obligor in any of the Transaction Documents shall be true and correct in all material respects (except to the extent any such representation and warranty itself is qualified by "materiality", "material adverse effect", "Adverse DOE Impact", "Clean Line Material Adverse Effect" or a similar qualifier, in which case it shall be true and correct in all respects) and (B) no Default, Event of Default or Event of Loss shall have occurred and be continuing, in each case, as of the Commencement Date:
- (xiii) the Clean Line Entities shall have granted to DOE a first priority perfected security interest in the Equity Collateral as required at such time pursuant to Section 11.6, together with such legal opinions, certificates and other documents in respect thereof as DOE may reasonably request; and
- (xiv) Holdings shall have delivered to DOE a certificate of an Authorized Officer as to the satisfaction of the foregoing conditions precedent.
- (b) After the occurrence of the Commencement Date, DOE's obligations to acquire or continue to acquire DOE Delegated Real Estate Rights through Voluntary Land Acquisitions or any other means shall only be subject to the following conditions being satisfied:

- (i) the Clean Line Entities shall have complied with all of the requirements and procedures set forth in Schedule 1 hereto with respect to the DOE Delegated Real Estate Rights to be acquired;
- (ii) Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments in respect of not less than 1,500 MW of the Electrical Capacity in the aggregate (calculated as the sum of (A) with respect to Acceptable Permitted Project Investments, the sum of each portion of the Electrical Capacity transferred (and for which the Clean Line Entities have received payment or will receive payment within three (3) years after the date of such Permitted Project Investment) pursuant to each Acceptable Permitted Project Investment Commitment and (B) with respect to the Acceptable Transmission Services Agreements, the sum of the average Electrical Capacity committed in the initial five (5) years of the term for each such Acceptable Transmission Services Agreement) shall be in full force and effect and no event shall have occurred and be continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Acceptable Transmission Services Agreement or such Acceptable Permitted Project Investment Commitment;
- (iii) the Converter Station Real Estate Rights Agreements shall be in full force and effect or the Clean Line Entities shall own in fee free and clear of all Liens other than Permitted Liens all Real Estate Rights necessary for the construction of the Converter Station Facility and the Intermediate Converter Station;
- (iv) the executed Interconnection Agreements for interconnection of the Project with the SPP-controlled transmission system and the MISO-controlled transmission system and the executed Project Work Agreement or Interconnection Agreement for interconnection of the TN Facilities with the TVA transmission system shall be in full force and effect and no event shall have occurred and be continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Interconnection Agreements or Project Work Agreement (except to the extent the Project Work Agreement has been replaced by an Interconnection Agreement with TVA);
- (v) either (A) the Project Equity Commitments (including Firm Project Equity Commitments that are then in full force and effect and that provide for commitments (together with amounts on deposit in the Advance Funding Account) in an amount equal to not less than 150% of the Remaining DOE Acquisition Costs as of any date on which any Clean Line Entity designates any Project Real Estate Right as a DOE Delegated Real Estate Right), Project Financing Commitments and any letters of intent delivered as a condition to the occurrence of the Commencement Date shall continue to be in full force and effect or (B) the Financing Condition shall be satisfied;

- (vi) (A) no Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (1) DOE's performance under this Agreement or any other Transaction Document then in effect or (2) DOE's participation in the Project and (B) no other final and non-appealable Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, (1) sets aside, enjoins or legally prohibits any Clean Line Entity's performance under this Agreement or any other Transaction Document then in effect or any Clean Line Entity's performance in any material respect under any Material Project Contract to which it is a party or (2) affects in any material respect the legality, validity or enforceability of any of the Transaction Documents, any Project Equity Commitment or any of the Material Project Contracts then in effect; and
  - (vii) no Event of Default shall have occurred and be continuing.
- (c) Following the Commencement Date, at any time Holdings delivers a written notice of designation of any Project Real Estate Rights as a DOE Delegated Real Estate Right, Holdings shall concurrently deliver a certificate of an Authorized Officer certifying as to the satisfaction of all conditions specified in Section 6.2(b) (a form of which is attached as Schedule 6 hereto).

## 6.3 Conditions Precedent to Acquisitions by Condemnation.

- (a) DOE's initial obligation to assist with the acquisition of Project Real Estate Rights by way of Acquisitions by Condemnation shall commence upon the satisfaction of the following conditions precedent:
  - (i) the Commencement Date shall have occurred;
  - (ii) the Clean Line Entities shall have complied with all of the requirements and procedures set forth in Schedule 1 hereto with respect to the DOE Delegated Real Estate Right to be condemned;
  - (iii) the Financing Condition shall be satisfied and Holdings shall have delivered to DOE certified copies of all applicable executed DOE Approved Project Financing Commitments, DOE Approved Project Equity Commitments, Project Equity Commitments and/or Project Financing Documents;
  - (iv) (A) all Performance Support in an amount not less than the Applicable Amount shall be in full force and effect, (B) to the extent that Project Financial Close has occurred, the Clean Line Entities shall have granted a perfected security interest in the Second Lien Collateral in accordance with Section 11.6 in favor of DOE (but only to the extent that the first priority security interest in favor of the Project Financing Parties has been granted and/or perfected) and shall have delivered to DOE such legal opinions, certificates and other documents in respect thereof as DOE shall have requested and (C) to the extent that Project Financial Close has occurred, the Intercreditor Agreement shall be in full force and effect:

- (A) Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments in respect of not less than 2,000 MW of the Electrical Capacity in the aggregate (calculated as the sum of (A) with respect to Acceptable Permitted Project Investments, the sum of each portion of the Electrical Capacity transferred (and for which the Clean Line Entities have received payment or will receive payment within three (3) years after the date of such Permitted Project Investment) pursuant to each Acceptable Permitted Project Investment Commitment and (B) with respect to the Acceptable Transmission Services Agreements, the sum of the average Electrical Capacity committed in the initial five (5) years of the term for each such Acceptable Transmission Services Agreement) shall be in full force and effect and no event shall have occurred and be continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Acceptable Transmission Services Agreement or such Acceptable Permitted Project Investment Commitment and (B) the Converter Station Real Estate Rights Agreements shall be in full force and effect or the Clean Line Entities shall own in fee free and clear of all Liens other than Permitted Liens all Real Estate Rights necessary for the construction of the Converter Station Facility and the Intermediate Converter Station:
- (vi) (A) the Clean Line Parties shall have delivered to DOE certified copies of the Interconnection Agreements necessary for the operation of the Project, each of which shall have been duly executed and delivered and shall be in full force and effect, (B) each of the Material Interconnection Studies shall have been completed and (C) the Clean Line Parties are in compliance with the Interconnection Agreements;
- (vii) the Clean Line Parties shall have initiated joint discussions among officials and representatives of SPP, MISO and TVA to address any necessary inter-balancing area coordination and operational issues for the drafting of the applicable Operating Agreement;
- (viii) Holdings shall have delivered to DOE evidence that all Required Insurance is in full force and effect (including written binding verification of such coverage);
- (ix) (A) no Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (1) DOE's performance under this Agreement or any other Transaction Document then in effect or (2) DOE's participation in the Project and (B) no other final and non-appealable Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, (1) sets aside, enjoins or legally prohibits any Clean Line Entity's performance under this Agreement or any other Transaction Document then in effect or any Clean Line Entity's performance in any material respect under any Material Project Contract to which it is a party or (2) affects in any material respect the legality, validity or

enforceability of any of the Transaction Documents, any Project Equity Commitment or any of the Material Project Contracts then in effect;

- (x) (A) all representations and warranties made by any Clean Line Obligor in any of the Transaction Documents shall be true and correct in all material respects (except to the extent any such representation and warranty itself is qualified by "materiality", "material adverse effect", "Adverse DOE Impact", "Clean Line Material Adverse Effect" or a similar qualifier, in which case it shall be true and correct in all respects) and (B) no Default, Event of Default or Event of Loss shall have occurred and be continuing, in each case, as of the applicable date;
- (xi) Holdings shall have delivered to DOE an updated Project Development Progress Report and an updated Project Budget, which shall include a summary and explanation of any deviations from the Project Budget and the Project Schedule delivered as a condition to the occurrence of the Commencement Date and which shall include any Project Costs associated with the Interconnection Agreements or any Operating Agreement then in effect; and
- (xii) Holdings shall have delivered to DOE a certificate of an Authorized Officer as to the satisfaction of the foregoing conditions precedent.
- (b) After the satisfaction of the foregoing conditions precedent to DOE's obligation to assist with the acquisition of Project Real Estate Rights by way of Acquisitions by Condemnation, DOE's obligations to acquire or continue to acquire any DOE Delegated Real Estate Rights by way of Acquisitions by Condemnation shall only be subject to the following conditions being satisfied at all times:
  - (i) the Clean Line Entities shall have complied with all of the requirements and procedures set forth in Schedule 1 hereto with respect to the DOE Delegated Real Estate Rights to be acquired;
  - (ii) Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments in respect of not less than 2,000 MW of the Electrical Capacity in the aggregate (calculated as the sum of (A) with respect to Acceptable Permitted Project Investments, the sum of each portion of the Electrical Capacity transferred (and for which the Clean Line Entities have received payment or will receive payment within three (3) years after the date of such Permitted Project Investment) pursuant to each Acceptable Permitted Project Investment Commitment and (B) with respect to the Acceptable Transmission Services Agreements, the sum of the average Electrical Capacity committed in the initial five (5) years of the term for each such Acceptable Transmission Services Agreement) shall be in full force and effect and no event shall have occurred and be continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Acceptable Transmission Services Agreement or such Acceptable Permitted Project Investment Commitment;

- (iii) the Converter Station Real Estate Rights Agreements delivered pursuant to the foregoing conditions precedent shall continue to be in full force and effect and neither any Clean Line Entity nor any other Person party thereto shall be in default thereunder (or the Clean Line Entities shall then own in fee free and clear of Liens other than Permitted Liens all Real Estate Rights necessary for the construction of the Converter Station Facility and the Intermediate Converter Station):
- (iv) the Interconnection Agreements delivered pursuant to the foregoing conditions precedent shall continue to be in full force and effect and neither any Clean Line Entity nor any other Person party thereto shall be in default thereunder;
  - (v) the Financing Condition shall be satisfied;
- (vi) (A) no Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (1) DOE's performance under this Agreement or any other Transaction Document then in effect or (2) DOE's participation in the Project and (B) no other final and non-appealable Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, (1) sets aside, enjoins or legally prohibits any Clean Line Entity's performance under this Agreement or any other Transaction Document then in effect or any Clean Line Entity's performance in any material respect under any Material Project Contract to which it is a party or (2) affects in any material respect the legality, validity or enforceability of any of the Transaction Documents, any Project Equity Commitment or any of the Material Project Contracts then in effect; and
  - (vii) no Event of Default shall have occurred and be continuing.
- (c) After the conditions precedent to DOE's initial obligation to assist with the acquisition of Project Real Estate Rights by way of Acquisitions by Condemnation have been satisfied, at any time Holdings delivers a written notice of designation of any Project Real Estate Rights as a DOE Delegated Real Estate Right, Holdings shall concurrently deliver a certificate of an Authorized Officer certifying as to the satisfaction of all conditions specified in Section 6.3(b) (a form of which is provided as Schedule 7 hereto).
- (d) Notwithstanding the foregoing, from and after the Commencement Date, DOE may, at its sole option, elect to pursue an Uncontested Acquisition prior to the satisfaction in full of the conditions precedent set forth above, so long as the conditions precedent required to be satisfied for DOE's acquisition of DOE Delegated Real Estate Rights through a Voluntary Land Acquisition are satisfied.

### 6.4 Conditions Precedent to Notice to Proceed.

(a) Prior to the issuance by any Clean Line Entity of any notice to proceed under any Material Construction Contract that involves any physical construction activity

on any Project Real Estate Right, Holdings shall first have received a Notice to Proceed from DOE. DOE shall issue a Notice to Proceed to Holdings promptly upon satisfaction of the following conditions precedent:

- (i) the applicable Material Construction Contract shall have been duly executed and delivered and shall be in full force and effect;
- the Financing Condition is satisfied and Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments in respect of not less than 2,000 MW of the Electrical Capacity in the aggregate (calculated as the sum of (A) with respect to Acceptable Permitted Project Investments, the sum of each portion of the Electrical Capacity transferred (and for which the Clean Line Entities have received payment or will receive payment within three (3) years after the date of such Permitted Project Investment) pursuant to each Acceptable Permitted Project Investment Commitment and (B) with respect to the Acceptable Transmission Services Agreements, the sum of the average Electrical Capacity committed in the initial five (5) years of the term for each such Acceptable Transmission Services Agreement) shall be in full force and effect and no event shall have occurred and be continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Acceptable Transmission Services Agreement or such Acceptable Permitted Project Investment Commitment;
- (iii) the Performance Support in an amount not less than the Applicable Amount shall be in full force and effect;
- (iv) Holdings shall have delivered to DOE evidence that all Required Insurance is in full force and effect (including written binding verification of such coverage);
- (v) the Converter Station Real Estate Rights Agreements shall be in full force and effect or the Clean Line Entities shall then own in fee free and clear of Liens other than Permitted Liens all Real Estate Rights necessary for construction of the Converter Station Facility and the Intermediate Converter Station;
- (vi) (A) no Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (1) DOE's performance under this Agreement or any other Transaction Document then in effect or (2) DOE's participation in the Project and (B) no other final and non-appealable Governmental Order shall be in effect nor shall any Change of Law have occurred that, in either case, as applicable, (1) sets aside, enjoins or legally prohibits any Clean Line Entity's performance under this Agreement or any other Transaction Document then in effect or any Clean Line Entity's performance in any material respect under any Material Project Contract to which it is a party or (2) affects in any material respect the legality, validity or

enforceability of any of the Transaction Documents, any Project Equity Commitment or any of the Material Project Contracts then in effect;

- (vii) the Clean Line Entities and DOE shall have executed the NERC Agreement;
- (viii) all Interconnection Agreements and Material O&M Agreements shall have been duly executed and be in full force and effect; <u>provided</u> that to the extent that the Project Financing is in effect, the Material O&M Agreements shall be limited to those that are required to be in effect at such time as required under the terms of the Project Financing Documents;
- (ix) to the extent that DOE (or the Department of Labor, as the case may be) shall have made a determination that the Davis-Bacon Act applies to this Agreement and/or the Project, Holdings shall have delivered a certificate of an Authorized Officer to DOE dated not less than fifteen (15) days prior to the delivery of the Notice to Proceed stating either that the Clean Line Parties are in compliance with all applicable Davis-Bacon Requirements and, to the extent the Davis-Bacon Act is applicable, have included the provisions and wage determinations set forth in Schedule 15 hereto (as such Schedule is supplemented from time to time in accordance with Section 8.24(b)) in each applicable contract for construction, as defined in Department of Labor regulations at 29 C.F.R. § 5.2(j);
- (x) (A) all representations and warranties made by any Clean Line Obligor in any of the Transaction Documents shall be true and correct in all material respects (except to the extent any such representation and warranty itself is qualified by "materiality", "material adverse effect", "Adverse DOE Impact", "Clean Line Material Adverse Effect" or a similar qualifier, in which case it shall be true and correct in all respects) and (B) no Default, Event of Default or Event of Loss shall have occurred and be continuing, in each case, as of the applicable date; and
- (xi) Holdings shall have delivered to DOE a certificate of an Authorized Officer as to the satisfaction of the foregoing conditions precedent.
- (b) Notwithstanding anything herein to the contrary, the Clean Line Entities shall not be obligated to satisfy the conditions specified in this Section 6.4 in order to issue any purchase orders, work authorizations or limited notices to proceed (however titled) under any Material Construction Contract that concern (i) design, engineering, procurement or other non-site activities; (ii) civil, environmental and/or geotechnical surveys; (iii) pre-construction activities on Real Estate Rights held by the Clean Line Entities on which the Converter Station Facility and the Intermediate Converter Station will be constructed or installed such as installation of roads for access and clearing, grading and installation of appropriate base material (e.g., rock); (iv) clearing of rightsway on Real Estate Rights that are not DOE Acquired Real Property and/or (v) construction activities on Real Estate Rights in Oklahoma held by the Clean Line

Entities on which the Converter Station Facility will be constructed or installed; <u>provided further</u> that to the extent that the Performance Support then in effect is equal to the Applicable Amount that applies from and after the issuance of the Notice to Proceed (regardless of whether the Notice to Proceed has been issued or the other conditions to the issuance thereof have been satisfied), the Clean Line Entities shall not be obligated to satisfy the other conditions specified in this Section 6.4 in order to issue any purchase orders, work authorizations or limited notices to proceed (however titled) under any Construction Contract solely to commence clearing of rights of-way on any Project Real Estate Rights on DOE Acquired Real Property.

(c) Notwithstanding Section 6.4(b), to the extent that DOE (or the Department of Labor, as the case may be), determines that the Davis-Bacon Act is applicable to this Agreement and/or the Project, all construction, as defined in Department of Labor regulations at 29 C.F.R. § 5.2(j), shall be performed in compliance with the Davis-Bacon Requirements and the provisions and wage determinations set forth in Schedule 15 hereto (as such Schedule is supplemented from time to time in accordance with Section 8.24(b)) shall be incorporated into all such applicable contracts for construction.

# ARTICLE VII TERM, TERMINATION, EVENTS OF DEFAULT AND REMEDIES

# 7.1 Term and Termination.

- (a) Except to the extent terminated as contemplated below, the term of this Agreement shall commence on the Effective Date and continue until retirement from service of the Project Facilities and the completion of the Wind-Up Events, including the payment of all costs and expenses associated with the Wind-Up Events, at which point this Agreement shall terminate (save for any obligations that expressly survive such termination).
- (b) DOE may, at its option, terminate this Agreement upon the occurrence of the following events:
  - (i) (A) if at any time there is a final and non-appealable Governmental Order from a court of competent jurisdiction (not initiated or issued by DOE) finding that DOE is legally prohibited from participating in the Project or performing its obligations under the Transaction Documents or (B) any Change of Law shall have occurred that sets aside or legally prohibits DOE's participation in the Project;
  - (ii) if (A) the Financing Condition is not satisfied by December 31, 2021 or (B) the Commencement Date has not occurred by December 31, 2018; or
  - (iii) if any Event of Default occurs and is continuing; <u>provided</u> that from and after issuance of the Notice to Proceed, (A) DOE shall not terminate this Agreement unless such Event of Default is a Fundamental Event of Default and (B) DOE's right to terminate this Agreement shall be subject to the terms and conditions of the DOE Direct Agreement and the Intercreditor Agreement, if any;

provided, <u>further</u> that, after Project Completion, DOE shall not terminate this Agreement for an Operational EOD if the remedy described in Section 7.4(a)(v) is available to DOE.

(c) DOE's obligations under any other Transaction Document shall, at its election, terminate without any other action or agreement on the Termination Date, except to the extent that DOE shall otherwise agree that any such obligation survives in such other Transaction Document.

## 7.2 Acquisition Option.

- (a) After the Effective Date, Holdings and DOE shall discuss and determine whether, under Applicable Law, DOE may grant to Holdings (or its nominee, assignee or designee) an option to acquire from DOE the DOE Acquired Real Property and AR Facilities after the Termination Date (the "Acquisition Option"). To the extent DOE determines that such an Acquisition Option may be granted under Applicable Law, DOE and Holdings shall cooperate in good faith to enter into an agreement to set forth all of the terms, conditions and procedures under which such an Acquisition Option may be exercised by Holdings (or its nominee, assignee or designee).
- (b) Following the Termination Date, subject to the Acquisition Option (if applicable) and the terms and conditions of the DOE Direct Agreement and the Intercreditor Agreement, if any, DOE shall have the right to Dispose of any of its rights or interests in any DOE Acquired Real Property or any of the AR Facilities, including through a dismantling of any of the AR Facilities and a Disposition of any DOE Acquired Real Property to any Person without any consent by any Clean Line Party or any other Person.
- 7.3 <u>Events of Default</u>. The following events or circumstances shall constitute events of default under this Agreement (collectively, "<u>Events of Default</u>"):
  - (a) the Clean Line Parties fail to fund the Advance Funding Account in accordance with the terms set forth in this Agreement and such failure continues for a period of thirty (30) days following written notice to Holdings from DOE;
  - (b) (i) any Performance Support required to be maintained by the Clean Line Parties shall cease to be in full force and effect; <u>provided</u> that no Event of Default shall have occurred under this clause (b)(i) to the extent that either (A) such Performance Support has been drawn on in full by DOE and the proceeds thereof placed in a DOE or U.S. Treasury account or a collateral account pledged solely to DOE as a result of either the applicable provider of such Performance Support no longer constituting an Acceptable Support Provider or the pending expiration of such Performance Support or (B) Holdings shall have reinstated such Performance Support in an amount equal to the then Applicable Amount within thirty (30) days following written notice to Holdings from DOE, or (ii) following a draw on the Performance Support by DOE to satisfy any payment obligation of a Clean Line Party hereunder, the Clean Line Entities fail to replenish or reinstate such Performance Support to the then Applicable Amount in

accordance with Section 11.5(a) within fifteen (15) Business Days following written notice to Holdings from DOE;

- (c) (i) any Clean Line Obligor, any of their respective Controlling Persons or any Principal Person of any Clean Line Obligor or any of their respective Controlling Persons becomes (whether through a transfer or otherwise) a Prohibited Person, (ii) any Clean Line Obligor enters into a transaction with a Person who is a Prohibited Person (other than as required by Applicable Law) and such transaction is not voided or unwound (to the extent permissible under Applicable Law) within thirty (30) days following written notice to Holdings from DOE or (iii) any Clean Line Obligor, any of their respective Controlling Persons or any Principal Person, employee or agent of any Clean Line Obligor or any of their respective Controlling Persons fails to comply with any AM Laws, Anti-Corruption Laws or Sanctions;
- (d) any Clean Line Obligor fails to pay, in accordance with the terms of this Agreement or any other Transaction Document, any amounts required to be paid by such Clean Line Obligor pursuant thereto, and such failure to pay shall continue unremedied for a period of thirty (30) days after written notice from DOE that such amount was due;
- (e) any representation or warranty made or deemed made by any Clean Line Obligor in any Transaction Document or in any certificate or other document provided by or on behalf of any Clean Line Obligor to DOE are found to have been incorrect, false or misleading in any material respect (except to the extent any such representation and warranty itself is qualified by "materiality", "material adverse effect", "Adverse DOE Impact", "Clean Line Material Adverse Effect" or a similar qualifier, in which case it shall be true and correct in all respects) when made or deemed to have been made;
- (f) DOE or any Clean Line Obligor fails to perform or observe any of its material obligations under any other term, covenant or agreement set forth in this Agreement or any other Transaction Document where such failure has not been remedied within thirty (30) days (if such default is remediable) after such Party receives notice of such failure from the non-defaulting Party; provided, that if such Party or Clean Line Obligor, as applicable, commences and diligently pursues efforts to cure such default within such thirty (30) day period, and such default (i) in the case of a default by any Clean Line Obligor, could not reasonably be expected to have an Adverse DOE Impact and (ii) is capable of being cured, such Party or Clean Line Obligor, as applicable, may continue to effect such cure and such default will not be deemed to be an Event of Default for an additional one-hundred twenty (120) days so long as such Party or Clean Line Obligor, as applicable, is diligently pursuing such cure;
- (g) following the grant of a security interest in any Collateral, (i) any of the Security Documents shall fail in any material respect to provide the Liens, security interests, rights, titles, interests, remedies, powers or privileges intended to be created thereby (including the priority intended to be created thereby), or any Lien or security interest on the Collateral fails to have the priority contemplated therefor in such Security Document, or (ii) any such Security Document, Lien or security interest ceases to be in full force and effect, or the validity thereof or the applicability thereof to any obligation

of any Clean Line Obligor under the Transaction Documents are disaffirmed by or on behalf of any Clean Line Obligor;

- (h) any Transaction Document or any material provision thereof, at any time, for any reason, (i) is or becomes invalid, illegal, void or unenforceable or any Clean Line Obligor has repudiated or disavowed or taken any action to challenge the validity or enforceability of any Transaction Document or (ii) ceases to be in full force and effect, except in connection with its expiration or termination in accordance with its terms in the ordinary course;
- (i) one or more final and non-appealable Governmental Orders (not initiated or issued by DOE) are entered against any Clean Line Obligor that has an Adverse DOE Impact and such Governmental Order(s) are not vacated, discharged or stayed or bonded pending appeal for any period of thirty (30) days;
- (j) any Insolvency Event occurs with respect to any Clean Line Obligor (other than any Immaterial Obligor);
  - (k) any Change of Control occurs that is not otherwise consented to by DOE;
- (l) any of the Clean Line Entities fails to obtain, renew, maintain or comply in all material respects with any Required Approval, or having been obtained, any such Required Approval is, pursuant to a final and non-appealable order of the applicable Governmental Authority (not initiated or issued by DOE), (i) rescinded, terminated, suspended, modified, withdrawn or withheld, (ii) is determined to be invalid, (iii) ceases to be in full force and effect or (iv) is amended or modified so as to result in a Clean Line Material Adverse Effect or causes an Adverse DOE Impact, and such failure or other event described above has not been remedied within sixty (60) days (if such default is remediable) after any Clean Line Entity obtains Knowledge of such failure or other event;
- (m) all or substantially all of the Project Facilities are destroyed or become permanently inoperative as a result of an Event of Loss that is not covered by insurance or not repaired or restored with Loss Proceeds in accordance with the terms of this Agreement, unless within sixty (60) days of such Event of Loss, Holdings presents to the Coordination Committee a remedial and financing plan to restore or rebuild such Project Facilities, and such plan has been approved by the Coordination Committee within thirty (30) days thereafter;
  - (n) an Abandonment occurs;
- (o) any Clean Line Obligor is debarred or suspended from contracting with the United States government or any agency or instrumentality thereof;
- (p) at any time that the Project Financing is in effect, (i) any of the Project Financing Parties has declared any Indebtedness owed to the Project Financing Parties under any of the Project Financing Documents to be due and payable or required to be prepaid or redeemed (other than by a regularly scheduled required prepayment or

redemption), purchased or defeased or an offer to prepay, redeem, purchase or defease such Indebtedness is required to be made, in each case, prior to the stated maturity thereof or (ii) the Project Financing Parties have instituted foreclosure action in respect of any of the Collateral;

- (q) from and after the Commencement Date but prior to the occurrence of Project Completion, any of the Project Equity Commitments required to be in effect as a condition to DOE's obligation to acquire any DOE Delegated Real Estate Rights ceases to be in full force and effect, and such Project Equity Commitments are not replaced within a period of thirty (30) days after written notice from DOE or the Person providing such Project Equity Commitments are no longer in full force and effect: or
- (r) the occurrence of any other event specified to be an "Event of Default" or similar event in any Performance Support.

#### 7.4 Remedies Upon Event of Default.

- (a) Upon the occurrence of and during the continuance of any Event of Default, subject to the terms of the DOE Direct Agreement and the Intercreditor Agreement, DOE shall be entitled to exercise any and all of the following remedies (following any applicable notice and cure periods):
  - (i) DOE shall be entitled to seek temporary or permanent injunction, specific performance or other equitable relief specifically to enforce the obligations of the Clean Line Obligors under this Agreement or any other Transaction Document (and each of the Clean Line Parties hereby acknowledges and agrees that its failure to perform its obligations under this Agreement and the other Transaction Documents will cause irreparable harm to DOE and that the remedy at law for any violation or threatened violation thereof would be inadequate);
  - (ii) DOE may elect that Holdings' representatives on the Coordination Committee shall not have any right to decide, approve, authorize or vote on any matters before the Coordination Committee specifically relating to remedies to be taken against the Clean Line Parties upon such Event of Default;
  - (iii) DOE shall be entitled to suspend (without any consequence to DOE hereunder) performance of any of its condemnation or acquisition obligations under this Agreement or any other Transaction Document;
  - (iv) DOE shall be entitled to exercise all of its rights as a secured creditor of the Clean Line Entities in respect of the Collateral;
  - (v) if an Operational EOD occurs and is continuing, DOE may, after notice and the expiration of any applicable cure period, exercise replacement rights with respect to the Clean Line Entities by appointing another qualified and experienced Person to step in and assume management and operational control of

the Project (at the sole cost and expense of the Clean Line Parties) and in such circumstances, DOE may elect that (A) Holdings' representatives on the Coordination Committee shall cease to have any right to decide, approve, authorize or vote on any matters that would otherwise be decided by the Coordination Committee and (B) the Clean Line Parties shall cease to have any rights to enter into or use any DOE Acquired Real Property or the AR Facilities (except for the rights with respect to Electrical Capacity provided pursuant to Section 2.3);

- (vi) DOE shall be entitled to draw on the Performance Support to the extent necessary to satisfy any payment obligations of (A) DOE in respect of any Covered Cost, Covered Liability or any other payment paid or payable by DOE in connection with the Project or (B) any Clean Line Obligor due and owing to DOE or any Covered Party;
- (vii) DOE shall be entitled to default interest at the Default Rate on any overdue and unpaid amounts owing to DOE by any Clean Line Obligor; and
- (viii) DOE shall be entitled to exercise any and all other remedies available to it at law or in equity.
- (b) Prior to any exercise of remedies by DOE, DOE shall provide Holdings notice of the occurrence of the applicable Event of Default. Any costs and expenses incurred by DOE in connection with its exercise of any of its remedial rights shall be for the sole account of the Clean Line Parties. Except as otherwise set forth herein, each right and remedy of DOE hereunder shall be cumulative and shall be in addition to every other right or remedy provided herein or now or hereafter existing at law or in equity or by statute or otherwise, and the exercise or the beginning of the exercise by DOE of any one or more of any such rights or remedies shall not preclude the simultaneous or later exercise by DOE of any or all other such rights or remedies.
- (c) Upon the occurrence of and during the continuance of any Event of Default by DOE, the Clean Line Entities shall be entitled to exercise any and all other remedies available to it at law or in equity (following any applicable notice and cure periods); provided that, for the avoidance of doubt, no Event of Default by DOE shall have occurred to the extent that such Event of Default arises as a result of any Governmental Order or Change of Law that sets aside, enjoins or legally prohibits DOE's performance under this Agreement or any other Transaction Document or DOE's participation in the Project so long as such Governmental Order or Change of Law is not directly caused by actions of DOE that are specifically targeted at any of the Clean Line Entities or the Project (and not of a more generally applicable nature) or is a result of a violation of Applicable Law by any of the Clean Line Entities or the occurrence of an Event of Default. Except as otherwise set forth herein, each right and remedy of any Clean Line Party shall be cumulative and shall be in addition to every other right or remedy provided herein or now or hereafter existing at law or in equity or by statute or otherwise, and the exercise or the beginning of the exercise by a Clean Line Party of any

one or more of any such rights or remedies shall not preclude the simultaneous or later exercise by such Clean Line Party of any or all other such rights or remedies.

# 7.5 <u>Winding-Up of the Project.</u>

- (a) Upon retirement of the Project and the Project Facilities from service or to the extent required by DOE in connection with its exercise of its rights under Section 7.1, but subject to the Acquisition Option, the DOE Direct Agreement and the Intercreditor Agreement, the Clean Line Entities shall promptly wind-up the activities of the Project, which shall include, if requested by DOE, the following actions (the "Wind-Up Events"):
  - (i) dismantling, demolishing and removing all equipment, facilities and structures;
  - (ii) terminating applicable agreements in accordance with the terms thereof:
  - (iii) securing, maintaining and disposing of debris with respect to the Project Facilities and any Project Real Estate Rights; and
  - (iv) performing any activities necessary to comply with Applicable Law and Prudent Utility Practices and that are otherwise prudent to retire the Project Facilities, restore the Project Real Estate Rights to the original condition and protect the Parties from liability.
- (b) All costs and expenses related to the Wind-Up Events shall be borne by the Clean Line Parties.
- Wind-Up Reserve Account. Commencing no earlier than the twentieth (20<sup>th</sup>) 7.6 anniversary of Project Completion, Holdings shall establish and maintain a depositary account on terms reasonably acceptable to DOE (the "Wind-Up Reserve Account"), which Wind-Up Reserve Account shall be pledged on a first priority basis to DOE and shall not be pledged to any other Person; provided that if on the twentieth (20<sup>th</sup>) anniversary of the date of Project Completion, the remaining useful life of the Project and the Project Facilities is reasonably estimated to be in excess of ten (10) years, Holdings may delay the establishment of the Wind-Up Reserve Account until a date that is reasonably estimated by an Independent Engineer to be ten (10) years prior to the expiration of the useful life of the Project. Simultaneously with the establishment of the Wind-Up Reserve Account, and each subsequent year thereafter, Holdings shall deposit an amount into the Wind-Up Reserve Account equal to (a)(i) the current estimated costs to implement all of the Wind-Up Events, as determined by the Coordination Committee plus a reasonable contingency amount thereon as determined by an Independent Engineer (on an annual basis) less (ii) the amount on deposit in the Wind-Up Reserve Account; divided by (b) the estimated number of years, as determined by the Coordination Committee (on an annual basis), until commencement of the Wind-Up Events. Notwithstanding anything to the contrary set forth above, Holdings shall have the option of funding the then-required amount of the Wind-Up Reserve Account with an Acceptable Letter of Credit or cash, or any combination thereof.

- Event of Loss. To the extent that the Loss Proceeds associated with any Event of Loss (or the time contemplated for repair or replacement of any affected Property) are reasonably anticipated to be less than the Loss Threshold, the Clean Line Entities shall be entitled to elect to repair and restore any Property affected by such Event of Loss and shall be entitled to all Loss Proceeds payable in connection with such Event of Loss. To the extent that the Loss Proceeds associated with any Event of Loss (or the time contemplated for repair or replacement of any affected Property) are reasonably anticipated to exceed the Loss Threshold, then the Parties shall initiate the Wind-Up Events unless the Parties and, if applicable, the Project Financing Parties, mutually agree to repair or replace the affected Property or the Clean Line Entities present a remedial and financing plan approved by the Coordination Committee within one hundred twenty (120) days after such Event of Loss to repair, replace or restore such affected Property. Any Loss Proceeds payable in respect of any Event of Loss shall be applied as follows: first, to the extent that the Clean Line Entities are entitled to repair or restore any affected Property and have so elected to repair and restore such affected Property, such Loss Proceeds shall be paid to the Clean Line Entities to enable the repair and restoration of such affected Property; second, to the extent of any excess Loss Proceeds remaining after any such repair or restoration is completed, an amount determined by DOE as necessary to be reserved (taking into consideration the amount of the Performance Support and any amounts available in the Wind-Up Reserve Account) to cover any potential additional claims for damages to DOE relating to such Event of Loss shall be set aside in a reserve account pledged to the benefit of DOE and maintained for a period of two (2) years or such shorter time period as agreed to by the Coordination Committee (and to the extent necessary shall be applied to the payment of any such damages); and third, any remaining excess Loss Proceeds shall be released to the Clean Line Entities, subject to the terms of the Project Financing Agreements.
- 7.8 <u>Survival of Obligations</u>. The rights and obligations of the Clean Line Parties under Sections 2.3, 7.2, 11.1, 11.3, 11.4, 11.8, 11.9, 13.17, 13.18 and 13.20, and Article IX shall survive the Termination Date; <u>provided</u> that Section 7.2(a) shall terminate within six (6) months after the Termination Date (unless otherwise extended as agreed to by DOE); <u>provided further</u> that the survival of any rights of the Clean Line Entities, other than the Acquisition Option (if applicable), shall not in any way limit DOE's right to dispose, transfer, sell, dismantle or take any other actions with respect to any of the AR Facilities or DOE Acquired Real Property after the Termination Date.

# ARTICLE VIII COVENANTS

- 8.1 <u>Recordkeeping</u>. Holdings shall and shall cause all of its Subsidiaries to keep proper records and books of account in which full, true and correct entries in accordance with GAAP and FERC standards and all Applicable Laws are made in respect of all dealings and transactions relating to the Project and the conduct of their business.
- 8.2 <u>DOE's Access Rights, Etc.</u> Upon reasonable advance notice and during normal business hours, DOE (through its officers, agents and designated representatives) shall have:
  - (a) the right to visit and inspect the Project, subject to reasonable safety and security requirements of which DOE receives prior written notice;

- (b) access to books, documents, papers and records of the Clean Line Parties for the purposes of audit, examination, inspection and monitoring;
- (c) the right to discuss the affairs, finances and accounts of the Clean Line Parties with representatives of the Clean Line Parties (including any auditors or accountants of the Clean Line Entities); and
- (d) the independent right to (i) monitor the development, design, engineering, construction, financing, ownership, operation, maintenance and management of the Project (including participating in any acceptance testing) and (ii) review and comment on draft copies of all Material Project Contracts, and applications for Governmental Approvals for the Project.

The Clean Line Entities shall coordinate and cooperate, and require their Contractors to coordinate and cooperate, with DOE to facilitate DOE's access rights set forth above.

- 8.3 <u>Reporting Requirements</u>. During the term of this Agreement, Holdings shall furnish to DOE the following items:
  - (a) after the Commencement Date and prior to the issuance of the Notice to Proceed, within twenty (20) Business Days after the end of each calendar quarter, a Project Development Progress Report based upon Holdings' good faith reasonable estimates of the information contained therein;
  - (b) after issuance of the Notice to Proceed through Project Completion, within twenty (20) Business Days after the end of each calendar quarter, a Construction Progress Report based upon Holdings' good faith reasonable estimates of the information contained therein;
  - (c) from and after Project Completion, within thirty (30) days after the end of each calendar quarter, quarterly operating reports in a form to be mutually agreed between Holdings and DOE and which shall include (i) an update on all material issues with respect to the Project (including any material Events of Loss or Actions that have arisen or exist with respect to the Project or any material noncompliance with any Required Approval then in effect), (ii) a summary of the operating status of the Project (including with respect to Electrical Capacity, availability, forced outages, safety statistics and outage status for planned outages) and (iii) such other information as DOE may reasonably request to be included from time to time;
  - (d) as soon as available, but in any event within sixty (60) days after the end of each of the first three (3) fiscal quarters of each fiscal year of Holdings, unaudited Financial Statements for Holdings and (on a consolidated basis) its Subsidiaries for such fiscal quarter and the then elapsed portion for the relevant fiscal year and comparative figures for the same periods in the immediately preceding fiscal year;
  - (e) as soon as available, but in any event within one hundred twenty (120) days after the end of each fiscal year of Holdings (starting with the fiscal year ending December 31, 2016), (i) audited Financial Statements of Holdings and (on a consolidated

basis) its Subsidiaries for such fiscal year, accompanied by a report and opinion of an independent auditor to the effect that such financial statements present fairly in all material respects the financial condition, results of operations, shareholders' equity and cash flows of Holdings and its Subsidiaries for such fiscal year, which report and opinion is prepared in accordance with GAAP and (ii) a certificate of an Authorized Officer of Holdings, which certificate shall state that such Financial Statements fairly represent the financial condition and results of operations of Holdings and its Subsidiaries for such fiscal year;

- (f) concurrently with the delivery of any Financial Statements pursuant to clauses (c) and (d) above, a certificate of an Authorized Officer of Holdings certifying that, to the Knowledge of the Clean Line Parties, no Default or Event of Default exists, or, if such certification cannot be made, the nature and period of existence of such Default or Event of Default and what corrective action Holdings and its Subsidiaries have taken or propose to take with respect thereto;
- (g) within sixty (60) days after the end of each calendar year commencing with the calendar year in which Project Completion occurs, a report prepared by Holdings detailing the proposed maintenance and outage program for the Project for such calendar year;
- (h) promptly, but in any event within ten (10) Business Days (or, in the case of clause (xii), notice of the occurrence of any Safety Event or any other accident related to the Project that involves a loss of life within twenty-four (24) hours) after any of the Clean Line Parties obtains Knowledge thereof or information pertaining thereto, notice of:
  - (i) following Project Financial Close but prior to Project Completion, the occurrence of any event or circumstance that has resulted in, or could reasonably be expected to result in, a failure to satisfy the Permitted Draw Conditions:
  - (ii) the occurrence of any event that constitutes a Default or Event of Default, specifying the nature thereof, together with a certificate of an Authorized Officer of Holdings indicating any steps that the Clean Line Parties have taken or propose to take to remedy the same;
  - (iii) the occurrence of (A) any Action, pending or threatened, that relates to the legality, validity or enforceability of any of the Transaction Documents, (B) any material Action, pending or threatened, that relates to the Project or to which a Clean Line Party is a party or (C) any material hearing or proceeding initiated against any Clean Line Party by any Governmental Authority that specifically affects the Project;
  - (iv) any actual or proposed termination, rescission, discharge (otherwise than by performance), amendment, supplement, modification, waiver or indulgence or breach in any material respect of any Material Project Contract

- or Required Approval together with a copy of any notice or correspondence received in respect thereof and copies of any proposed amendment, supplement, modification or waiver in respect of such Material Project Contract or Required Approval;
- (v) other than Permitted Liens, any Lien being granted or established or becoming enforceable over any of the Properties of the Clean Line Entities or the Equity Interests in any of the Clean Line Entities, together with a description thereof:
- (vi) any proposed material change in the nature or scope of the Project or the business or operations of the Clean Line Parties, together with a description thereof;
- (vii) the occurrence of any Event of Loss that is reasonably likely to result in Loss Proceeds in excess of \$5,000,000, together with a description thereof:
- (viii) any non-compliance of any Performance Support with the criteria established with respect thereto and any event, condition or circumstance that represents or could reasonably be expected to lead to non-compliance by any issuer with the required criteria with respect thereto or the renewal thereof;
- (ix) the occurrence of any event of Force Majeure affecting, or that any Clean Line Party claims would affect, the performance by such Person of any obligation under any Transaction Document or is otherwise reasonably likely to have a Clean Line Material Adverse Effect;
- (x) any material dispute between any Clean Line Entity and any Project Participant party to any Material Project Contract (where DOE is not also a party to such dispute), together with a copy of any material notice or material correspondence received in respect thereof;
- (xi) any proposed cancellation or material change in any Required Insurance maintained by any Clean Line Entity or by any other Person for the benefit of any Clean Line Entity or DOE, together with a report describing such event and the potential insurance-related impact thereof;
- (xii) the occurrence of any Safety Event or any other accident related to the Project that involves a loss of life together with a reasonably detailed report describing such Safety Event or accident, the impact of such Safety Event or accident and the remedial efforts required and (as and when taken) implemented with respect thereto;
- (xiii) any actual or alleged violations in any material respect by any of the Clean Line Parties of any Environmental Laws or any applicable NERC reliability standards in connection with the Project, together with a reasonably detailed summary of such violations, copies of any material notices or material

correspondence received in connection therewith and a description of the remedial efforts that the Clean Line Parties propose to take in connection therewith;

- (xiv) any material dispute between a Clean Line Entity and a Governmental Authority with respect to the Project's compliance with a term or condition of a Governmental Approval or Governmental Order, together with a copy of any notice or correspondence in respect thereof; and
- (xv) any material Environmental Claims related to the Project, together with a copy of any material correspondence relating thereto and a description of any steps that the Clean Line Parties are taking or propose to take with respect thereto:
- (i) promptly, but in any event no later than ten (10) Business Days after receipt, filing, delivery or sending thereof, copies of:

#### (i) Reserved;

- (ii) any notice of a delinquent payment owed by any Clean Line Party to any Project Participant pursuant to the terms of any Project Contract if such payment is more than thirty (30) days delinquent and is in excess of \$5,000,000, together with a copy of all correspondence received or sent by any Clean Line Party in respect of such delinquent payment;
- (iii) any notices or material correspondence from any Project Participant relating to (A) any material delay in the completion of the Project or (B) the occurrence of any event that could reasonably be expected to interrupt operation of the Project for more than thirty (30) Business Days;
- (iv) any material reports filed by any Clean Line Party with any Governmental Authority relating to the Project or any other financial information, statutory audits, proxy materials or other material information delivered or provided by any Clean Line Entity to any Governmental Authority;
- (v) any material notices, certificates or reports delivered by any Clean Line Party to the Project Financing Parties or any material notices or other material written correspondence received by any of the Clean Line Parties from or on behalf of the Project Financing Parties (including any notices of the occurrence of a default or event of default in respect thereof); and
- (vi) any Required Approval issued to or on behalf of the Clean Line Entities in respect of the Project.
- (j) as soon as available but in any event no later than ten (10) Business Days following execution thereof, copies of any executed Project Financing Documents and any amendments, modifications, supplements or waivers in respect thereof;

- (k) (i) no later than five (5) Business Days prior to the commencement of any commissioning testing in respect of the Project, written notice thereof; (ii) written notice of the occurrence of Project Completion (which shall include a certificate by an Authorized Officer of Holdings as to the satisfaction of the conditions to the occurrence of Project Completion) and (iii) promptly upon receipt of delivery thereof, a copy of any notice of the occurrence of the commencement of commercial operations to any party to any Transmission Services Agreement;
- (l) on an annual basis after the Commencement Date and within thirty (30) days of each anniversary thereof, a certificate from an Authorized Officer of Holdings that the Clean Line Entities are in compliance with Section 8.5 and that all Required Insurance is in full force and effect, accompanied by a certification from Holding's insurance broker confirming the foregoing; and
- (m) promptly upon request, such other information or documents as DOE may reasonably request from time to time.

#### 8.4 <u>Authorizations and Approvals.</u>

- (a) The Clean Line Entities shall obtain, and in the case of Required Approvals of Contractors, shall cause its Contractors to obtain, all Required Approvals, at their sole cost and expense. No later than ninety (90) days after the Commencement Date, Holdings shall provide a schedule of all Known Required Approvals to DOE and a plan for the acquisition of such Required Approvals. Concurrently with the delivery of any Financial Statements pursuant to Section 8.3(c) or 8.3(d) above, Holdings shall report, on a quarterly basis, the status of all applications for Required Approvals. In the event that a Required Approval is denied or includes terms and conditions that may materially affect Project operations, Holdings shall immediately notify DOE of such event and consult with DOE on measures taken to remedy such adverse event.
- (b) For any Required Approval that is issued in any Clean Line Entity's name or otherwise is applicable to any Clean Line Entity, the applicable Clean Line Entity shall, at its own cost and expense, comply with all conditions imposed by and undertake all actions required by and all actions necessary to maintain in full force and effect all Required Approvals. To the extent that a Required Approval is issued solely in DOE's name or jointly to both DOE and any Clean Line Entity, the Clean Line Entities shall be responsible for any costs or expenses that DOE incurs in taking all actions necessary to maintain in full force and effect such Required Approval.
- (c) Prior to Project Completion, the Clean Line Entities shall have obtained from FERC any exemptions or waivers from regulation under PUHCA for which it may be eligible under FERC's regulations.
- (d) In the event that any Required Approval must be issued in DOE's name, Holdings shall undertake necessary efforts to obtain such Required Approval, subject to DOE's reasonable cooperation with Holdings at Holdings' sole expense, as such cooperation by DOE is limited by Section 4.10. In the event that DOE must act as the

lead agency and directly coordinate with any Governmental Authority in connection with obtaining any Governmental Approval, the Clean Line Entities shall promptly provide all necessary support consistent with Applicable Law to facilitate the approval, mitigation or compliance process for such Governmental Approval.

(e) Each Clean Line Entity shall, at its own cost and expense, (i) obtain all Governmental Approvals or any other approvals, consents, exemptions, authorizations or other actions by, or notices to, or filings with, any other Person that may be necessary or required from time to time in connection with the performance by such Clean Line Entity of its obligations and undertakings under this Agreement or any other Transaction Document (in light of the current stage of construction, management and/or operation of the Project as of any date of determination) and (ii) comply with the terms and conditions of any such Governmental Approval or other approval, consent, exemption, authorization, notice or filing (if applicable) to the extent in effect from time to time, in each case, except where the failure to so obtain or comply with any such Governmental Approval or other approval, consent, exemption, authorization, notice or filing could not reasonably be expected to result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.

#### 8.5 Insurance.

- (a) The Clean Line Entities shall obtain, maintain or cause to be maintained insurance of the types, in the amounts and with the deductibles specified in the Insurance Agreement, as in effect from time to time, and in all cases in accordance with Prudent Utility Practices.
- (b) The Clean Line Entities shall cause DOE to be named as an "additional insured" and as "loss payee" under each of its insurance policies to the extent required under the terms of the Insurance Agreement. The Parties will also determine appropriate insurance protections to be set forth and agreed in the Insurance Agreement for DOE and the other Covered Parties through insurance policies procured by the Construction Contractors and other major Contractors, including additional insured and loss payee endorsements.
- (c) Each of the Clean Line Entities shall use all commercially reasonable efforts to enforce any Contractual Obligations by any Construction Contractor under a Material Construction Contract to obtain and maintain any of the Required Insurance.
- (d) In the event that any Clean Line Entity fails to procure or maintain (or cause to be procured or maintained) any Required Insurance, DOE may (but shall not be obligated to) take out the Required Insurance and pay the premiums on the same. All amounts so advanced for such purpose by DOE shall be a Covered Liability owed by the Clean Line Parties to DOE and the Clean Line Parties shall forthwith pay any such amounts to DOE.
- 8.6 <u>Payment of Taxes and Other Amounts</u>. Each of the Clean Line Parties shall pay or arrange for the payment of (before they become overdue) all present and future (a) Taxes

(including stamp taxes), duties, fees, expenses, or other charges payable on or in connection with the Project or the execution, issue, delivery, registration, or notarization of, or for the legality, validity, or enforceability of, this Agreement, the Security Documents and the other Transaction Documents, (b) claims, levies, or liabilities (including claims for labor, services, materials and supplies) for sums that have become due and payable and that have resulted in or, if unpaid, might result in the imposition of a Lien upon the Property of any Clean Line Entity with respect to the Project Real Estate Rights on the AR Facilities (or any part thereof), (c) Taxes, payments, fees and expenses relating to the acquisition of the Project Real Estate Rights and (d) Local Government Contribution Payments.

# 8.7 Maintenance of Existence and Property.

- (a) Each Clean Line Party shall preserve and maintain (i) its legal existence as a limited liability company and (ii) all of its licenses, rights, privileges and franchises material to the conduct of its business and the Project.
- (b) Each Clean Line Party shall engage only in the business consistent with the Transaction Documents and the Project Financing Documents to which it is a party and any business reasonably incidental or related thereto (including with respect to the Other Facilities).
- (c) Each Clean Line Entity shall (i) keep (or cause to be kept) all its Properties (including with respect to the Project) in good working order and condition to the extent necessary to ensure that its business can be conducted properly at all times and (ii) develop, construct, operate, maintain and repair the Project or cause the Project to be developed, constructed, operated, maintained and repaired in all material respects in accordance with (A) the standards set forth in the Clean Line Documents as in effect from time to time, (B) in all material respects in accordance with manufacturer's recommendations (to the extent required to maintain material warranties in effect), (C) Required Approvals and (D) Prudent Utility Practices.
- 8.8 <u>Compliance with Applicable Laws</u>. Each of the Clean Line Entities shall, and with respect to the construction, operation and maintenance of the Project shall use commercially reasonable efforts to enforce and diligently pursue all contractual remedies available to it to cause each Project Participant to, comply with and conduct its business and operations in compliance with all Required Approvals and Applicable Laws, including Environmental Laws and Cultural Resource Agreements, except where the failure to comply could not reasonably be expected to result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.
- 8.9 <u>Diligent Construction of the Project</u>. Each of the Clean Line Entities shall use diligent efforts to construct and complete, or cause to be constructed and completed, the Project in all material respects in accordance with the Project Contracts, Required Approvals, Prudent Utility Practices, the Project Schedule, the Project Budget and the terms and conditions of the applicable Transaction Documents.

- 8.10 Performance of Obligations. Each of the Clean Line Entities shall (a) perform and observe all of its material covenants and obligations contained in any Material Project Contract or Required Approval, (b) take all reasonable and necessary action to prevent the termination, suspension or cancellation of any Material Project Contract or Required Approval (except for the expiration of any Material Project Contract or Required Approval in accordance with its terms and not as a result of a breach or default thereunder by any Clean Line Entity) and (c) enforce against the relevant Project Participant each material covenant or obligation under each Material Project Contract to which such Person is a party in accordance with its terms, in each case except where failure to do so could not reasonably be expected to have a Clean Line Material Adverse Effect or an Adverse DOE Impact.
- 8.11 <u>Permitted Liens</u>. Each Clean Line Entity shall not, and shall not agree to, create, assume or otherwise permit to exist (a) any Lien upon any of the Collateral or any of its other material Property, whether now owned or hereafter acquired, or in any proceeds or income therefrom, other than Permitted Liens or (b) any Lien upon its Equity Interests other than Permitted Liens.
- 8.12 <u>Merger; Bankruptcy; Dissolution; Transfer of Assets</u>. Each Clean Line Entity shall not, and shall not agree to:
  - (a) enter into any transaction of merger, combination or consolidation;
  - (b) liquidate, wind-up or dissolve itself or otherwise commence any Insolvency Event in respect of itself or file any petition or pass a resolution seeking the same:
  - (c) Dispose of all or any part of its Property, including its interest in the Project, whether now owned or hereafter acquired, except for Permitted Dispositions;
  - (d) acquire by purchase or otherwise the business, Property or assets of, or Equity Interests or other evidence of beneficial ownership interests in, any Person, other than purchases or other acquisitions of inventory or materials or spare parts or Capital Expenditures, each in the ordinary course of business or any Emergency Capital Expenditures or Emergency Operating Expenses; or
  - (e) transfer or release (other than as permitted by clause (c) above) the Collateral or any portion thereof.
- 8.13 New Subsidiaries; Partnerships. Without the prior written consent of DOE, no Clean Line Entity shall: (a) form or have any Subsidiaries other than (i) those in existence as of the Effective Date or (ii) new Project Subsidiaries that become a party to this Agreement pursuant to Section 8.18, (b) enter into any partnership, joint venture or similar arrangement, (c) acquire any Equity Interests in or make any capital contribution to any other Person (other than to another Clean Line Entity or, in the case of Holdings only, acquire any Equity Interests in or make any capital contribution to PECL and any PECL Subsidiary) or (d) enter into any management contract or similar arrangement whereby its business or operations are managed by any other Person.

- 8.14 <u>Subsidiaries of Holdings</u>. No Project Subsidiary shall own (a) any real Property rights other than those relating to the Project or (b) any Equity Interests other than Equity Interests in any Subsidiary of Holdings or any other Person that is also a Project Subsidiary. None of the Project Subsidiaries shall be a party to any Contractual Obligation relating to the ownership, development, construction, procurement, operation, management or maintenance of any Properties other than those relating to the Project and the Other Facilities.
- 8.15 Other Transactions. Except for the Transaction Documents, the Project Equity Commitments, the Project Financing Commitments and the Project Financing Documents, no Clean Line Entity shall, directly or indirectly, enter into any transaction or series of related transactions with any Affiliate other than in the ordinary course of business on fair and reasonable terms no less favorable to the Clean Line Entities than those that would be included in an arm's-length transaction with a non-Affiliate.
- 8.16 <u>Testing</u>. The applicable Clean Line Entities shall (a) provide, or cause to be provided, reasonable prior notice to DOE regarding the startup testing of the Project pursuant to the Material Construction Contracts, (b) provide DOE (or its representatives, agents or consultants) with the opportunity to observe the startup testing of the Project and (c) provide DOE with any material data or material reports received by any Clean Line Entity in connection with the startup testing of the Project pursuant to the Material Construction Contracts.

# 8.17 <u>Creation and Perfection of Security Interests; Additional Documents; Filings and Recordings.</u>

- (a) Each of the Clean Line Entities shall execute and deliver, from time to time as reasonably requested by DOE at the expense of the Clean Line Entities, such other documents as shall be necessary or advisable or that DOE may reasonably request in connection with the rights and remedies of DOE granted or provided for by the Transaction Documents and to consummate the transactions contemplated therein.
- Each of the Clean Line Entities shall, at its own expense, take all actions that have been or shall be reasonably requested of such Clean Line Entity or that any Clean Line Entity knows is necessary to establish, maintain, protect, perfect and continue the perfection of the security interests of DOE created by the Security Documents with the priority provided for under the Security Documents (subject to Permitted Liens and, with respect to the Second Lien Collateral, only to the extent that the first priority security interest in favor of the applicable Financing Parties has been established and/or perfected) and shall furnish timely notice of the necessity of any such action, together with such instruments, in execution form, and such other information as may be required or reasonably requested to enable DOE to effect any such action. Without limiting the generality of the foregoing, each of the Clean Line Entities shall, at its own expense, (i) execute or cause to be executed and shall file or cause to be filed or register or cause to be registered such financing statements, continuation statements, fixture filings and mortgages or deeds of trust in all places necessary or advisable (in the reasonable opinion of counsel for DOE) to establish, maintain and perfect such security interests and in all other places that DOE shall reasonably request (provided, however, that, with respect to the Second Lien Collateral, the Clean Line Entities shall not be obligated to execute, file

or register any such statements, filings, mortgages or deeds of trust or other documents that are not required to be executed, filed or registered in respect of the first priority security interest granted in favor of the applicable Financing Parties), (ii) discharge all other Liens (other than Permitted Liens) adversely affecting the rights of any Clean Line Entity in the Collateral and (iii) deliver or publish all notices to third parties that may be reasonably required to establish or maintain the validity, perfection or priority of any Lien created pursuant to the Security Documents (provided, however, that, with respect to the Second Lien Collateral, the Clean Line Entities shall not be obligated to deliver or publish any notices that are not required to be delivered or published in respect of the first priority security interest granted in favor of the applicable Financing Parties).

#### 8.18 Additional Project Subsidiaries and Subsidiary Guarantors.

- (a) Within twenty (20) Business Days following the formation or acquisition, directly or indirectly (including through any merger or consolidation), by any Clean Line Entity of any Project Subsidiary, Holdings shall, at the sole cost and expense of the Clean Line Entities, cause such Project Subsidiary to become a Party hereto by executing and delivering to DOE a joinder agreement to this Agreement in form and substance reasonably satisfactory to DOE along with the documents set forth in clause (c) below.
- (b) Within twenty (20) Business Days following the formation or acquisition, directly or indirectly (including through any merger or consolidation), by any Clean Line Party of any PECL Subsidiary, Holdings shall, at the sole cost and expense of the Clean Line Parties, cause such PECL Subsidiary to become a Subsidiary Guarantor and be obligated for all Guaranteed Obligations by executing and delivering to DOE a joinder agreement to this Agreement in form and substance reasonably satisfactory to DOE along with the documents set forth in clause (c) below.
- (c) Together with the delivery of any joinder agreement referenced in either clause (a) or (b) above, the applicable Clean Line Party shall deliver to DOE (i) certified Organizational Documents of such Clean Line Party, (ii) secretary's certificates, officer's certificates, resolutions and good standing certificates for such Clean Line Party (including certificates certifying to such matters as DOE shall reasonably require) and (iii) if requested by DOE, legal opinions from counsel to such Clean Line Party.
- 8.19 <u>Lobbying Disclosure Requirement</u>. Each of the Clean Line Parties shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions.
- 8.20 <u>Improper Use</u>. Unless required under Applicable Law or as otherwise provided for under this Agreement, no Clean Line Entity shall use, operate or occupy, or allow (directly or indirectly) the use, maintenance, operation or occupancy of, any portion of the Project Site or the Project in any manner or for any purpose: (a) that could reasonably be expected to have a Clean Line Material Adverse Effect or an Adverse DOE Impact, (b) that may make void, voidable or cancelable any insurance or material warranty then in force with respect to the Project Facilities or (c) other than for the intended purpose thereof in the construction, operation and maintenance of the Project Facilities.

#### 8.21 Hazardous Substance Management.

- (a) In the event of a Release or discovery of Hazardous Substances on Property on which the Project Facilities are located, the responsible Clean Line Entity shall, or shall cause another Person to take all reasonable actions, consistent with Prudent Utility Practice, Applicable Law and all applicable provisions of the Transaction Documents and the Project Documents, to report, investigate, oversee, manage, treat, handle, store, remediate, remove, transport (where applicable), deliver or dispose of such Hazardous Substances; <u>provided</u> that where consistent with Applicable Law and Prudent Utility Practice, the Hazardous Substances may be left *in situ*.
- (b) If any Clean Line Entity or any Construction Contractor Releases or causes to Release Hazardous Substances in connection with the Project or any Project Real Estate Rights in an amount, type, quality or location that would require reporting or notification to any Governmental Authority or other Person or taking any preventive or remedial action, in each case under Applicable Law, Governmental Approvals or any applicable provision of the Transaction Documents and Project Contracts, the responsible Clean Line Entity shall (i) promptly notify DOE in writing and advise DOE of any obligation to notify any state or federal Governmental Authorities under Applicable Law and (ii) notify any such state or federal Governmental Authorities.
- (c) The responsible Clean Line Entity shall, or shall cause another Person to take reasonable steps, including design and/or construction technique modifications, to avoid and/or minimize disturbance of known *in situ* Hazardous Substances. Where the disturbance of Hazardous Substances, including excavation or dewatering, is unavoidable or is required by Applicable Law, the Clean Line Entities shall utilize appropriately trained Contractors.

#### 8.22 Safety Compliance.

- (a) DOE shall use good faith efforts to inform Holdings at the earliest practicable time of any circumstance or information relating to the Project which, in DOE's reasonable judgment, is likely to result in a Safety Compliance Order. Except in the case of an Emergency, DOE shall consult with Holdings prior to issuing a Safety Compliance Order concerning the risk to public or worker safety, alternative compliance measures, cost impacts, and the availability of Clean Line Entity resources to fund the Safety Compliance work.
- (b) Subject to conducting such prior consultation as required (unless in the case of an Emergency), DOE may issue Safety Compliance Orders to the Clean Line Entities at any time.
- (c) The Clean Line Entities shall implement all Safety Compliance work as expeditiously as reasonably possible following issuance of the Safety Compliance Order. The Clean Line Entities shall diligently prosecute the work necessary to achieve such Safety Compliance until completion. The Clean Line Entities shall perform all work

required to achieve Safety Compliance at the sole cost and expense of the Clean Line Entities.

(d) The Clean Line Entities shall adopt and comply with those applicable safety and Emergency response measures for the Project adopted in accordance with the DOE Mitigation Action Plan and all other safety-related or emergency response measures required under Applicable Law.

#### 8.23 Prohibited Persons.

- (a) Each Clean Line Party shall provide immediate written notice (including a brief description relating thereto) to DOE if, at any time, it learns that the representations made with respect to Prohibited Persons (including in respect of the Debarment Regulations) were erroneous when made or have become erroneous by reason of changed circumstances.
- (b) If any Project Participant or any Person that controls a Project Participant or any of their respective Principal Persons becomes a Prohibited Person, the Clean Line Parties shall, within sixty (60) days of Knowing that such Person has become a Prohibited Person, engage and continue to engage in constructive discussions with DOE regarding the removal or replacement of such Person or, if such removal or replacement is not reasonably feasible, the implementation of other mitigation matters.

#### 8.24 Davis-Bacon Act.

- To the extent that DOE (or the Department of Labor, as the case may be) (a) has determined that the Davis-Bacon Act is applicable to this Agreement and/or the Project, the Clean Line Entities shall (i) in respect of this Agreement, comply with all Davis-Bacon Requirements (including the provisions and wage determinations set forth in Schedule 15 hereto (as such Schedule is supplemented from time to time in accordance with Section 8.24(b)) and the provisions and applicable wage determinations set forth in such Schedule 15 shall be deemed incorporated into this Agreement as if set out in their entirety in this Section 8.24 and (ii) in respect of any applicable Project Contract (A) be responsible for the compliance by any applicable Contractor performing construction, as defined in Department of Labor regulations at 29 C.F.R. § 5.2(j), with the Davis-Bacon Requirements and (B) cause each applicable Contractor performing construction, as defined in Department of Labor regulations at 29 C.F.R. § 5.2(j), to include in such contract to which it is a party the provisions and applicable wage determinations set forth in Schedule 15 hereto (as such Schedule is supplemented from time to time in accordance with Section 8.24(b)).
- (b) To the extent DOE (or the Department of Labor, as the case may be) has determined that the Davis-Bacon Act is applicable to this Agreement and/or the Project, from time to time after such determination is made and in any event prior to Clean Line entering into any applicable Project Contract subject to the Davis-Bacon Act, DOE may supplement Schedule 15 hereto to incorporate the wage determinations containing locally

prevailing wages as determined by the Secretary of Labor applicable to this Agreement or such Project Contract, as the case may be.

- 8.25 <u>AM Laws, Anti-Corruption Laws Etc.</u> Each Clean Line Party shall and shall cause its Principal Persons, employees and agents to (a) comply with all applicable AM Laws and Anti-Corruption Laws in obtaining any Required Approvals, Project Real Estate Rights or any other consents, rights or privileges with respect to the Project, (b) conduct the business of the Project in compliance with all applicable AM Laws and Anti-Corruption Laws and (c) maintain internal management and accounting practices and controls that are adequate to ensure the Clean Line Parties' compliance with all applicable AM Laws and Anti-Corruption Laws.
- 8.26 <u>ACL Indebtedness</u>. ACL shall not incur any Indebtedness owed to any other Clean Line Entity or any Affiliate thereof unless such Clean Line Entity or Affiliate has granted a Lien on its rights of payment in respect of such Indebtedness pursuant to Security Documents that are in form and substance acceptable to DOE as contemplated by Section 11.6.
- 8.27 <u>Renewable Energy Transmission</u>. At any time during which any Transmission Services Agreements are in effect, the Clean Line Entities shall use all commercially reasonable efforts to ensure that at least 75% of the total Electrical Capacity covered by all Transmission Services Agreement that are then in effect to be covered by Transmission Services Agreements used for the transmission of renewable energy resources; <u>provided</u> that, to the extent the transmission of energy from non-renewable resources is required by Applicable Law (including pursuant to any open access tariff rules), such events would not render the underlying Transmission Services Agreement from being disqualified toward the 75% threshold.

# ARTICLE IX GUARANTEE

#### 9.1 Guarantee of the Obligations.

- (a) Subject to the provisions of Section 9.2, PECL and each PECL Subsidiary (each, a "Subsidiary Guarantor" and, collectively, the "Subsidiary Guarantors") jointly and severally hereby absolutely, irrevocably and unconditionally guarantee to DOE, for the benefit of DOE and each other Covered Party, the due and punctual payment in full of all obligations of Holdings and the Project Subsidiaries under this Agreement, whether direct or indirect, absolute or contingent, when the same shall become due (including amounts that would become due but for the operation of the automatic stay under Section 362(a) of the Bankruptcy Code, 11 U.S.C. § 362(a)) (collectively, the "Guaranteed Obligations") and agrees to pay any and all expenses incurred by DOE in enforcing its rights under this Article IX.
- (b) Each Subsidiary Guarantor and DOE hereby confirms that it is the intention of all such Persons that the guarantee set forth in this Article IX and the Guaranteed Obligations of each Subsidiary Guarantor hereunder not constitute a fraudulent transfer or conveyance for purposes of Bankruptcy Law, the Uniform Fraudulent Conveyance Act and the Uniform Fraudulent Transfer Act. To effectuate the foregoing intention, DOE and the Subsidiary Guarantors hereby irrevocably agree that

the obligations of each Subsidiary Guarantor under this Article IX at any time shall be limited to the maximum amount as will result in the obligations of such Subsidiary Guarantor under this Article IX not constituting a fraudulent transfer or conveyance.

- 9.2 <u>Contribution by Subsidiary Guarantors</u>. All Subsidiary Guarantors desire to allocate among themselves (collectively, the "<u>Contributing Subsidiary Guarantors</u>"), in a fair and equitable manner, their obligations arising under this Guarantee. Accordingly, in the event of any payment or distribution is made on any date by a Contributing Subsidiary Guarantor (a "<u>Funding Subsidiary Guarantor</u>") under this Guarantee such that its Aggregate Payments exceeds its Fair Share as of such date, such Funding Subsidiary Guarantor shall be entitled to a contribution from each of the other Contributing Subsidiary Guarantors in an amount sufficient to cause each Contributing Subsidiary Guarantor's Aggregate Payments to equal its Fair Share as of such date. The amounts payable as contributions hereunder shall be determined as of the date on which the related payment or distribution is made by the applicable Funding Subsidiary Guarantor. The allocation among Contributing Subsidiary Guarantors of their obligations as set forth in this Section 9.2 shall not be construed in any way to limit the liability of any Contributing Subsidiary Guarantor hereunder. Each Guarantor is a third party beneficiary to the contribution agreement set forth in this Section 9.2.
- 9.3 Payment by Subsidiary Guarantors. Subject to Section 9.2, the Subsidiary Guarantors hereby jointly and severally agree, in furtherance of the foregoing and not in limitation of any other right which Holdings or the Project Subsidiaries may have at law or in equity against any Subsidiary Guarantor by virtue hereof, that upon the failure of Holdings or the Project Subsidiaries to pay any of the Guaranteed Obligations when and as the same shall become due (including amounts that would become due but for the operation of the automatic stay under Section 362(a) of the Bankruptcy Code, 11 U.S.C. § 362(a)), Subsidiary Guarantors will upon demand pay, or cause to be paid, in cash, to DOE, an amount equal to the sum of the unpaid amount of all Guaranteed Obligations then due plus accrued and unpaid interest on such Guaranteed Obligations at the Default Rate (including interest which, but for Holdings or the Project Subsidiaries becoming the subject of a case under the Bankruptcy Code, would have accrued on such Guaranteed Obligations, whether or not a claim is allowed against Holdings or the Project Subsidiaries for such interest in the related bankruptcy case) and all other Guaranteed Obligations then owed to Holdings or the Project Subsidiaries as aforesaid.
- 9.4 <u>Guarantee Absolute</u>. Each Subsidiary Guarantor guarantees that the Guaranteed Obligations will be paid strictly in accordance with the terms of this Agreement, regardless of any law, regulation or order now or hereafter in effect in any jurisdiction affecting any of such terms or the rights of any Covered Party with respect thereto. The obligations of each Guarantor under this Article IX are independent of the Guaranteed Obligations or any other obligations of any Clean Line Entity under or in respect of this Agreement and the other Transaction Documents, and a separate action or actions may be brought and prosecuted against each Subsidiary Guarantor to enforce this guarantee under this Article IX, irrespective of whether any action is brought against any Clean Line Entity or whether any Clean Line Entity is joined in any such action or actions.
- 9.5 <u>Liability of Guarantors Absolute; Waivers</u>. Each Subsidiary Guarantor agrees that its obligations hereunder are irrevocable, absolute, independent and unconditional and shall

not be affected by any circumstance which constitutes a legal or equitable discharge of a guarantor or surety other than payment in full of the Guaranteed Obligations (other than contingent or indemnification obligations for which no claim has been made) or valid release of a Subsidiary Guarantor with DOE's consent. Each Subsidiary Guarantor hereby waives, for the benefit of DOE and each Covered Party: (a) any right to require DOE, as a condition of payment or performance by such Subsidiary Guarantor, to (i) proceed against Holdings or any Project Subsidiary, any other guarantor (including any other Subsidiary Guarantor) of the Guaranteed Obligations or any other Person, (ii) proceed against or exhaust any Collateral or other security held by DOE or (iii) pursue any other remedy in the power of DOE whatsoever, (b) any defense arising by reason of the incapacity, lack of authority or any disability or other defense of Holdings or any Project Subsidiary or any other Subsidiary Guarantor including any defense based on or arising out of the lack of validity or the unenforceability of the Guaranteed Obligations or any agreement or instrument relating thereto or by reason of the cessation of the liability of Holdings or any Project Subsidiary or any other Subsidiary Guarantor from any cause other than payment in full of the Guaranteed Obligations, (c) any defense based upon any statute or rule of law which provides that the obligation of a surety must be neither larger in amount nor in other respects more burdensome than that of the principal, (d) any defense based upon DOE's errors or omissions in the administration of the Guaranteed Obligations, except for errors or omissions determined in the final, non-appealable judgment of a court of competent jurisdiction to have resulted directly and primarily from DOE's or any Covered Party's gross negligence or willful misconduct, (e) (i) any principles or provisions of law, statutory or otherwise, which are or might be in conflict with the terms hereof and any legal or equitable discharge of such Subsidiary Guarantor's obligations hereunder, (ii) the benefit of any statute of limitations affecting such Subsidiary Guarantor's liability hereunder or the enforcement hereof, (iii) any rights to set-offs, recoupments and counterclaims, and (iv) promptness, diligence and any requirement that DOE protect, secure, perfect or insure any security interest or Lien or any Property subject thereto, (f) notices, demands, presentments, protests, notices of protest, notices of dishonor and notices of any action or inaction, including acceptance hereof, notices of default hereunder or any agreement or instrument related thereto, notices of any renewal, extension or modification of the Guaranteed Obligations or any agreement related thereto, notices of any extension of credit to Holdings or any Project Subsidiary, (g) any duty on the part of DOE to disclose to such Subsidiary Guarantor any matter, fact or thing relating to the business, condition (financial or otherwise), operations, performance, Properties or prospects of Holdings or any Project Subsidiary now or hereafter Known by DOE, (h) any defenses or benefits that may be derived from or afforded by law which limit the liability of or exonerate guarantors or sureties, or which may conflict with the terms hereof, (i) any defenses related to any change in the time, manner or place of payment of, or in any other term of, all or any of the Guaranteed Obligations or any other amendment or waiver of or any consent to departure from the Transaction Documents, including any increase in the Guaranteed Obligations, (j) any defense related to the taking, exchange, release or non-perfection of any Collateral or any manner of application of the Collateral or the proceeds thereof to all or any of the Guaranteed Obligations, or any manner of sale or other disposition of any Collateral for all or any of the Guaranteed Obligations, (k) any defense related to any change, restructuring or termination of the corporate structure or existence of Holdings or any Project Subsidiary, (1) any defense related to the failure of any other Person to execute or deliver the guarantee under this Article IX or the release or reduction of liability of any Subsidiary Guarantor or other guarantor or surety with respect to the Guaranteed Obligations

- and (m) defenses related to any other circumstance or any existence of or reliance on any representation by DOE that might otherwise constitute a defense available to, or a discharge of, Holdings, any Project Subsidiary or any other guarantor or surety.
- 9.6 <u>Continuing Guarantee</u>; <u>Assignments</u>. The guarantee in this Article IX is a continuing guarantee and shall (a) remain in full force and effect until the latest of (i) the payment in full in cash of the Guaranteed Obligations and all other amounts payable under this Article IX and (ii) the Termination Date, (b) be binding upon each Subsidiary Guarantor, its successors and assigns and (c) inure to the benefit of and be enforceable by the Covered Parties and their successors, transferees and assigns.
- 9.7 <u>Obligations and Rights of Subsidiary Guarantors</u>. Except to the extent set forth in this Article IX and rights that inure to all Parties under this Agreement, no Subsidiary Guarantor shall have any rights under this Agreement.

# ARTICLE X FORCE MAJEURE

10.1 <u>Force Majeure</u>. To the extent a Party is delayed or prevented by Force Majeure from performing, in whole or in part, its obligations under this Agreement, and such Party (a "<u>Claiming Party</u>") gives written notice and details of such Force Majeure to the other Party as soon as reasonably practicable after such Party becomes aware of the occurrence of such Force Majeure, then the Claiming Party shall be excused from the performance of its obligations under this Agreement (other than any obligation of the Clean Line Parties to make payments to DOE under the Transaction Documents, including under Sections 11.1, 11.3 and 11.4 and Article IX) during such Force Majeure, but for no longer period and only to the extent performance of such obligations are prevented or delayed by such Force Majeure. The Claiming Party shall exercise due diligence to remedy such Force Majeure within a reasonable period, at the Clean Line Entities' cost in all cases. The occurrence of a Force Majeure shall not entitle any Party to terminate this Agreement.

# ARTICLE XI COST AND EXPENSE FUNDING, ADVANCE FUNDING, INDEMNITY AND COLLATERAL

11.1 Cost and Expense Funding. Each of the Clean Line Entities, jointly and severally, agree to pay for (a) all out-of-pocket costs, expenses and disbursements incurred by DOE (including the costs, expense and disbursements of all internal personnel, consultants, advisors, agents and counsel engaged by DOE) in connection with the development, preparation, negotiation and execution of, and any amendment, supplement or modification to, this Agreement, the other Transaction Documents, the Real Estate Rights Agreements and any other documents prepared in connection therewith and in connection with the Project, (b) all costs, expenses, fees, Taxes, disbursements and payments made by DOE (including the costs, expenses and disbursements of surveyors, appraisers, agents, consultants, advisors and counsel (including Shearman & Sterling LLP) engaged by DOE, and all Local Government Contribution Payments) in connection with the acquisition by DOE of any Project Real Estate Rights, (c) all costs (including the costs, expense and disbursements of all internal personnel, consultants, advisors,

agents and counsel engaged by DOE) incurred by DOE in connection with the administration, inspection, enforcement, defense or preservation of any rights or claims under this Agreement, the other Transaction Documents and the Real Estate Rights Agreements, (d) any costs incurred by DOE in connection with its performance or undertaking of any non-delegable obligations or responsibilities under the DOE Mitigation Action Plan, any Cultural Resource Agreement with NHPA, the Endangered Species Act or any other Applicable Law, and (e) any other costs, expenses and disbursements incurred by DOE in connection with its participation in the Project, including, any costs and expenses associated with any of the Wind-Up Events or otherwise arising as a result of DOE's exercise of any rights or remedies under this Agreement, any other Transaction Document, any Real Estate Rights Agreement or any expenditures which DOE deems necessary to protect its and the public's interest in respect of the Project (collectively, the "Covered Costs"). Without limiting the above, Covered Costs shall include all costs and expenses that would not have been incurred by DOE but for its participation in the Project.

Participation Amount. Commencing on and after the Project Completion, Holdings shall pay to DOE at the end of each fiscal quarter an amount equal to 2% of the gross revenues received by the Clean Line Parties from the Project during such fiscal quarter resulting from the sale of transmission service in connection with the Project (as such gross revenue amount is reflected in Holdings' Financial Statements for such fiscal quarter, including, with respect to the first such fiscal quarter, sales of transmission service which occurred at any time prior to Project Completion) (the "Participation Amount"). The Clean Line Parties shall only be required to pay the Participation Amount after (a) the payment of operating costs and expenses in respect of the Project then due and debt service in respect of the Project Financing then due, (b) the funding of a customary debt service reserve account in favor of the Project Financing Parties under the Project Financing and (c) the funding of the Capital Repairs Reserve Account then required, for such quarterly period; provided that such amounts shall not be deducted from gross revenues for purposes of calculating the Participation Amount due and payable pursuant to this Section 11.2. The Participation Amounts shall be made available to DOE to offset costs associated with federal hydropower infrastructure or for any other authorized purpose.

#### 11.3 Advance Cost Funding.

- (a) Within ten (10) Business Days following the Effective Date, Holdings shall deposit, or cause to be deposited, an amount equal to the Required Amount, as notified by DOE to Holdings on or before the Effective Date, into the account specified by DOE in such notice (the "Advance Funding Account").
- (b) No later than fifteen (15) Business Days prior to the end of each fiscal quarter of Holdings, DOE shall deliver to Holdings a request for funding of the Advance Funding Account in an amount equal to the then applicable Required Amount (which request shall include a calculation of the Required Amount and shall take into consideration amounts already on deposit in the Advance Funding Account). In addition, if at any time, DOE shall determine that the amount on deposit in the Advance Funding Account is less than the Base Amount applicable as of such time, DOE shall have the right to deliver to Holdings an additional request for funding of the Advance Funding Account in an amount equal to the then applicable Base Amount (including a calculation of such Base Amount). Within fifteen (15) Business Days of Holdings' receipt of any

request for funding of the Advance Funding Account from DOE, Holdings shall cause the applicable Required Amount or Base Amount to be deposited into the Advance Funding Account.

- (c) DOE will provide Holdings quarterly statements of the Covered Costs expended by DOE and reasonable supporting documentation of such expenditures within thirty (30) days of the close of each quarter.
- (d) Subject to limits established under Applicable Law, Holdings shall have the right to conduct, at its own expense, reasonable audits of the books, records, and documents of DOE relating to the items on any particular accounting statement provided by DOE.
- (e) DOE agrees to account for its costs incurred pursuant to this Agreement under an accounting procedure in customary usage for accounting of Federal project expenses. Holdings shall have the right to audit DOE's cost records and accounts to verify statements of costs submitted by DOE. DOE agrees to refund any amounts paid if they are found in such audit to exceed the total amount due DOE for its actual costs incurred pursuant to this Agreement without any penalty or interest. The Clean Line Entities agree that such audit of DOE's records and accounts is for the sole purpose of verifying that an accounting statement sets forth the actual costs as reflected by the records, and that accounts are maintained in accordance with the established accounting procedures.
- (f) DOE may withdraw at any time from the Advance Funding Account amounts necessary to pay for Covered Costs incurred by DOE.
- (g) DOE agrees not to enter into any Contractual Obligations in connection with the Project that contain amounts payable from time to time by DOE over a period in excess of five (5) future years except to the extent that (i) Holdings has approved any such Contractual Obligation, (ii) such Contractual Obligation is for an initial five (5) year term but has an extension option that can be exercised by, or consented to by, DOE in its sole discretion to extend the length of such Contractual Obligation prior to the termination thereof for additional rolling periods of up to five (5) years or (iii) DOE's obligations to make payments for any period that is later than five (5) years after any date of determination are subject to the availability of funding to DOE with which to make payments in respect thereof.

#### 11.4 Indemnification.

(a) Each Clean Line Entity shall, jointly and severally, indemnify and defend each Covered Party against, and hold each of them harmless from, any and all Covered Liabilities, including, any Covered Liabilities incurred by any Covered Party as a result of any investigation, Action or inquiry (whether or not such Covered Party is a party thereto) related to DOE's entry into and performance under each Transaction Document and its participation in the Project or otherwise arising as a result of the Clean Line Parties operations and business; provided, however, that a Covered Party will not be

indemnified for any Covered Liability to the extent (i) determined in the final, non-appealable judgment of a court of competent jurisdiction to have resulted directly and primarily from such Covered Party's gross negligence or willful misconduct or (ii) resulting from a claim brought by or on behalf of any Clean Line Entity against such Covered Party for breach in bad faith of such Covered Party's obligations hereunder or under any other Transaction Document, if such Clean Line Party has obtained a final and non-appealable judgment in its favor in respect of such claim as determined by a court of competent jurisdiction.

- (b) In the case of any Covered Liability indemnified by the Clean Line Entities that is covered by a policy of insurance maintained by the Clean Line Entities or by third parties pursuant to the Project Contracts, DOE agrees to cooperate, at the Clean Line Entities' expense, with the insurers in the exercise of their rights to investigate, defend or compromise such Covered Liability as may be required to retain the benefits of such insurance with respect to such Covered Liability.
- DOE shall, promptly after it has any actual knowledge thereof, notify Holdings of any Covered Liability as to which indemnification is sought; provided, however, that the failure to deliver such prompt notice shall not release any Clean Line Entity from any of its obligations to indemnify any Covered Party. Any Covered Liability payable to any Covered Party shall be paid on or prior to the date which is the later of (i) the date ten (10) days after receipt by Holdings of a written demand therefor from DOE accompanied by a written statement describing in reasonable detail each Covered Liability that is the subject of, and the basis for, such indemnity and the computation of the amount so payable and (ii) the date two (2) Business Days prior to the date on which any Covered Liability is payable. Subject to the rights of insurers under policies of insurance maintained by the Clean Line Entities, the Clean Line Entities may, unless an Event of Default shall have occurred and be continuing, with respect to any Covered Liability for which indemnification is sought and for which they shall have acknowledged liability to the DOE in writing, and (if requested by DOE in writing after any such acknowledgment of liability has been given by the Clean Line Entities) at the sole cost and expense of the Clean Line Entities, investigate and, if permitted by Applicable Law, defend any Covered Liability for which indemnification is sought with counsel reasonably acceptable to DOE, and DOE shall cooperate, at the Clean Line Entities' expense, with all reasonable requests of the Clean Line Entities in connection therewith; provided that in the event that in the course of the investigation or defense of any such Covered Liability, the Clean Line Entities shall in good faith reasonably determine that they are not liable for indemnification with respect thereto notwithstanding such acknowledgment of liability, Holdings may give notice to the DOE of such fact and, in such case, any acknowledgment theretofore made by the Clean Line Entities of their liability with respect to such Covered Liability shall be deemed revoked, and the Clean Line Entities may thereupon cease to defend such Covered Liability; provided that (A) Holdings shall have given DOE reasonable prior written notice of the Clean Line Entities' intention to renounce such acknowledgment, (B) the Clean Line Entities' conduct regarding the defense of such Covered Liability or any decision to withdraw from such defense shall not materially prejudice or have materially prejudiced DOE's ability to contest such Covered Liability (taking into account, among other things,

the timing of the Clean Line Entities' withdrawal and the theory or theories upon which the Clean Line Entities shall have based their defense) and (C) the Clean Line Entities shall have given DOE all materials, documents and records relating to their defense of such Covered Liability as DOE shall have requested in connection with the assumption by DOE of the defense of such Covered Liability at the cost and expense of the Clean Line Entities. If the Clean Line Entities shall cease to defend any Covered Liability pursuant to the preceding sentence, the Clean Line Entities shall indemnify the Covered Parties to the extent that the actions of the Clean Line Entities in defending such Covered Liability or the manner or the time of the Clean Line Entities' election to withdraw from the defense of such Covered Liability shall have caused any Covered Party to incur any cost, loss, liability or expense which such Covered Party would not have incurred had the Clean Line Entities not assumed and thereafter ceased the defense of such Covered Liability in such manner or at such time.

Notwithstanding the foregoing provisions, the Clean Line Entities shall not be entitled to assume and control the defense of any such Covered Liability if an Event of Default has occurred and is continuing or if such Covered Liability involves or could reasonably be expected to result in, in the sole judgment of DOE, (i) an Action involving a possible imposition of any criminal liability or penalty or civil penalty on any Covered Party, (ii) the granting of injunctive relief against any Covered Party affecting Property or activity not related to the transactions contemplated by the Transaction Documents, or (iii) a conflict of interest between any of the Covered Parties and the Clean Line Entities or a risk of the sale, forfeiture or loss of any of the Project Facilities or Project Real Estate Rights or any material portion thereof or interest therein, and, in either case, DOE informs Holdings that such Covered Party desires to be represented by separate counsel, in which case the fees and expenses of such separate counsel shall be borne by the Clean Line Entities. The Clean Line Entities shall provide to any Covered Party as to which any Covered Liability has been or may be asserted, such documents and other information relating thereto as such Covered Party may reasonably request from time to time. The Clean Line Entities shall also assist and testify in all proceedings at the request of DOE even if no Clean Line Entity is involved in such proceedings, all at the Clean Line Entities' cost. The Clean Line Entities shall not enter into any compromise or settlement of Covered Liability if such settlement would have any unindemnified adverse effect on the Project, the Project Facilities, the Project Real Estate Rights or the ability of the Clean Line Entities to perform their obligations under any of the Transaction Documents or Real Estate Rights Agreements or would require any Covered Party to admit any wrongdoing on its part, without the prior written consent of DOE and each No Covered Party shall be entitled to indemnification applicable Covered Party. hereunder with respect to any Covered Liability with respect to which it shall have entered into any settlement or other compromise, unless it shall theretofore have given notice to Holdings of such Covered Liability and the material facts relating thereto, and the Clean Line Entities shall have had a period which shall end on the earlier of (A) the thirtieth (30<sup>th</sup>) day after the date of delivery to Holdings of such notice and (B) the later of the date on which the relevant offer of settlement or compromise shall have expired and the tenth (10<sup>th</sup>) day after the date of delivery to Holdings of such notice, in which to acknowledge liability pursuant to the preceding section. No such consent shall be

required if (x) an Event of Default shall have occurred and be continuing or (y) such Covered Party waives its right to be indemnified with respect to such Covered Liability.

#### 11.5 Performance Support.

- Commencing on the Commencement Date and throughout the term of this Agreement, Holdings shall provide DOE Performance Support, and maintain in full force and effect such Performance Support, in an amount equal to the then Applicable Amount. DOE shall be entitled to make a drawing or demand payment under any such Performance Support in respect of any of the following: (i) to pay any Project Costs (including any costs or expenses associated with any Wind-Up Event) or other payment obligations in respect of the Project to the extent any Clean Line Party has failed to do so, (ii) to pay any Covered Costs to the extent that there are insufficient funds available to DOE in the Advance Funding Account with which to make payment of any such Covered Costs, (iii) as payment of any Covered Liability which any Clean Line Party has otherwise failed to make in accordance with the terms of this Agreement, including the terms set forth under Section 11.4, (iv) to pay any Capital Repairs to the extent that there are insufficient funds available to the Clean Line Entities in the Capital Repairs Reserve Account or to the extent the Clean Line Entities fail to make such Capital Repairs, following the notice and cure period specified in Section 4.8(b), (v) in all other cases in which any Clean Line Party is obligated to make a payment to DOE or any Covered Party pursuant to the terms of this Agreement and has failed to make any such payment within fifteen (15) Business Days of a request for such payment, in the amount of any such payment and (vi) under any other circumstances expressly contemplated by such Performance Support (including as a result of the provider thereof no longer constituting an Acceptable Support Provider, in which case DOE shall hold such funds in trust and apply such funds in the same manner as permitted in respect of the Performance Support). Following a draw made under the Performance Support which reduces the amount available for drawing thereunder (or in the case where the full amount of the Performance Support has been drawn and the proceeds thereof are placed in a DOE or U.S. Treasury account or a collateral account pledged solely to DOE as a result of either the applicable provider of such Performance Support no longer constituting an Acceptable Support Provider or the pending expiration of such Performance Support, following any application of funds in such account), Holdings shall replenish or reinstate such Performance Support to the then Applicable Amount within fifteen (15) Business Days following notice from DOE.
- (b) Subject to draw or demand provisions relating to the termination or expiration of the applicable Performance Support as provided in such Performance Support or the failure of the provider of such Performance Support to be an Acceptable Support Provider (in which case DOE shall be entitled to draw on, or demand payment under, such Performance Support subject to the grace periods provided in such Performance Support), DOE shall not make a drawing or demand under the Performance Support if funds in the Advance Funding Account or Wind-Up Reserve Account (other than funds which have been allocated for a specific purpose) are available to satisfy any payment obligation of the Clean Line Entities. Prior to making any drawing or demand on the Performance Support, DOE shall provide to Holdings ten (10) days advance

written notice; <u>provided</u> that no such notice shall be required in connection with any drawing or demand arising as a result of the pending termination or expiration of the applicable Performance Support or the failure of the provider of such Performance Support to be an Acceptable Support Provider.

#### 11.6 Collateral.

- (a) To secure their obligations under this Agreement, the Clean Line Entities shall grant for the benefit of DOE:
  - (i) (A) on or prior to Project Completion, a first priority perfected security interest in the Wind-Up Reserve Account and all funds on deposit in, or credited in each such account from time to time and (B) on or prior to Project Completion, a perfected security interest (which, prior to Project Financial Close, shall be a first priority Lien and, from and after Project Financial Close, to the extent the Clean Line Entities have granted any security interest to any Financing Party, shall be a second priority Lien) on the Capital Repairs Reserve Account (collectively, the "Account Collateral");
  - (ii) on or before the Commencement Date, a perfected security interest (which, prior to Project Financial Close, shall be a first priority Lien and, from and after Project Financial Close, to the extent the Clean Line Entities has granted any security interest to any Financing Party, shall be a second priority Lien) on 100% of the Equity Interests in ACL and any Indebtedness owed by ACL to Holdings, PECL or any other Subsidiary or Affiliate of Holdings (including any Person making any equity investment in ACL as part of a Permitted Project Investment) from time to time (collectively, the "Equity Collateral"); and
  - (iii) concurrently with the grant by the Clean Line Entities of any security interest to any Financing Party, a second priority Lien on all of the Clean Line Entities' Properties and assets (including the Capital Repairs Reserve Account but excluding other Account Collateral and the Other Facilities) in which the Clean Line Entities have granted a security interest for the benefit of such Financing Parties (the "Second Lien Collateral").
- (b) The Collateral shall be free and clear of any other security interest or Lien, except for Permitted Liens.
- 11.7 <u>Intercreditor Agreement</u>. In connection with the closing of the Project Financing, DOE shall agree to enter into an intercreditor agreement (the "<u>Intercreditor Agreement</u>") with the Financing Parties pursuant to which it shall agree to subordinate its security interest in the Capital Repairs Reserve Account, the Equity Collateral and Second Lien Collateral under the Security Documents to the security interest in the Capital Repairs Reserve Account, the Equity Collateral and Second Lien Collateral created for the benefit of the Financing Parties. The Intercreditor Agreement will be on usual and customary terms for transactions with a first priority Lien and junior Lien, including (a) a provision whereby DOE shall agree (i) to a 365-day standstill period (which shall be extended for so long as the Project Financing Parties are

exercising remedies) with respect to taking any action in respect of any shared Collateral and (ii) to not object to the use of cash collateral and (b) provisions for debtor in possession financings and adequate protection payments for the senior lien holders.

#### 11.8 Limitation of Liability to DOE.

- (a) None of the Covered Parties shall have any liability whatsoever for payment of any obligations incurred by any Clean Line Obligor or any of its Affiliates or any other Person in connection with the Project, the Other Facilities, the Project Contracts, the Project Financing Documents or any of the transactions contemplated thereby. Each Clean Line Party hereby waives on behalf of itself and its Affiliates all rights to assert any claims against DOE or any of the Covered Parties in connection with the Project and the Other Facilities on the basis of breach of contract, misrepresentation, tort, detrimental reliance, promissory estoppel or any other legal principle in the event that (i) a final and non-appealable Governmental Order finds that DOE is legally prohibited from participating in the Project or performing its obligations under the Transaction Documents, (ii) if there is a Change in Law that sets aside or legally prohibits DOE's participation in the Project or (iii) the Project is delayed in any respect as a result of any Action or Governmental Order that affects DOE's ability to comply with its undertakings hereunder; provided that DOE itself shall agree to not take the position that it lacks the statutory authority to participate in the Project.
- (b) DOE's review of any of the Project Contracts or the Project Financing Documents shall not be considered to be a guaranty or endorsement of any of the terms thereof or of any of the obligations of the Clean Line Obligors or any other Person thereunder or any information provided by the Clean Line Obligors or any other Person in connection with the negotiation, execution and delivery or performance of the Project Contracts or the Project Financing Documents (including any projections or other financial information provided in connection therewith) and is not a representation, warranty or other assurance as to the ability of the Clean Line Entities or any other Person party to the Project Contracts or the Project Financing Documents to perform their obligations thereunder.

# 11.9 <u>Consequential Damages</u>.

- (a) No claim shall be made by any Clean Line Obligor or any of its Affiliates or representatives against DOE or any other Covered Party for any special, indirect, consequential or punitive damages (whether or not the claim therefor is based on contract, tort or duty imposed by law) in connection with, arising out of or in any way related to the transactions contemplated by this Agreement, the other Transaction Documents, the Project, the Other Facilities or any act or omission or event occurring in connection therewith, and each Clean Line Obligor and each of its Affiliates hereby waives and releases any such claim for any such damages, whether or not accrued and whether or not Known or suspected to exist in its favor.
- (b) No claim shall be made by DOE or any other Covered Party against any Clean Line Obligor or any of its Affiliates for any special, indirect, consequential or

punitive damages (whether or not the claim therefor is based on contract, tort or duty imposed by law) in connection with, arising out of or in any way related to the actions contemplated by this Agreement, the other Transaction Documents or any act or omission or event occurring in connection therewith, and DOE and/or any other applicable Covered Party shall waive and release any such claim for any such damages, whether or not accrued and whether or not Known or suspected to exist in its favor; provided that nothing in this paragraph shall limit each Clean Line Entity's indemnity obligations as set forth in Section 11.4 to the extent that such special, indirect, consequential or punitive damages are included in any claim by a third party unaffiliated with any Clean Line Obligor with respect to which DOE or any other Covered Party is entitled to indemnification.

#### 11.10 Release Provision.

The Clean Line Entities shall use commercially reasonable efforts to include in the Project Financing Documents and in each Material Project Contract (other than any Interconnection Agreement) that the Clean Line Entities or DOE enters into in respect of the Project a provision pursuant to which the Project Financing Parties or such Project Participant, as applicable, shall agree that such Person has no recourse to any Covered Party under such Project Financing Document (other than in respect of DOE's express obligations or undertakings pursuant to the Transaction Documents, the Intercreditor Agreement and/or the DOE Direct Agreement) or applicable Material Project Contract (other than any Interconnection Agreement) and shall expressly release any Covered Party from any claim, liability or other obligation under such Project Financing Document or the applicable Material Project Contract (other than any Interconnection Agreement) (the "Release Provision"). The Release Provision included in the Project Financing Documents or any applicable Material Project Contract (other than any Interconnection Agreement) shall be in form and substance acceptable to DOE; provided that a provision substantially similar to the following shall be deemed to be in form and substance acceptable to DOE:

"Each of the parties hereby, in consideration of \$1000, receipt of which is hereby acknowledged, releases and waives any and all claims, remedies or rights against the [Covered Parties] with respect to any and all liabilities (including, without, limitation, any liabilities arising as a result of negligence, warranty, statutory, product, strict or absolute liability, liability in tort or otherwise), obligations, losses, settlements, damages, penalties, fines, sanctions, taxes, claims, actions, demands, suits, judgments or proceedings of any kind and nature, costs, payments, expenses and disbursements (including fees and expenses of consultants, advisors, external counsel and allocable fees and expenses of internal personnel and attorneys) of whatsoever kind and nature (whether or not any of the transactions contemplated by [this Agreement] are consummated), imposed on, incurred or suffered by, or asserted against such party in any way relating to or arising out of [this Agreement, the Project] or the transactions contemplated hereby; for all purposes of this

provision, the [Covered Parties] shall be deemed to be third party beneficiaries in all respects."

- (b) The bracketed language in the foregoing provision may be conformed as necessary to reflect the terms and provisions of the relevant Project Financing Document or Material Project Contract; provided that any such changes shall not narrow the scope of the Release Provision, except that (i) the Clean Line Entities shall be permitted to add an appropriate exception to the Release Provision in the Project Financing Documents for any direct and express obligations or undertakings made by DOE in the Transaction Documents, the Intercreditor Agreement and/or the DOE Direct Agreement in favor of such Project Financing Party and (ii) the Clean Line Entities shall be permitted to add an appropriate exception to the Release Provision in any Material Project Contract for any direct and express obligations or undertakings by DOE in favor of the applicable Project Participant under such Material Project Contract (and/or under any related agreement entered into between DOE and the applicable Project Participant).
- (c) None of the Clean Line Parties shall enter into any Project Financing or into any Material Project Contract (other than an Interconnection Agreement) unless the applicable Project Financing Documents or Material Project Contract (other than an Interconnection Agreement) includes a Release Provision and none of the Clean Line Parties shall agree to amend, modify or otherwise waive any Release Provision included in the Project Financing Documents or Material Project Contract without the consent of DOE. If Clean Line Entities are unable to obtain a Release Provision for a Material Project Contract, then no Clean Line Entity shall enter into such Material Project Contract without (i) delivering to DOE a substantially final draft of such Material Project Contract and (ii) receiving from DOE its written consent for the Clean Line Entities to enter into such Material Project Contract (not to be unreasonably withheld or delayed); provided, however, that DOE shall consent to such Material Project Contract if it is satisfied that such Material Project Contract does not include any terms or conditions that could reasonably be expected to expose DOE or any other Covered Party to any material obligation or liability.

## ARTICLE XII REPRESENTATIONS AND WARRANTIES

12.1 <u>Representations and Warranties of the Clean Line Parties</u>. Each Clean Line Party makes the following representations and warranties, as applicable, to and in favor of DOE as of the Effective Date, on each of the dates required to be made pursuant to Article VI and on any other date on which any such representation and warranty is required to be made (or deemed to be made) by the express terms of any other Transaction Document or any notice or other document or instrument required to be delivered to DOE pursuant to the terms hereof or the terms of any other Transaction Document (any of the foregoing dates, a "<u>Representation Date</u>"):

### (a) Organization.

(i) Holdings is a limited liability company organized, validly existing and in good standing under the laws of the State of Delaware. PECL is a limited

liability company organized, validly existing and in good standing under the laws of the State of Arkansas. ACL is a limited liability company organized, validly existing and in good standing under the laws of the State of Delaware. PECL OK is a limited liability company organized, validly existing and in good standing under the laws of the State of Oklahoma. OLA is a limited liability company organized, validly existing and in good standing under the laws of the State of Delaware. Each other Subsidiary of Holdings that is a party hereto from time to time after the Effective Date is a limited liability company, corporation or limited liability partnership duly constituted, validly organized or formed, as applicable, and existing and in good standing under the laws of the jurisdiction of its organization or formation, as applicable.

- (ii) Each Clean Line Party (A) is duly qualified and in good standing under the laws of each jurisdiction where its business requires such qualification, except where the failure to be so qualified could not reasonably be expected to result in a Clean Line Material Adverse Effect and (B) has all requisite power and authority to (1) conduct its business, (2) to own or hold under lease its Properties, (3) to execute, deliver, and perform its obligations under this Agreement and any other Clean Line Document to which it is a party as of the applicable Representation Date, and (4) to grant the Liens contemplated by any of the Security Documents to which it is a party as of the applicable Representation Date.
- (b) Authorization. Each of the Clean Line Parties has duly authorized, executed and delivered this Agreement and any other Clean Line Document to which it is a party that has been executed and delivered by it as of as of the applicable Representation Date. No Governmental Approval or any approval, consent, exemption, authorization, or other action by, or notice to, or filing with, any other Person is necessary or required in connection with (i) the execution and delivery by any applicable Clean Line Party of this Agreement, (ii) to the extent applicable as of the applicable Representation Date, the grant by such Clean Line Entity of the Liens granted by it pursuant to the Security Documents or (iii) to the extent applicable, the perfection or maintenance of the Liens created under the Security Documents (including the first or second priority nature thereof, as applicable, and in respect of the Second Lien Collateral, only to the extent that the first priority security interests in favor of the Project have been established and/or perfected), except in each case for such Governmental Approvals or other authorizations, approvals, actions, notices and filings as which have been duly obtained, taken, given or made and are in full force and effect. Except as set forth on Schedule 11 (as such Schedule may be updated in accordance with Section 12.3), as of the applicable Representation Date, the Clean Line Entities have obtained, taken, given or made, as applicable, all Governmental Approvals or any approval, consent, exemption, authorization, or other action by, or notice to, or filing with, any other Person that is necessary or required in connection with the performance by any applicable Clean Line Party of this Agreement and the other Transaction Documents in light of the current stage of construction, management and/or operation of the Project as of such Representation Date, except where the failure to obtain, take, give or make could not reasonably be

expected to result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.

- (c) No Conflict. Neither the execution and delivery of this Agreement nor any other Clean Line Document to which the applicable Clean Line Party is a party as of the applicable Representation Date, nor the performance by such Clean Line Party of any of its obligations thereunder does or will: (i) contravene, conflict with or violate any provision in any Organizational Document or any other agreement relating to the management or affairs of such Clean Line Party, (ii) except as could not reasonably be expected to result in a Clean Line Material Adverse Effect, contravene, conflict with, violate or fail to comply with any final and non-appealable Governmental Order or Governmental Approval applicable to such Clean Line Party or any Applicable Law, (iii) except as could not reasonably be expected to result in a Clean Line Material Adverse Effect, contravene or result in any breach or default under any Contractual Obligation to which the applicable Clean Line Party is a party or by which it or any of its Properties may be bound, or (iv) result in, or require the creation or imposition of any Lien (other than Permitted Liens) upon or with respect to any Properties of the applicable Clean Line Party now owned or hereafter acquired.
- (d) <u>Binding Obligation</u>. Each Clean Line Document that has been executed and delivered as of the applicable Representation Date is a valid and binding obligation of the applicable Clean Line Party enforceable against it in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws in effect from time to time that affect creditors' rights generally and by legal and equitable limitations on the availability of specific remedies.
- (e) Capitalization. All of the Equity Interests of PECL, ACL, PECL OK and OLA have been duly authorized, validly issued, are fully paid and non-assessable and are owned directly or indirectly by Holdings, and all of the Equity Interests of Holdings have been duly authorized, validly issued, and are fully paid and non-assessable, in each case free and clear of all Liens (other than any Permitted Liens). As of the date of this Agreement, all of the Equity Interests of Holdings are owned directly by CLEP. Except as set forth in Schedule 5 (as such Schedule may be updated in accordance with Section 12.3) hereto there are no outstanding options, warrants or rights for conversion into or acquisition, purchase or transfer of Equity Interests of any Clean Line Party or any agreements or arrangements for the issuance by any Clean Line Party of additional Equity Interests. Except as set forth in Schedule 5 (as such Schedule may be updated in accordance with Section 12.3) hereto, each Clean Line Party does not have outstanding (i) any securities convertible into or exchangeable for its Equity Interests or (ii) any rights to subscribe for or to purchase, or any option for the purchase of, or any agreement, arrangement or understanding providing for the issuance (contingent or otherwise) of, or any call, loan commitment or claims of any character relating to, its Equity Interests.

- (f) <u>Litigation</u>. With respect to any pending or threatened Actions to which DOE is not otherwise a party:
  - (i) Except as set forth in Schedule 9 (as such Schedule may be updated in accordance with Section 12.3) hereto, there is no pending or, to any Clean Line Party's Knowledge, threatened Action (A) that relates to the legality, validity or enforceability of this Agreement or any of the other Transaction Documents, (B) that relates to the Project or (C) that relates to any Clean Line Document to which any Clean Line Party is a party as of the applicable Representation Date which, in the case of clauses (B) or (C), either singly or in the aggregate has had a continuing, or could reasonably be expected to have a Clean Line Material Adverse Effect or otherwise materially and adversely affect the interests of DOE, including in its capacity as an agency of the United States government.
  - (ii) No Clean Line Party has failed to observe, in any material respect, any final and non-appealable Governmental Order that has, or could reasonably be expected to have, a Clean Line Material Adverse Effect. There is no injunction, writ, or preliminary restraining order of any nature issued by a Governmental Authority directing that any transactions contemplated by any of the Transaction Documents not be consummated as herein or therein provided.
  - (iii) No final and non-appealable Governmental Order has been entered against any Clean Line Party that has, or could reasonably be expected to have, a Clean Line Material Adverse Effect or an Adverse DOE Impact.

### (g) Taxes.

- (i) Each Clean Line Party (A) has timely filed or caused to be filed all material federal, state and local Tax returns required by Applicable Law to be filed by it, and each such Tax return was complete and accurate in all material respects, and (B) has timely paid or caused to be paid (1) all material Taxes payable by it that have become due pursuant to such Tax returns and (2) all other material Taxes and assessments payable by it that have become due, in each case, other than those Taxes subject to Contest.
- (ii) The Clean Line Parties do not owe any delinquent Indebtedness to any Governmental Authority of the United States, including in respect of any Tax liability, except to the extent such delinquency is the subject of a Contest or has been resolved, or is in the process of being resolved, with the appropriate Governmental Authority in accordance with the standards of the Debt Collection Improvement Act.
- (h) <u>Fees</u>. No Clean Line Party has any obligation to pay any Person in respect of any finder's, broker's, or other similar fees in connection with this Agreement or any other Transaction Document.

(i) <u>Financial Statements</u>. Each of the Financial Statements delivered to DOE pursuant to the terms of this Agreement has been prepared in accordance with GAAP consistently applied and presents fairly, in all material respects, the financial condition of such Person as of the respective dates of the balance sheets included therein and the results of operations of such Person for the respective periods covered by the statements of income included therein, subject to the absence of notes and normal year-end audit adjustments with respect to the quarterly unaudited financial statements. Except as reflected in such Financial Statements, as of the date of such Financial Statements, there are no material liabilities or obligations of such Person of any nature whatsoever as of the balance sheet date contained in such financial statements that are required to be disclosed in accordance with GAAP, subject to the absence of notes for the quarterly unaudited financial statements.

### (j) Project Plans, Base Case Projections and Sufficiency of Funds.

- (i) The Project Schedule, the Project Plans and the Base Case Projections, as amended or supplemented from time to time in accordance with the terms of this Agreement, when prepared or made (A) were complete and based on assumptions that the Clean Line Entities believed to be reasonable, (B) are consistent with the provisions of any Clean Line Documents and Project Financing Documents then in effect, (C) have been prepared in good faith and with due care and (D) fairly represented the Clean Line Entities' expectation as to the matters covered thereby.
- (ii) The Project Schedule, as amended or supplemented from time to time in accordance with the terms of this Agreement, accurately specifies in summary form the work necessary to reach Project Completion on a specified timeline.
- (iii) The Project Budget as amended or supplemented from time to time in accordance with the terms of this Agreement represents the Clean Line Entities' good faith, best estimate of total Project Costs anticipated to be incurred to construct the Project in the manner contemplated by the Project Plans.
- (k) <u>Immunity</u>. None of the Clean Line Parties and none of their respective Properties enjoys any right of immunity from set off, suit or execution with respect to any Property or obligations under any Transaction Document.
- (l) <u>Compliance with Applicable Laws; Environmental Matters</u>. Except as set forth in Schedule 10 (as such Schedule may be updated in accordance with Section 12.3) hereto:
  - (i) Each Clean Line Party is in compliance with, and has conducted (or caused to be conducted) its business and operations and the business and operations of the Project in compliance with, all Applicable Laws, including Environmental Laws, in each case applicable to such Clean Line Party or the Project, except where the failure to comply could not reasonably be expected to

result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.

- (ii) As of the applicable Representation Date, there are no Environmental Claims pending or, to the Knowledge of any Clean Line Party, threatened against such Clean Line Party, any Property of such Clean Line Party or the Project, except as could not reasonably be expected to result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.
- (iii) As of the applicable Representation Date, to the Knowledge of each Clean Line Party, there are no present or past actions, activities, circumstances, conditions, events or incidents, including the Release of any Hazardous Substances that could reasonably be expected to form the basis of an Environmental Claim against any Clean Line Party or DOE or in respect of the Project Site, except as could not reasonably be expected to result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.
- (iv) As of the applicable Representation Date, none of the Clean Line Parties nor, to the Knowledge of each Clean Line Party, any other Person, has used, released, discharged, generated, manufactured, produced, stored or disposed of in, on, under or about the Project Site or transported thereto or therefrom, any Hazardous Substances that could reasonably be expected to form the basis of an Environmental Claim related to the Project site or cause any of the Clean Line Parties or the Project Site to be subject to any restrictions arising under Environmental Laws or otherwise have a material adverse environmental or social effect which is prohibited under Applicable Law, except as could not reasonably be expected to result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.
- (v) No Clean Line Party has received any letter or request for information under Section 104 of the Comprehensive Environmental Response Compensation and Liability Act (42 U.S.C. Sections 9640 *et seq.*) or comparable state laws, and, to the Knowledge of each of the Clean Line Parties, none of the operations of any of the Clean Line Parties relating to the Project is the subject of any investigation by a Governmental Authority evaluating whether any remedial action is needed to respond to a Release or threatened Release of any Hazardous Substances relating to the Project or the Project Site or at any other location, including any location to which any Clean Line Party has transported, or arranged for the transportation of, any Hazardous Substances with respect to the Project Site.

### (m) Insolvency Events; Solvency.

(i) None of the Clean Line Parties is subject to any pending or to the Knowledge of each of the Clean Line Parties, threatened, Insolvency Event.

- (ii) The Clean Line Parties, on a consolidated basis, are Solvent.
- (n) <u>No Defaults</u>. No Default or Event of Default has occurred and is continuing.

### (o) Full Disclosure.

- (i) All written information contained in all documents, reports or other written information pertaining to the Project (other than any projections or forward-looking statements), together with all written updates of such information from time to time (collectively, the "Information"), that have been furnished by or on behalf of the Clean Line Parties to DOE, are, as of the date such information was so furnished and taken as a whole, true and correct in all material respects and do not contain any material misstatement of fact or omit to state a material fact or any fact necessary to make the statements contained therein not materially misleading in light of the circumstances in which they were made; provided, that with respect to any Information that is expressly identified as being obtained from a publicly-available third-party source, this representation is made only to the Knowledge of the Clean Line Parties.
- (ii) As of the date of this Agreement, to the Knowledge of each of the Clean Line Parties it is technically feasible for the Project to be constructed, completed, operated and maintained in all material respects in accordance with the specifications and other information contained in that 1222 Program Part 2 Application submitted by the Clean Line Parties to DOE in January 2015.

### (p) <u>Security Interests; Liens</u>.

- (i) The Security Documents that have been delivered on or prior to the applicable Representation Date are effective to create, in favor of DOE, a legal, valid and enforceable Lien on and security interest in all of the Collateral purported to be covered thereby, and all necessary recordings and filings with respect to such Security Documents have been made in all necessary public offices, and all other necessary and appropriate action has been taken, so that the security interest created by such Security Document is a perfected Lien on and security interest in all right, title and interest of the applicable Clean Line Entity in the Collateral purported to be covered thereby, prior and superior to all other Liens other than Permitted Liens (provided, with respect to the Second Lien Collateral, the Clean Line Entities shall not be obligated to make any filings or recordings or take any other action necessary to create or perfect a Lien that are not required in respect of the first priority security interest granted in favor of the Project Financing Parties).
- (ii) No Lien (other than a Permitted Lien) or other instrument or recordation covering all or any part of the Collateral purported to be covered by the Security Documents on or prior to the applicable Representation Date is on file in any recording office or public registry.

- (iii) Except for Permitted Liens, no Clean Line Entity has created and is not under any obligation to create, and has not entered into any Contractual Obligation that would, or could reasonably be likely to, result in the imposition of, any Lien upon any of its Properties.
- (q) <u>Insurance</u>. Following adoption of the Insurance Agreement, the Clean Line Entities' insurance coverage for the Project required pursuant to the Insurance Agreement to be in effect at such time is in full force and effect, and all premiums then due and payable under the applicable policies have been paid.

### (r) Business.

- (i) None of the Clean Line Parties has conducted any business, other than the business contemplated by the Transaction Documents and the other Clean Line Documents, the Project Contracts, the Project Equity Commitments, the Project Financing Commitments, the Project Financing Documents and such other business as may be related to the Project and the Other Facilities.
- (ii) None of the Clean Line Parties has any outstanding Indebtedness other than Permitted Indebtedness.
- (iii) None of the Project Subsidiaries owns (A) any real Property rights other than those relating to the Project and (B) any Equity Interests other than Equity Interests in any other Subsidiary of Holdings that is also a Project Subsidiary.
- (iv) None of the Project Subsidiaries is a party to or bound by any Contractual Obligation other than (A) the Transaction Documents, (B) the Project Contracts, (C) the Project Financing Documents and (D) any other Contractual Obligation that relates to the ownership, development, construction, procurement, operation, management or maintenance of the Project and the Other Facilities.

### (s) United States Government Requirements.

- (i) <u>Davis-Bacon Requirements</u>. If the Davis-Bacon Act has been determined by DOE or the Department of Labor, as the case may be, to be applicable to the Project, each Clean Line Party is in compliance with all applicable Davis-Bacon Requirements. To the extent the Davis-Bacon Act applies to the Project, each applicable contract for construction, as defined in Department of Labor regulations at 29 C.F.R. § 5.2(j), includes the Davis-Bacon Requirement provisions set forth in Schedule 15 hereto (as such Schedule is supplemented from time to time in accordance with Section 8.24(b)).
- (ii) <u>Prohibited Persons.</u> (A) None of the Clean Line Parties, any Controlling Person of any Clean Line Party or any Principal Person of any Clean Line Party or any Principal Person of any Controlling Person of a Clean Line Party is a Prohibited Person, (B) to each Clean Line Party's Knowledge no event has occurred and no condition exists that is likely to result in any Clean Line

Party, any Controlling Person of any Clean Line Party or any Principal Person of a Clean Line Party or any Principal Person of any Controlling Person of a Clean Line Party becoming a Prohibited Person and (C) to each Clean Line Party's Knowledge, no Project Participant is a Prohibited Person.

(iii) <u>Anti-Corrupt Practices Laws, Etc.</u> (A) Each Clean Line Party, each Controlling Person of a Clean Line Party and each Principal Person, employee and agent of each Clean Line Party and each Controlling Person of a Clean Line Party have complied with all AM Laws, Anti-Corruption Laws and Sanctions and (B) each Clean Line Party has implemented and maintains in effect policies and procedures reasonably designed to ensure compliance by such Clean Line Party and its respective Principal Persons with AM Law, Anti-Corruption Laws and Sanctions.

### (t) <u>Energy Regulatory Status.</u>

### (i) Federal Energy Regulatory Status.

- (A) The appropriate Clean Line Party or Clean Line Parties are authorized, pursuant to Section 205 of the FPA, to charge negotiated rates for transmission rights on the Project and such authorization is in full force and effect.
- (B) As of the Effective Date and until the earlier of (i) the date on which any of the Project Facilities are energized, (ii) the date of FERC's order accepting the rate schedule or OATT filed by the appropriate Clean Line Party or Clean Line Parties or (iii) the effective date of the rate schedule or OATT filed by the appropriate Clean Line Party or Clean Line Parties, none of CLEP (solely as a result of the Project) or any of the Clean Line Parties is or will be a "public utility" under the FPA.
- (C) As of the Effective Date and until the date on which any of the Project Facilities are energized, none of CLEP (solely as a result of the Project) or any of the Clean Line Parties is or will be subject to regulation under PUHCA.
- (D) As of the earlier of (i) the date on which any of the Project Facilities are energized, (ii) the date of FERC's order accepting the rate schedule or OATT filed by the appropriate Clean Line Party or Clean Line Parties, or (iii) the effective date of the rate schedule or OATT filed by the appropriate Clean Line Party or Clean Line Parties as permitted by FERC, each of the appropriate Clean Line Party or Clean Line Parties will be a "public utility" under the FPA.
- (E) As of the date on which any of the Project Facilities are energized, each of the Clean Line Parties will be subject to regulation under PUHCA to the extent applicable.

### (ii) <u>State Energy Regulatory Status.</u>

- (A) By Order No. 9 issued by the APSC on January 11, 2011, in Docket No. 10-041-AU (the "APSC 2011 Order"), the APSC denied PECL's application for authority to operate as a public utility in the State of Arkansas. The APSC 2011 Order is final and is no longer subject to rehearing before the APSC.
- (B) By Order No. 590530 issued by the OCC on October 28, 2011 (the "OCC 2011 Order"), the OCC granted PECL OK's request for authority to operate as a transmission-only public utility in Oklahoma. The OCC 2011 Order is final, in full force and effect, and is no longer subject to rehearing before the OCC. PECL is in compliance with the OCC 2011 Order in all material respects.
- (C) By order issued by the Tennessee Regulatory Authority on May 5, 2015, in Docket No. 14-00036 (the "TRA 2015 Order"), PECL has been granted a Certificate of Public Convenience and Necessity to construct the transmission facilities in Tennessee that will interconnect with the Project. The TRA 2015 Order of the TRA is in full force and effect and is no longer subject to rehearing before the TRA. PECL is in compliance with the TRA 2015 Order in all material respects.
- (u) <u>Investment Company Act.</u> No Clean Line Party is required to register as an "investment company" as defined in, or subject to regulation under, the Investment Company Act.

### (v) Required Approvals.

- (i) Except as set forth in Schedule 16 (as such Schedule may be updated in accordance with Section 12.3), each Required Approval that is necessary for the Project in light of the current stage of construction, management and/or operation of the Project as of the applicable Representation Date, except for any *de minimis* Required Approval that is of a routine nature and obtainable in the ordinary course of business, (A) has been obtained, filed or made with the corresponding Governmental Authority, (B) is validly issued and in full force and effect and (C) there are no proceedings pending, or to any Clean Line Entity's Knowledge, threatened, seeking to rescind, terminate, adversely and materially modify, suspend, revoke or invalidate such Required Approval, except where such event or circumstance could not reasonably be expected to result, individually or in the aggregate, in a Clean Line Material Adverse Effect or an Adverse DOE Impact.
- (ii) None of the Clean Line Entities has any credible reason to believe that any Required Approval, that is not necessary for the Project in light of the current stage of construction, management and/or operation as of the applicable Representation Date but which will be required in the future, will not be obtained

on terms and conditions that are not materially inconsistent with the Clean Line Entities' performance under the Clean Line Documents on or prior to the date required or necessary for the continued construction, management and/or operation of the Project in accordance therewith.

- (iii) Each of the Clean Line Entities is in compliance in all material respects with each Required Approval that has been issued to it as of the applicable Representation Date.
- 12.2 <u>Survival</u>. The representations and warranties contained herein shall survive the execution and delivery of this Agreement.
- 12.3 <u>Disclosure Schedule</u>. The Clean Line Parties may, on any Representation Date, update, supplement or amend Schedules 3, 5, 9, 10, 11 and 16 (collectively the "<u>Disclosure Schedules</u>"), to correct any matter that would otherwise constitute a breach of any representation or warranty contained herein. If the matter or event giving rise to such an updated, supplemented or amended Disclosure Schedule, when taken together with all other matters and events that have given rise to updated, supplemented or amended Disclosure Schedules, could reasonably be expected, individually or in the aggregate, to result in a Clean Line Material Adverse Effect or an Adverse DOE Impact, then the Clean Line Parties shall not have the right to make any such update, supplement or amendment without the consent of DOE. Certain information set forth in the Disclosure Schedule is included solely for informational purposes and is not an admission of liability with respect to the matters covered by the information.

### ARTICLE XIII MISCELLANEOUS TERMS AND PROVISIONS

### 13.1 Notices; Consents; Approvals.

(a) The names and addresses of the Clean Line Parties and DOE for the purpose of receiving notices, invoices, payments and other communications required or permitted under this Agreement and the other Transaction Documents are as set forth below, which addresses may be changed from time to time by written notice to the other Party as provided herein.

Clean Line Parties: Plains and Eastern Clean Line Holdings LLC

1001 McKinney, Suite 700 Houston, Texas 77002 Attention: Cary Kottler Telephone: 832-319-6320 Facsimile: 832-319-6311

Email: CKottler@cleanlineenergy.com

With copies to: Latham & Watkins LLP

555 Eleventh Street NW

**Suite 1000** 

Washington, DC 20004 Attention: Paul J. Hunt Telephone: 202-637-2241 Facsimile: 202-637-2201 Email: Paul.Hunt@lw.com

DOE: U.S. Department of Energy

Office of the General Counsel 1000 Independence Ave., SW

Washington, DC, 20585

Attention: Samuel Walsh – Deputy General Counsel for Energy

Policy

Telephone: 202-586-6732 Facsimile: 202-586-4116

Email: Samuel.walsh@hq.doe.gov

With a copy to: Southwestern Power Administration

One West Third Street Tulsa, OK 74103-3502

Attention: Scott Carpenter – Administrator of Southwestern Power

Administration

Telephone: 918-595-6601 Facsimile: 918-595-6755

Email: scott.carpenter@swpa.gov

- (b) All notices or other communications required or permitted under this Agreement shall be in writing, properly addressed as provided in paragraph (a) above, and given by (i) hand delivery, (ii) a national overnight courier service, (iii) confirmed facsimile transmission, followed by a hard copy, or (iv) certified or registered mail, return receipt requested, and postage pre-paid. Any such notice or other communication shall be deemed to have been duly given (A) as of the date delivered if by hand delivery, national overnight courier service, email or confirmed facsimile transmission (provided a hard copy promptly follows by other means provided herein within five (5) days of the facsimile transmission), or (B) five (5) days after mailing if by certified or registered mail.
- (c) Time is of the essence under this Agreement. Wherever in this Agreement provision is made for the giving of consent or approval by either Party, unless otherwise specified, such consent or approval shall be (i) provided as soon as reasonably practicable following the request for such consent or approval and (ii) be in writing as provided above.
- 13.2 <u>Further Assurances</u>. Each Party shall, at the request of the other Party, execute and deliver or cause to be executed and delivered such documents and instruments as reasonably requested; <u>provided</u> that such documents and instruments are reasonably acceptable to the Party to whom the request is directed and are not otherwise specified herein, and take or cause to be taken all such other reasonable actions, as may be necessary to more fully and effectively carry out the intent and purposes of this Agreement.

- 13.3 Amendment; Waiver. No amendment or other modification of any provision of this Agreement shall be valid or binding unless it is in writing and signed by each of the Parties. No waiver of any provision of this Agreement shall be valid or binding unless it is in writing and signed by the applicable Party waiving compliance with such provision. No delay or omission in exercising any right, power, privilege or remedy under this Agreement or any other Transaction Document, including any rights and remedies in connection with the occurrence of an Event of Default or any right of termination shall impair any such right, power, privilege or remedy of DOE nor shall it be construed to be a waiver of any right, power, privilege or remedy or of any breach or default, or an acquiescence therein, or in any similar breach or default thereafter occurring, nor shall any waiver of any single right, power, privilege or remedy, or of any breach or default be deemed a waiver of any other right, power, privilege or remedy or of any other breach or default therefore or thereafter occurring.
- 13.4 <u>Lender-and Financing-Related Provisions</u>. Subject to DOE's rights and obligations under Applicable Law and the terms and conditions of this Agreement, at the request of Holdings, DOE shall agree to execute and deliver to the Project Financing Parties the DOE Direct Agreement, the Intercreditor Agreement and such other ancillary documents customary and reasonable for financing projects of a type similar to the Project reasonably requested by the Project Financing Parties and reasonably acceptable to DOE, all at the cost and expense of the Clean Line Entities; <u>provided</u> that neither DOE nor any of its counsel shall be obligated to provide any legal opinion related to the Project to any Project Financing party or any other Person other than the Section 1222 Decision.

### 13.5 Project Financing Document Provisions.

- (a) The Clean Line Entities shall be solely responsible for obtaining (and repaying) any necessary financing for the development, design, engineering, construction, ownership, operation, maintenance and management or any Capital Repair relating to the Project at its own cost and risk and without recourse to DOE, the Project or any other Covered Party. None of the Covered Parties shall have any obligation to pay any debt service or repay any Indebtedness issued or incurred by any Clean Line Party or any of its Affiliates or any other Person in connection with the Project or any of the transactions contemplated by the Transaction Documents.
- (b) The Project Financing Documents shall include the terms and conditions set forth in Schedule 2.
- (c) For the avoidance of doubt, subject to Section 6.4(a)(ii), nothing in this Agreement shall require that the Clean Line Entities fund Construction Costs or Project Costs with Project Financing or to enter into any Project Financing Documents, so long as the Project Equity Commitments are sufficient to fund all Construction Costs and to enable the Clean Line Entities to otherwise satisfy its obligations under the Transaction Documents.
- 13.6 <u>Grant of Security Interest</u>. The Clean Line Entities may grant security interests in, or assign the entirety of the Clean Line Entities' interests in and under the Transaction Documents to the Project Financing Parties for purposes of securing the Project Financing,

subject to any terms and conditions contained in the Transaction Documents, the Intercreditor Agreement and the DOE Direct Agreement. The Clean Line Entities shall be strictly prohibited from pledging or encumbering its interest under the Transaction Documents to secure any Indebtedness or any other obligations other than the Project Financing (except for Permitted Liens).

- 13.7 <u>DOE Review Standard</u>. Except as otherwise set forth in this Agreement, to the extent that any document, agreement, report, certificate, opinion or other evidence of any matter or condition is required to be delivered to DOE pursuant to the terms of this Agreement or any other Transaction Document, such document, agreement, report, certificate, opinion or other evidence shall be required to be in form and substance that is satisfactory to DOE. In addition, to the extent that: (a) any document or agreement is required to be executed by DOE, (b) any determination is contemplated to be made by DOE or (c) any condition is required to be satisfied or waived pursuant to the terms of this Agreement or any other Transaction Document, DOE shall make such determination or confirm satisfaction or waiver of any such condition acting in its sole and absolute discretion.
- 13.8 <u>DOE Delegation</u>. DOE shall be entitled to execute or perform any of its rights, remedies, power, privileges, duties or obligations under this Agreement, any other Transaction Document or, to the extent applicable, any Required Approval through any of its nominees (including any other federal agency) or agents.
- 13.9 <u>Assignments</u>. Except as otherwise expressly permitted pursuant to Section 13.6, no Party may assign its rights and obligations under this Agreement or any other Transaction Document without the prior written consent of the other Parties.
- 13.10 <u>Successors and Assigns</u>. Each and all of the covenants, terms, provisions and agreements contained in this Agreement shall be binding upon and inure to the benefit of the Parties hereto and, to the extent permitted by this Agreement, their respective successors and permitted assigns.
- 13.11 <u>Joint and Several Obligations</u>. Notwithstanding anything to the contrary in this Agreement, each Clean Line Party shall be jointly and severally liable for all obligations of each other Clean Line Party under this Agreement and each Clean Line Entity shall be jointly and severally liable for all obligations of each other Clean Line Entity under this Agreement.
- 13.12 <u>Right to Intervene</u>. Nothing in this Agreement shall prohibit any Party from intervening in any regulatory proceeding relating to the Project or the Other Facilities and taking any position in any such proceeding that it deems appropriate.
- 13.13 <u>Publication; Public Statements</u>. None of the Clean Line Parties or any of its representatives may issue any press release or make any other public statement directly or indirectly relating to DOE, this Agreement, the other Transaction Documents and DOE's involvement in the transaction contemplated thereby without DOE's prior written consent (other than information that is generally available to the public and background or summary information of a general nature concerning the Project).

13.14 <u>Third Parties</u>. Except as expressly provided otherwise in this Agreement (including the provisions for the protection of all Covered Parties), none of the promises, rights or obligations contained in this Agreement shall inure to the benefit of any Person that is not a Party to this Agreement, and no action may be commenced or prosecuted against any Party by any third Person claiming to be a third-person beneficiary of this Agreement or the transactions contemplated thereby.

### 13.15 <u>Independent Contractor Status</u>.

- (a) The Clean Line Parties' interests under this Agreement and any other Transaction Document shall be solely those of an independent contractor, and the Clean Line Parties and DOE are not in a relationship of co-venturers, partners, lessor-lessee or principal-agent (except to the extent that the Transaction Documents expressly appoint any Clean Line Entity as DOE's agent for specified purposes (including for purposes of the Construction Contracts)).
- (b) Nothing contained in this Agreement or in any other Transaction Document shall be deemed or construed to create a partnership, tenancy in common, joint tenancy, joint venture or co-ownership by, between or among DOE or any other Covered Party and the Clean Line Parties, or any other Person.
- 13.16 TN and TX Facilities. Except to the extent expressly set forth in this Agreement, none of the TN Facilities or the TX Facilities are covered by, or shall be subject to, this Agreement or any other Transaction Document.
- 13.17 <u>Governing Law</u>. This Agreement shall be governed by, and construed and interpreted in accordance with, the Federal law of the United States of America. To the extent that Federal law does not specify the appropriate rule of decision for a particular matter at issue, it is the intention and agreement of the parties thereto that the laws of the State of New York shall be adopted as the governing Federal rule of decision.

### 13.18 <u>Jurisdiction</u>. Each Clean Line Party irrevocably and unconditionally:

- (a) submits itself and its Properties, in any legal action or proceeding against it arising out of or in connection with this Agreement or any other Transaction Document or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general jurisdiction of (i) the courts of the United States of America for the District of Columbia; (ii) the courts of the United States of America in and for the Southern District of New York; (iii) any other federal court of competent jurisdiction in any other jurisdiction where it or any of its Property may be found; and (iv) appellate courts from any of the foregoing;
- (b) consents that any such action or proceeding may be brought in or removed to such courts, and waives any objection, or right to stay or dismiss any action or proceeding, that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;

- (c) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to any Clean Line Party at its address set forth in Section 13.1 or at such other address that it shall notify DOE hereunder;
- (d) agrees that nothing herein shall (i) affect the right of any Covered Party to effect service of process in any other manner permitted by law; or (ii) limit the right of any Covered Party to commence proceedings against or otherwise sue the Clean Line Parties or any other Person in any other court of competent jurisdiction, nor shall the commencement of proceedings in any one or more jurisdictions preclude the commencement of proceedings in any other jurisdiction (whether concurrently or not) if, and to the extent, permitted by the Applicable Laws; and
- (e) subject to rights to appeal in accordance with Applicable Laws, agrees that judgment against it in any such action or proceeding shall be conclusive and may be enforced in any other jurisdiction within or without the U.S. by suit on the judgment or otherwise as provided by law, a certified or exemplified copy of which judgment shall be conclusive evidence of the fact and amount of the Clean Line Parties' obligation.

### 13.19 <u>Dispute Resolution</u>.

- (a) If both Holdings and DOE agree, the Parties may first attempt in good faith to resolve any dispute under this Agreement and any other Transaction Document through informal negotiations by their respective representatives on the Coordination Committee, which can be escalated to the senior officers of each party if necessary or desirable.
- (b) For disputes that are construction-related, operational-related or in respect of technical, financial or accounting issues, the Parties shall have the right to appoint an independent technical or financial expert to assist in resolving any such dispute. The Clean Line Entities shall bear the cost of such independent technical or financial expert.
- 13.20 <u>Waiver of Jury Trial</u>. EACH PARTY HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION, OR IN ANY PROCEEDING RELATED THERETO, ARISING UNDER THIS AGREEMENT OR THE TRANSACTIONS RELATED HERETO, WHETHER NOW EXISTING OR HEREAFTER ARISING, WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE. EACH PARTY AGREES THAT IT WILL NOT SEEK A TRIAL BY JURY IN RESPECT OF ANY SUCH CLAIM, DEMAND, ACTION, CAUSE OF ACTION OR PROCEEDING.
- 13.21 <u>Negotiation and Documentation of this Agreement</u>. Each of the Parties acknowledges and agrees that it has had the opportunity to have its legal counsel review this Agreement and participate in the negotiation and documentation hereof, and the Parties are fully familiar with each of the provisions of this Agreement and the effect thereof.
- 13.22 <u>Severability</u>. In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality

and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby.

- 13.23 <u>Counterparts</u>. This Agreement may be executed by the Parties in two or more separate counterparts (including by PDF or facsimile transmission), each of which shall be deemed an original, and all of said counterparts taken together shall be deemed to constitute one and the same instrument.
- 13.24 Entire Agreement. This Agreement contains the entire agreement and understanding of the Parties with respect to the subject matter hereof and supersedes all prior agreements and understandings, whether written or oral, of the Parties relating to the subject matter hereof. Any oral or written representation, warranty, course of dealing or trade usage not contained or referenced herein shall not be binding on either Party.
- 13.25 <u>Time is of the Essence</u>. Each of the Parties acknowledges that timely achievement of commercial operation of the Project is essential, and therefore time is of the essence in performing all obligations set forth herein.
- 13.26 Confidentiality of Information. Each of the Parties agrees that it shall treat all information exchanged or provided by and among the Parties in connection with the Project as confidential and shall not disclose any such information to any other Person except (a) to the extent such information is required to be disclosed pursuant to Applicable Laws (including the Freedom of Information Act, 5 U.S.C. §552 and DOE's implementing regulations set forth in 10 C.F.R. Part 1004), (b) to the extent such information is required to be disclosed by any Governmental Authority, (c) to any officer, director, employee, advisor, agent or representative of such Party, solely in the context of such Party's evaluation, consideration and participation of the Project, (d) to other Persons that agree to be bound by obligations of confidentiality at least as restrictive as this Section 13.26 or (e) as otherwise agreed in writing by the Parties from time to time; provided, however, that nothing herein shall prohibit disclosure of any information that (i) is or becomes generally available to the public other than as a result of disclosure by a Party in violation of this Section 13.26 or (ii) was known to the disclosing Party through means independent of receipt of such information by another Party; provided further that the Parties hereby agree that the terms and conditions of this Agreement shall not be treated as "information" subject to the provisions of this Section 13.26 and that the Parties shall be entitled to freely disclose the terms and conditions of this Agreement.

### 13.27 Non-Exclusivity.

(a) Notwithstanding anything to the contrary in this Agreement, the Clean Line Parties agree and acknowledge that each of DOE and each other Covered Party will have the unfettered right in its sole discretion, at any time and without liability, regardless of impacts on the Project or the Other Facilities, to finance, develop, approve, expand, improve, modify, upgrade, add capacity to, reconstruct, rehabilitate, restore, renew and replace any existing and new transmission lines or other facilities. Such right extends to facilities whether adjacent to, nearby or otherwise located as to affect the Project, its operation and maintenance and/or its revenues.

- (b) The foregoing facilities include those owned or operated by (i) DOE, including those owned or operated by a private entity pursuant to a contract with DOE, (ii) a joint powers authority or similar entity to which DOE is a member, (iii) a Governmental Authority pursuant to a contract with DOE and (iv) a Governmental Authority with respect to which DOE has contributed funds, in-kind contributions or other financial or administrative support.
- (c) DOE will have the right, without liability, to make discretionary and non-discretionary distributions of federal and other funds for any transmission projects, programs and planning, and to exercise all its authority to advise and recommend on transmission and energy planning, development and funding.

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their officers thereunto duly authorized as of the day and year first above written.

U.S. DEPARTMENT OF ENERGY

By:

Name: Ernest J. Moniz

Title: Secretary of Energy

[Signature Page to Participation Agreement]

## PLAINS AND EASTERN CLEAN LINE HOLDINGS LLC By: Name: Michael Skelly Title: President ARKANSAS CLEAN LINE LLC Name: Michael Skelly Title: President PLAINS AND EASTERN CLEAN LINE OKLAHOMA LLC By: Name: Michael Skelly Title: President OKLAHOMA LAND ACQUISITION COMPANY LLC Name: Michael Skelly Title: President PLAINS AND EASTERN CLEAN LINE LLC By: Name: Michael Skelly Title: President

[Signature Page to Participation Agreement]

# Schedules to Participation Agreement

This document constitutes all of the schedules (the "Schedules") referenced in the Participation Agreement, dated as of March 25, 2016 (as amended, modified or restated from time to time, this "Participation Agreement"), by and among the United States Department of Energy, Plains and Eastern Clean Line Holdings LLC, Arkansas Clean Line LLC, Plains and Eastern Clean Line Oklahoma LLC, Oklahoma Land Acquisition Company LLC and Plains and Eastern Clean Line LLC. Unless otherwise defined in the Schedules, all capitalized names and terms set forth herein and not otherwise defined herein will have the same meanings as set forth in the Participation Agreement.

The captions appearing herein are for the convenience of the Parties only and will not be construed to affect the meaning of the provisions of the Participation Agreement.

With respect to the Disclosure Schedules only:

- (a) The inclusion of any contract, lease, document, claim, action or any other item (individually, an "<u>Item</u>") on any Disclosure Schedule will not constitute a representation by any Clean Line Party that such Item is material, or that the non-inclusion of such Item on such Disclosure Schedule (or any other Schedule) would result in a misrepresentation or breach of warranty on the part of the Clean Line Parties;
- (b) Any Item set forth in the Disclosure Schedules that is not required to be disclosed pursuant to the Participation Agreement has been disclosed solely for informational purposes and such disclosure will not be construed to broaden the scope of any representation or warranty;
- (c) No disclosure in any Disclosure Schedule relating to any possible breach or violation of any agreement, law, regulation or other legal requirement will be construed as an admission or indication that any such breach or violation exists or has actually occurred; and
- (d) Inclusion of an Item under one Disclosure Schedule shall be deemed to be an inclusion of such Item on one or more other Disclosure Schedules where it is reasonably apparent from the text of such disclosure.

### **Clean Line Entities Real Estate Rights Acquisition Procedures**

In connection with the Clean Line Entities' acquisition of any Project Real Estate Rights, each Clean Line Entity shall comply in all material respects with the Routing and ROW Plan and all other procedures set forth in this Agreement (including this Schedule 1) prior to any such Project Real Estate Right being deemed to be a DOE Delegated Real Estate Right.

In connection with any acquisition of a Project Real Estate Right, each Clean Line Entity shall comply with the Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, as amended (42 U.S.C. § 4601 *et seq.*) and the regulations promulgated thereunder and set forth in 49 C.F.R. Part 24 (collectively, the "<u>Uniform Act</u>"), DOE Policies and all other Applicable Laws. The Routing and ROW Plan shall be agreed between the Parties as a condition to the occurrence of the Commencement Date. Any amendments or modifications to the Routing and ROW Plan or the Project Plans shall be subject to the terms of this Agreement. The Routing and ROW Plan shall include, among other requirements and procedures, the requirements and procedures set forth herein and in the Clean Line Uniform Act Execution Plan set forth as Schedule 17. To the extent of any conflict or inconsistency between this Schedule 1 and Schedule 17, the provisions set forth in this Schedule 1 shall control. To the extent of any conflict or inconsistency between this Schedule 1 and the Uniform Act, the requirements of the Uniform Act shall control.

1. <u>Defined Terms</u>. For purposes of this Schedule 1, capitalized terms shall have the meanings defined for such terms in Article I of the Agreement. In addition, the following capitalized terms shall have the following meanings:

"Curative Party" means any Person from whom any consent or other necessary action is required in order for the applicable Landowner to be able to grant or convey a clean or marketable Project Real Estate Right in favor of the Clean Line Entities without any clouds or defects on title.

"Documentation Package" means, with respect to any Project Real Estate Right, a comprehensive written file maintained by the Clean Line Entities documenting all activities undertaken by the Clean Line Entities with respect to the acquisition of such Project Real Estate Right, including: (a) the relevant Title Search, (b) contact information in respect of any applicable Landowner, Curative Party or tenant, (c) any related market data studies, (d) all appraisals undertaken in respect of such Project Real Estate Right, (e) surveys in respect of such Project Real Estate Right, (g) minutes of all meetings, discussions and telephone calls held by any Clean Line Entity with the applicable Landowner, any applicable Curative Party or any applicable tenant and (h) all other communication and correspondence received or sent by any Clean Line Entity in connection with its acquisition of such Project Real Estate Right.

"<u>Landowner</u>" means, with respect to any Project Real Estate Right, the Person (or Persons) that own the fee simple or leasehold or other applicable Real Estate Right to which such Project Real Estate Right relates.

"<u>Title Defect</u>" means any Lien, easement, mortgage, encumbrance or other restriction (whether contractual or otherwise) that affects or limits the ability of any Landowner to grant or convey a clean and marketable Project Real Estate Right in favor of the Clean Line Entities.

"Waiver Parcel" means any Project Real Estate Right that is determined to be a "Waiver Parcel" in accordance with the provisions set forth in Schedule 17.

2. <u>Title Searches, Identification of Title Issues and Location of Landowner</u>. For each Project Real Estate Right, the Clean Line Entities shall obtain a thirty (30) year title search or limited certificate of title from a reputable national title company or reputable land acquisition company (a "<u>Title Search</u>") in order to enable it to identify the current Landowner in respect of the underlying Real Estate Right and any applicable Title Defects. The Clean Line Entities shall use all commercially reasonable efforts to locate each applicable Landowner through any available search methods, including through a tax record search, review of publicly available information, using a private investigator to conduct a search for such Landowner, inquiries with relatives, neighbors or other individuals that could reasonably be likely to know the location of such Landowner and/or publication of a notice or advertisement in a newspaper. The Clean Line Entities shall attempt to contact any applicable Landowner by at least three (3) different forms of contact including by phone, in person, first class mail, certified mail or leaving messages with a neighbor or family member of the applicable Landowner.

### 3. <u>Negotiations; Appraisal and Offer.</u>

- (a) <u>Initial Notice and Landowner Materials</u>. The Clean Line Entities shall notify any applicable Landowner of their interest in acquiring a Project Real Estate Right and offer to meet with such Landowner in person. In connection with its efforts to acquire any Project Real Estate Right, the Clean Line Entities shall provide each applicable Landowner with:
  - (i) a proposed form of easement and/or other applicable documentation relating to the conveyance of the proposed Project Real Estate Right;
  - (ii) a payment calculation sheet or other documentation in respect of any compensation proposed to be paid to such Landowner in connection with the applicable Project Real Estate Right; <u>provided</u>, <u>however</u>, that with respect to any parcel that is not a Waiver Parcel, such payment calculation sheet or other documentation shall only be provided after the appraisal has been performed;
  - (iii) a sketch identifying the boundaries and the nature of the applicable Project Real Estate Right;
  - (iv) a construction questionnaire designed to gather necessary information in respect of conditions at the location of the applicable Project Real Estate Right;

- (v) a copy of the Clean Line Entities' Codes of Conduct for acquisitions of Project Real Estate Rights (which is attached as Schedule 12 to this Agreement);
- (vi) a request for permission to conduct a survey of the applicable Project Real Estate Right; and
- (vii) in respect of any Project Real Estate Right located in Oklahoma, a copy of the Private Rights Settlement Agreement, dated January 14, 2011 (the "Private Rights Settlement Agreement"), and the Order from the OCC, dated October 28, 2011, approving the PECL OK's application to conduct business as a public utility in Oklahoma.

The initial notice and materials sent to any applicable Landowner shall otherwise comply with the requirements set forth in the Uniform Act and Schedule 17.

- (b) The Clean Line Entities shall follow the provisions and procedures set forth in Schedule 17 for purposes of determining whether any applicable Project Real Estate Right shall be treated as a Waiver Parcel or whether an appraisal is required in respect of such Project Real Estate Right. In all instances involving an appraisal, the Clean Line Entities shall ensure that such appraisal is undertaken by a certified independent reputable appraiser and that the applicable Landowner shall be given the opportunity to actively participate in, and be present for, the appraisal process as well as communicate with the applicable appraiser. Each appraisal will be further reviewed and confirmed by a separate independent reputable appraiser. The appraisal process in respect of any Project Real Estate Right that has not been determined to be a Waiver Parcel shall otherwise be undertaken in accordance with the provisions of the Uniform Act and Schedule 17.
- The Clean Line Entities shall make an offer to acquire any applicable (c) Project Real Estate Right pursuant to a settlement offer that shall include an offer to pay just compensation to the applicable Landowner in accordance with the Uniform Act and Schedule 17 in connection with its conveyance of such Project Real Estate Right. Compensation payable in connection with the acquisition of any applicable Project Real Estate Right shall include compensation in respect of: (i) the area comprising the applicable Project Real Estate Right (which amount shall not be less than the market value or appraised value, as applicable, of the Project Real Estate Right being conveyed, taking into consideration impacts (if any) on any adjacent or surrounding Real Estate Rights of the applicable Landowner that are not specifically contemplated to be included in the Project Real Estate Rights), (ii) each structure or improvement located on the applicable Project Real Estate Right, (iii) any damage to any crops, timber, livestock, structures or improvements of the Landowner that are reasonably likely to arise as a result of the conveyance of the applicable Project Real Estate Right and the Project and (iv) if applicable, removal and relocation costs as a result of the Project. An offer by a Clean Line Entity to acquire a Project Real Estate Right shall also include the terms set forth on Appendix A to this Schedule 1.
- (d) The Clean Line Entities shall make no less than three (3) attempts to meet with any affected Landowner personally to discuss its offer to acquire any applicable Project

Real Estate Rights and, to the extent that such Landowner cannot be contacted personally, the Clean Line Entities shall ensure that a copy of the notice of offer and all other required documentation relating thereto is delivered via certified mail or registered-first class mail-return receipt requested to the applicable Landowner. The Clean Line Entities shall use all other commercially reasonable efforts to acquire any Project Real Estate Rights and shall give any applicable Landowner or related tenant a reasonable opportunity to review and discuss any proposed offer to acquire such Project Real Estate Rights. The Clean Line Entities shall give full and fair consideration to any comments, questions or suggestions of any Landowner in respect of the proposed conveyance of the applicable Project Real Estate Rights including review and consideration of any materials any Landowner may provide as relevant to the determination of the value of such Project Real Estate Rights. Each Landowner shall be given a reasonable opportunity (including a period of reasonable length) to consider any offer to acquire any Project Real Estate Rights.

4. <u>Multiple Landowners</u>. The Clean Line Entities shall negotiate with all applicable Landowners in respect of its acquisition of any Project Real Estate Rights in accordance with the terms hereof. To the extent that the Clean Line Entities are unable to locate all applicable Landowners or any applicable Landowners are not willing to agree (or are restricted from agreeing) to the conveyance of a Project Real Estate Right, the Clean Line Entities shall use all commercially reasonable efforts to enter into a voluntary agreement with any Landowner that has been located and is otherwise willing to agree to convey a Project Real Estate Right to the Clean Line Entities in accordance with the terms of this Agreement prior to any such Project Real Estate Right being designated a DOE Delegated Real Estate Right.

### 5. Title Defects; Tenants.

- (a) To the extent that any Title Search identifies any Title Defects, then if determined appropriate in the reasonable judgment of the Clean Line Entities, the Clean Line Entities shall retain reputable local real estate counsel to determine what measures are available with respect to removing or addressing such Title Defect. The Clean Line Entities shall use all commercially reasonable efforts to locate any Curative Party through any available search methods, including through a tax record search, review of publicly available information, using a private investigator to conduct a search for such Curative Party, inquiries with relatives, neighbors or other individuals that could reasonably be likely to know the location of such Curative Party and/or publication of a notice or advertisement in a newspaper. Upon locating any Curative Party, the Clean Line Entities shall use all commercially reasonable efforts to obtain any consent from or other necessary action by any Curative Party to enable the applicable Landowner to convey a Project Real Estate Right to the Clean Line Entities.
- (b) To the extent that any tenant is present on any Project Real Estate Right, the Clean Line Entities shall use all commercially reasonable efforts to meet with such tenant to discuss any concerns or issues relating to the tenant. The Clean Line Entities shall compensate any tenant for any potential loss or damage to such tenant that could reasonably be anticipated to result from the conveyance of the applicable Project Real Estate Right to the Clean Line Entities, including any reasonable costs or expenses to the tenant associated with relocation or damage to crops or improvements.

### 6. <u>Commercially Reasonable Efforts; Course of Conduct.</u>

- (a) For purposes of complying with the terms hereof, the Clean Line Entities' use of "commercially reasonable efforts" (other than with respect to locating Persons) shall include:
  - (i) the Clean Line Entities' prompt and courteous response to any applicable Landowner's, Curative Party's or tenant's (or their designated representative or counsel) inquiry, comments or questions;
  - (ii) in the case of any Landowner, the Clean Line Entities' offer to pay compensation to such Landowner for the conveyance of the applicable Project Real Estate Right as contemplated herein;
  - (iii) in the case of any Curative Party, the Clean Line Entities' offer of fair compensation to such Curative Party in respect of any reasonable legal costs or other out-of-pocket costs or expenses incurred by such Curative Party in connection with any action requested of such Curative Party;
  - (iv) the Clean Line Entities' meeting with (either in person or by phone) any applicable Landowner, Curative Party or tenant (or their designated representative or counsel) to the extent reasonably possible or requested and providing an overview of the Project and the Clean Line Entities' applicable policies and procedures in respect thereof;
  - (v) in Oklahoma, to the extent that the Clean Line Entities and any Landowner have reached agreement on the form of any acquisition of a Project Real Estate Right but have been unable to reach agreement as to the appropriate compensation payable in respect thereof, at the request of any applicable Landowner, the Clean Line Entities' submission to binding arbitration in respect of the amount of compensation payable to such Landowner in accordance with the terms of the Private Rights Settlement Agreement;
  - (vi) engaging with any representative (including counsel) of the applicable Landowner in respect of the acquisition of the applicable Project Real Estate Right; and
  - (vii) any other commercially reasonable efforts under Prudent Utility Practice that the Clean Line Entities determine in their reasonable judgment is capable of taking under Applicable Law that could reasonably be expected to result in the conveyance of the applicable Project Real Estate Right to the Clean Line Entities for commercially reasonable terms.
- (b) Each of the Clean Line Entities shall be entitled to perform its obligations hereunder through any agent or designee; <u>provided</u> that no Clean Line Entities shall be relieved from compliance with the terms hereof due to any action or inaction by any such agent or designee. Each Clean Line Entities shall make it or its agents or designees available for in-person meetings with any Landowner, Curative Party or tenant, or any of their respective

agents or representatives, as may be reasonably necessary to acquire any Project Real Estate Right.

- (c) No Clean Line Entity shall engage in any coercive action with respect to any Landowner, Curative Party or tenant in respect of the undertakings required hereby.
- (d) The Clean Line Entities shall develop a standard script of talking points (subject to DOE's approval) describing DOE's participation in the Project and DOE's obligations in connection with any acquisition of Project Real Estate Rights, which standard script shall be applied and followed by each Clean Line Entity and its contractors in material respects in all communications and correspondence with any Landowner, Curative Party or tenant.

## 7. <u>Designation of Any Project Real Estate Right as a DOE Delegated Real Estate Right.</u>

- (a) Prior to designating any Project Real Estate Right as a DOE Delegated Real Estate Right, the Clean Line Entities shall notify the applicable Landowner (to the extent located) and Curative Party that it is making a "final" offer for the acquisition of the applicable Project Real Estate Right and indicating the terms thereof. Such "final" offer shall indicate that the Clean Line Entities will thereafter turn negotiations in respect of acquiring such Project Real Estate Right over to DOE.
- (b) Any designation by the Clean Line Entities of a Project Real Estate Right as a DOE Delegated Real Estate Right shall be made by written notice to DOE and shall be accompanied by a comprehensive Documentation Package in respect of the underlying Real Estate Right.
- Following receipt by DOE of a written notice of a designation by Holdings (c) of a Project Real Estate Right as a DOE Delegated Real Estate Right, DOE shall review the applicable Documentation Package and shall be entitled to discuss any questions or concerns it may have with respect thereto with Holdings and its agents and designees involved in the acquisition process. Promptly following such review and discussion, DOE shall notify Holdings in writing that either (i) DOE agrees with Holdings' designation of such Project Real Estate Right as a DOE Delegated Real Estate Right, in which case the Clean Line Entities shall thereafter cease to be responsible for the acquisition of such Project Real Estate Right, or (ii) DOE does not agree with Holdings' designation of such Project Real Estate Right as a DOE Delegated Real Estate Right, in which case such notification shall set forth the additional actions, steps, requirements or information that DOE requires be taken or provided by the Clean Line Entities, as applicable, prior to such Project Real Estate Right being approved as a DOE Delegated Real Estate Right and the Clean Line Entities ceasing to be responsible for the acquisition of such Project Real Estate Right; provided, however, DOE shall not be entitled to require any further actions, steps, requirements or information other than what is provided for herein, under the Uniform Act and in Schedule 17.
- 8. <u>Documentation Package, Etc.</u> With respect to any Project Real Estate Right that the Clean Line Entities anticipate may be designated as a DOE Delegated Real Estate Right, the

Clean Line Entities shall maintain a comprehensive written Documentation Package in respect of the underlying Real Estate Right. To the extent not inconsistent with the foregoing, the Clean Line Entities shall comply with all data and information maintenance requirements contemplated by Schedule 17, and DOE shall have access to all such data and information as it may request from time to time.

9. <u>Relocations</u>. In the event that residences or Persons are required to be relocated as a result of the Project, the Clean Line Entities will draft relocation policies and procedures that follow the Uniform Act, which policies and procedures will be submitted to DOE for review and approval.

### **Plains & Eastern Clean Line Compensation**

### 1. Clean Line Compensation Package

### a. Easement Payment

Landowners will receive a \$/per acre payment for the total acreage comprising the easement area. The \$/per acre price shall be based on 100% of the fair market value of the fee title of the land traversed by the easement area, rather than a typical discounted fair market value for an easement. In order to determine the fair market fee title value of land, Clean Line has engaged a real estate appraisal firm to provide county wide market data studies, which studies have produced an average fair market value for fee title to land for specific land uses in each county (the "Average Fair Market Per Acre Value), all as more particularly described in the Clean Line Uniform Act Execution Plan attached as Schedule 17 to this Agreement.

For purposes of the \$/per acre payment, Clean Line will pay Landowners <u>the greater</u> of the following: (i) the Average Fair Market Per Acre Value, or (ii) if an appraisal is required under the Uniform Act, the appraised value of the easement determined by such appraisal.

### **b.** Structure Payment

Landowners will receive a payment for each structure that is located within the easement area. The structure payment is calculated based on the type and number of structures. The Landowner has the right to elect to receive a one-time payment or annual payments. If selected, the annual payment will include a 2% annual escalator. Payments for structure types are as follows:

Type of Structure	One-Time Payment	<b>Annual Payment</b>
Monopole or Lattice Mast Structure	\$ 6,000	\$ 500
Lattice Structure	\$ 18,000	\$ 1,500
Guyed Lattice Structure	\$ 24,000	\$ 2,000

<sup>&</sup>lt;sup>1</sup> In calculating the appraised value of the easement, the appraiser will consider the per acre value of the easement strip of land (typically a discounted value from the fee sales price) <u>plus</u> the amount of any damages to the remainder of the landowner's property resulting from the presence of the easement.

### c. Damages Payment

Clean Line will pay Landowners for any damages to crops, timber, livestock, structures or improvements resulting from the construction, maintenance or operation of the Project, regardless of when they occur and without any cap on the amount of such damages. For example, if the Landowner experiences a loss in crop yields that is attributed to the operation of the Project (*i.e.*, an inability to spray certain rows of crops due to the presence of the transmission line) then Clean Line will pay the value of such loss in yield for so long as such losses occur. In other words, the intent is that the Landowner be made whole for any damages or losses that occur as a result of the Project at any time.

### 2. Minimum Payments

Some of the Project Real Estate Rights to be acquired may be very small in size. Therefore, in order to incentivize Landowners that might otherwise receive a very small payment, Landowners will receive a minimum payment of \$2,000 per parcel, regardless of the size of the easement area on their land. In addition, in the event that no structures are constructed on a Landowner's parcel, such Landowner will also receive a minimum structure payment. For such minimum structure payment, the Landowner will have the right to elect either (i) a one-time payment equal to \$1,500 or (ii) an annual payment equal to \$125, with a 2% annual escalator.

### 3. Arbitration

If Clean Line and a Landowner have reached agreement on the form of any easement or other document evidencing the Project Real Estate Right, but are unable to reach agreement on the appropriate compensation, at such Landowner's request, Clean Line will submit the issue of Landowner compensation to binding arbitration. Arbitration shall be administered by the American Arbitration Association (the "AAA") in accordance with its Commercial Arbitration Rules. Arbitration shall take place in, and shall be conducted in accordance with the laws of, the state in which the Project Real Estate Right is located. Arbitrators shall be appointed as provided in the AAA Commercial Arbitration Rules, but in all events shall be selected from a pool of qualified arbitrators within the state in which the Project Real Estate Right is located.

### **Provisions Required to be Incorporated into Project Financing Documents**

The Project Financing Documents and any amendments or supplements thereto, shall comply with the following terms and conditions:

- The proceeds of the Project Financing are obligated to be used exclusively for the purposes of (a) acquiring, designing, permitting, developing, constructing, equipping, improving, modifying, operating, owning, maintaining, reconstructing, restoring, rehabilitating, renewing or replacing the Project, or any Capital Repair relating to the Project, (b) paying principal and interest on the Project Financing, (c) paying premiums or costs for insurance, bonds and other performance security or paying reasonable development fees to the Clean Line Entities or to any Project Participant or its Affiliates for services related to the Project, (d) paying fees and premiums to any Project Financing Party or such Project Financing Party's agents in consideration for the Project Financing or the commitment thereof, (e) paying costs and fees in connection with the closing of the Project Financing, including fees and costs of counsel and consultants, (f) funding of the Advance Funding Account, Capital Repairs Reserve Account and Wind-Up Reserve Account and making payments and other amounts due to DOE and the Covered Parties under the Transaction Documents, (g) funding reserves required under the Project Financing Documents or Applicable Law, including securities laws and Environmental Laws, (h) payment of interest on subordinated debt and other financing costs (such as fees on letters of credit to the extent used to secure deferred equity contributions), (i) such other uses as are customary and permitted under the terms of the Project Financing Documents and (i) refinancing the Project Financing under clauses (a) through (i) above.
- 2. The Project Financing Documents (including any accounts or depositary agreement) shall provide that all amounts due and payable to DOE by any Clean Line Entity under this Agreement, any Real Estate Rights Agreement and any other Transaction Document (other than the Participation Amount required to be paid pursuant to Section 11.2) shall be paid as operating expenses of the Project at the top of any revenue application waterfall provided for in such Project Financing Documents and at the same priority level as other general operating costs paid in connection with the Project.
- 3. No Project Financing Document or other instrument purporting to mortgage, pledge, encumber, or create a Lien, charge or security interest on or against any of the Project Facilities owned by the Clean Line Entities or any of the Clean Line Entities' rights and interests in the Project (including the Clean Line Entities' rights and interest in the Transaction Documents) shall extend to or affect DOE's interest in the DOE Acquired Real Property and the AR Facilities, the Account Collateral (other than the Capital Repair Reserve Account) or any of DOE's other rights, privileges and interests under the Transaction Documents.
- 4. The Project Financing Documents shall include a conspicuous recital or provision to the effect that payment of the principal thereof and interest thereon is a valid claim only as against the Clean Line Entities and the security pledged by the applicable Clean Line Entities in respect thereof and is not an obligation, moral or otherwise, of DOE or any Covered Party, and

neither the full faith and credit nor the taxing power of DOE or Covered Party is pledged to the payment of the principal thereof and interest thereon.

- 5. Each Project Financing Document containing express provisions regarding default by a Clean Line Entity shall require that if such Clean Line Entity is in default thereunder and the Project Financing Parties (or an agent thereof) give notice of such default to any Clean Line Entity, then the Project Financing Parties (or an agent thereof) shall also give concurrent notice of such default to DOE. Each Project Financing Document that provides remedies to one or more Project Financing Parties for default by a Clean Line Entity or other applicable Person shall require that such Project Financing Parties deliver to DOE, concurrently with delivery to a Clean Line Entity or any other Person, every notice of election to sell, notice of sale or other notice required by Applicable Law or by the Project Financing Documents in connection with the exercise of remedies under the Project Financing Document.
- 6. The Project Financing Documents shall (a) expressly state that no Project Financing Party shall (i) name or join DOE or any Covered Party in any legal proceeding seeking collection of the Indebtedness incurred under the Project Financing or other obligations secured thereby or the foreclosure or other enforcement of such Project Financing Document (other than in connection with any express obligations of DOE under the DOE Direct Agreement and the Intercreditor Agreement) and (ii) seek any damages or other amounts from DOE or any Covered Party, whether for Indebtedness incurred under any Project Financing Document (other than in connection with any express obligations of DOE under the DOE Direct Agreement and the Intercreditor Agreement) or any other amount and (b) contain a Release Provision as provided in Section 11.10.
- 7. The DOE Direct Agreement shall expressly state that each Project Financing Party agrees to non-exclusive general jurisdiction and venue in any action by or against DOE or its successors and assigns of (a) the courts of the United States of America for the District of Columbia, (b) the courts of the United States of America in and for the Southern District of New York, (c) any other federal court of competent jurisdiction in any other jurisdiction where any Clean Line Entity or any of its Property may be found and (d) appellate courts from any of the foregoing.

## Schedule 3 to Participation Agreement

### **Existing Indebtedness**

None

### **LOCAL GOVERNMENT CONTRIBUTION PAYMENTS**

#### Arkansas

1. Arkansas Operations Payments: ACL will make certain payments to each county or appropriate taxing jurisdiction within each county in Arkansas on an annual basis for 40 years ("<u>Arkansas Annual Operations Payments</u>"). A good faith estimate of the Arkansas Annual Operations Payments as of the Effective Date are included in Exhibit A to this Schedule. The estimates will be adjusted based on the final transmission line mileage in each county.

If ACL becomes subject to property tax in Arkansas, it will pay the assessed taxes in accordance with local and state laws in lieu of the Arkansas Annual Operations Payments outlined in Exhibit A.

2. Arkansas Construction Payment: ACL will pay to each county in Arkansas one-time construction payments ("<u>Arkansas Construction Payments</u>") equal to \$7,500 per mile of transmission line in that county. A good faith estimate of the Arkansas Construction Payments as of the Effective Date is included in the table below. Final payment values are subject to the final routing of the transmission line.

County	Line Length (miles)	Arkansas Construction Payments (one-time payments)				
Cleburne	23.4	\$175,500				
Conway	21.6	\$162,000				
Crawford	28.2	\$211,500				
Cross	16.1	\$120,750				
Franklin	19.8	\$148,500				
Jackson	34.3	\$257,250				
Johnson	27.8	\$208,500				
Mississippi	16.2	\$121,500				
Poinsett	31.5	\$236,250				
Pope	27.6	\$207,000				
Van Buren	13.2	\$99,000				
White	17.5	\$131,250				
TOTAL (all counties)	277.2	\$2,079,000				

### Oklahoma

1. Oklahoma Construction Payments: PECL OK will pay to each county in Oklahoma one-time construction payments ("Oklahoma Construction Payments") equal to \$7,500 per mile of transmission line in that county. A good faith estimate of the Oklahoma Construction Payments as of the Effective Date is included in the table below. Final payment values are subject to the final routing of the transmission line.

County	Line Length (miles)	Oklahoma Construction Payments (one-time payments)				
Beaver	56	\$420,000				
Creek	27.5	\$206,250				
Garfield	22.2	\$166,500				
Harper	35.6	\$267,000				
Kingfisher	3.4	\$25,500				
Lincoln	10.1	\$75,750				
Logan	20.8	\$156,000				
Major	52.2	\$391,500				
Muskogee	39.5	\$296,250				
Okmulgee	27.7	\$207,750				
Payne	36.2	\$271,500				
Sequoyah	40.0	\$300,000				
Texas	23.9	\$179,250				
Woodward	32.4	\$243,000				
TOTAL (all counties)	427.5	\$3,206,250				

2. PECL OK will own all Project facilities located in Oklahoma and will make all applicable ad valorem tax payments to local jurisdictions, in accordance with local and state laws.

EXHIBIT A
Payment Schedule

Year	Crawford	Franklin	Johnson	Pope	Conway	Van Buren	Cleburne	White	Jackson	Poinsett	Cross	Mississippi
1	\$ 529,062	\$ 367,588	\$ 486,911	\$ 1,467,731	\$ 393,463	\$ 230,061	\$ 381,305	\$ 286,636	\$ 607,090	\$ 501,936	\$ 311,977	\$ 320,708
2	\$ 518,952	\$ 360,564	\$ 477,607	\$ 1,439,685	\$ 385,944	\$ 225,665	\$ 374,019	\$ 281,159	\$ 595,489	\$ 492,345	\$ 306,016	\$ 314,580
3	\$ 508,843	\$ 353,540	\$ 468,303	\$ 1,411,639	\$ 378,426	\$ 221,269	\$ 366,733	\$ 275,682	\$ 583,889	\$ 482,754	\$ 300,054	\$ 308,451
4	\$ 498,733	\$ 346,516	\$ 458,999	\$ 1,383,594	\$ 370,908	\$ 216,873	\$ 359,446	\$ 270,205	\$ 572,288	\$ 473,163	\$ 294,093	\$ 302,323
5	\$ 488,624	\$ 339,492	\$ 449,695	\$ 1,355,548	\$ 363,389	\$ 212,477	\$ 352,160	\$ 264,728	\$ 560,688	\$ 463,572	\$ 288,132	\$ 296,195
6	\$ 478,514	\$ 332,468	\$ 440,391	\$ 1,327,502	\$ 355,871	\$ 208,081	\$ 344,874	\$ 259,251	\$ 549,087	\$ 453,981	\$ 282,170	\$ 290,067
7	\$ 468,405	\$ 325,444	\$ 431,087	\$ 1,299,456	\$ 348,352	\$ 203,685	\$ 337,588	\$ 253,774	\$ 537,487	\$ 444,390	\$ 276,209	\$ 283,939
8	\$ 458,296	\$ 318,420	\$ 421,783	\$ 1,271,410	\$ 340,834	\$ 199,289	\$ 330,302	\$ 248,297	\$ 525,887	\$ 434,798	\$ 270,248	\$ 277,811
9	\$ 448,186	\$ 311,396	\$ 412,479	\$ 1,243,364	\$ 333,316	\$ 194,893	\$ 323,016	\$ 242,819	\$ 514,286	\$ 425,207	\$ 264,286	\$ 271,682
10	\$ 438,077	\$ 304,372	\$ 403,175	\$ 1,215,319	\$ 325,797	\$ 190,496	\$ 315,730	\$ 237,342	\$ 502,686	\$ 415,616	\$ 258,325	\$ 265,554
11	\$ 427,967	\$ 297,348	\$ 393,871	\$ 1,187,273	\$ 318,279	\$ 186,100	\$ 308,444	\$ 231,865	\$ 491,085	\$ 406,025	\$ 252,364	\$ 259,426
12	\$ 417,858	\$ 290,324	\$ 384,567	\$ 1,159,227	\$ 310,760	\$ 181,704	\$ 301,158	\$ 226,388	\$ 479,485	\$ 396,434	\$ 246,402	\$ 253,298
13	\$ 407,748	\$ 283,300	\$ 375,263	\$ 1,131,181	\$ 303,242	\$ 177,308	\$ 293,872	\$ 220,911	\$ 467,884	\$ 386,843	\$ 240,441	\$ 247,170
14	\$ 397,639	\$ 276,276	\$ 365,959	\$ 1,103,135	\$ 295,724	\$ 172,912	\$ 286,586	\$ 215,434	\$ 456,284	\$ 377,252	\$ 234,480	\$ 241,042
15	\$ 387,529	\$ 269,252	\$ 356,655	\$ 1,075,090	\$ 288,205	\$ 168,516	\$ 279,300	\$ 209,957	\$ 444,683	\$ 367,660	\$ 228,518	\$ 234,913
16	\$ 377,420	\$ 262,228	\$ 347,351	\$ 1,047,044	\$ 280,687	\$ 164,120	\$ 272,014	\$ 204,480	\$ 433,083	\$ 358,069	\$ 222,557	\$ 228,785
17	\$ 367,310	\$ 255,204	\$ 338,047	\$ 1,018,998	\$ 273,168	\$ 159,724	\$ 264,727	\$ 199,002	\$ 421,483	\$ 348,478	\$ 216,596	\$ 222,657
18	\$ 357,201	\$ 248,180	\$ 328,743	\$ 990,952	\$ 265,650	\$ 155,328	\$ 257,441	\$ 193,525	\$ 409,882	\$ 338,887	\$ 210,634	\$ 216,529
19	\$ 347,091	\$ 241,156	\$ 319,439	\$ 962,906	\$ 258,132	\$ 150,932	\$ 250,155	\$ 188,048	\$ 398,282	\$ 329,296	\$ 204,673	\$ 210,401
20	\$ 336,982	\$ 234,133	\$ 310,135	\$ 934,861	\$ 250,613	\$ 146,536	\$ 242,869	\$ 182,571	\$ 386,681	\$ 319,705	\$ 198,712	\$ 204,273
21	\$ 326,873	\$ 227,109	\$ 300,831	\$ 906,815	\$ 243,095	\$ 142,140	\$ 235,583	\$ 177,094	\$ 375,081	\$ 310,114	\$ 192,750	\$ 198,144
22	\$ 316,763	\$ 220,085	\$ 291,527	\$ 878,769	\$ 235,576	\$ 137,744	\$ 228,297	\$ 171,617	\$ 363,480	\$ 300,522	\$ 186,789	\$ 192,016
23	\$ 306,654	\$ 213,061	\$ 282,223	\$ 850,723	\$ 228,058	\$ 133,348	\$ 221,011	\$ 166,140	\$ 351,880	\$ 290,931	\$ 180,827	\$ 185,888
24	\$ 296,544	\$ 206,037	\$ 272,919	\$ 822,677	\$ 220,540	\$ 128,951	\$ 213,725	\$ 160,662	\$ 340,280	\$ 281,340	\$ 174,866	\$ 179,760
25	\$ 286,435	\$ 199,013	\$ 263,614	\$ 794,631	\$ 213,021	\$ 124,555	\$ 206,439	\$ 155,185	\$ 328,679	\$ 271,749	\$ 168,905	\$ 173,632
26	\$ 276,325	\$ 191,989	\$ 254,310	\$ 766,586	\$ 205,503	\$ 120,159	\$ 199,153	\$ 149,708	\$ 317,079	\$ 262,158	\$ 162,943	\$ 167,503
27	\$ 266,216	\$ 184,965	\$ 245,006	\$ 738,540	\$ 197,984	\$ 115,763	\$ 191,867	\$ 144,231	\$ 305,478	\$ 252,567	\$ 156,982	\$ 161,375
28	\$ 256,106	\$ 177,941	\$ 235,702	\$ 710,494	\$ 190,466	\$ 111,367	\$ 184,581	\$ 138,754	\$ 293,878	\$ 242,976	\$ 151,021	\$ 155,247
29	\$ 245,997	\$ 170,917	\$ 226,398	\$ 682,448	\$ 182,948	\$ 106,971	\$ 177,295	\$ 133,277	\$ 282,277	\$ 233,384	\$ 145,059	\$ 149,119
30	\$ 235,887	\$ 163,893	\$ 217,094	\$ 654,402	\$ 175,429	\$ 102,575	\$ 170,008	\$ 127,800	\$ 270,677	\$ 223,793	\$ 139,098	\$ 142,991
31	\$ 225,778	\$ 156,869	\$ 207,790	\$ 626,357	\$ 167,911	\$ 98,179	\$ 162,722	\$ 122,323	\$ 259,076	\$ 214,202	\$ 133,137	\$ 136,863
32	\$ 215,668	\$ 149,845	\$ 198,486	\$ 598,311	\$ 160,392	\$ 93,783	\$ 155,436	\$ 116,845	\$ 247,476	\$ 204,611	\$ 127,175	\$ 130,734
33	\$ 205,559	\$ 142,821	\$ 189,182	\$ 570,265	\$ 152,874	\$ 89,387	\$ 148,150	\$ 111,368	\$ 235,876	\$ 195,020	\$ 121,214	\$ 124,606
34	\$ 195,450	\$ 135,797	\$ 179,878	\$ 542,219	\$ 145,356	\$ 84,991	\$ 140,864	\$ 105,891	\$ 224,275	\$ 185,429	\$ 115,253	\$ 118,478
35	\$ 185,340	\$ 128,773	\$ 170,574	\$ 514,173	\$ 137,837	\$ 80,595	\$ 133,578	\$ 100,414	\$ 212,675	\$ 175,838	\$ 109,291	\$ 112,350
36	\$ 175,231	\$ 121,749	\$ 161,270	\$ 486,127	\$ 130,319	\$ 76,199	\$ 126,292	\$ 94,937	\$ 201,074	\$ 166,246	\$ 103,330	\$ 106,222
37	\$ 165,121	\$ 114,725	\$ 151,966	\$ 458,082	\$ 122,800	\$ 71,803	\$ 119,006	\$ 89,460	\$ 189,474	\$ 156,655	\$ 97,369	\$ 100,094
38	\$ 155,012	\$ 107,701	\$ 142,662	\$ 430,036	\$ 115,282	\$ 67,406	\$ 111,720	\$ 83,983	\$ 177,873	\$ 147,064	\$ 91,407	\$ 93,965
39	\$ 144,902	\$ 100,677	\$ 133,358	\$ 401,990	\$ 107,764	\$ 63,010	\$ 104,434	\$ 78,506	\$ 166,273	\$ 137,473	\$ 85,446	\$ 87,837
40	\$ 134,793	\$ 93,653	\$ 124,054	\$ 373,944	\$ 100,245	\$ 58,614	\$ 97,148	\$ 73,028	\$ 154,673	\$ 127,882	\$ 79,485	\$ 81,709

# Schedule 5 to Participation Agreement

## Capitalization

None

#### Officer's Certificate Regarding DOE Delegated Real Estate Rights

#### [DATE]

The undersigned, [NAME], [TITLE] of Plains and Eastern Clean Line Holdings LLC (the "Company") does hereby certify on behalf of the Company, pursuant to Section 6.2(c) of the Participation Agreement dated as of [DATE] (the "Participation Agreement"), by and among the Company, Arkansas Clean Line LLC, Plains and Eastern Clean Line Oklahoma LLC, Oklahoma Land Acquisition Company LLC, Plains and Eastern Clean Line LLC, and the United States Department of Energy, as follows:

- (i) the Clean Line Entities have complied with all of the requirements and procedures set forth in Schedule 1 to the Participation Agreement with respect to the DOE Delegated Real Estate Rights to be acquired;
- (ii) Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments in respect of not less than 1,500 MW of the Electrical Capacity in the aggregate (calculated as the sum of (A) with respect to Acceptable Permitted Project Investments, the sum of each portion of the Electrical Capacity transferred (and for which the Clean Line Entities have received payment or will receive payment within three (3) years after the date of such Permitted Project Investment) pursuant to each Acceptable Permitted Project Investment Commitment and (B) with respect to the Acceptable Transmission Services Agreements, the sum of the average Electrical Capacity committed in the initial five (5) years of the term for each such Acceptable Transmission Services Agreement) are in full force and effect and no event has occurred or is continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Acceptable Transmission Services Agreement or such Acceptable Permitted Project Investment Commitment;
- (iii) [the Converter Station Real Estate Rights Agreements are in full force and effect][the Clean Line Entities own in fee free and clear of all Liens other than Permitted Liens all Real Estate Rights necessary for the construction of the Converter Station Facility and the Intermediate Converter Station]<sup>1</sup>;
- (iv) the executed Interconnection Agreements for interconnection of the Project with the SPP-controlled transmission system and the MISO-controlled transmission system and the executed Project Work Agreement or Interconnection Agreement for interconnection of the TN Facilities with the TVA transmission system are in full force and effect and no event has occurred and is continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Interconnection Agreements or

<sup>&</sup>lt;sup>1</sup> Insert as appropriate.

Project Work Agreement (except to the extent the Project Work Agreement has been replaced by an Interconnection Agreement with TVA);

- (v) [the Project Equity Commitments (including Firm Project Equity Commitments that are currently in full force and effect and that provide for commitments (together with amounts on deposit in the Advance Funding Account) in an amount equal to not less than 150% of the Remaining DOE Acquisition Costs as of any date on which any Clean Line Entity designates any Project Real Estate Right as a DOE Delegated Real Estate Right), Project Financing Commitments and any letters of intent delivered as a condition to the occurrence of the Commencement Date are in full force and effect][the Financing Condition is satisfied]<sup>2</sup>;
- (vi) (A) no Governmental Order is in effect nor has any Change of Law occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (1) DOE's performance under the Participation Agreement or any other Transaction Document currently in effect or (2) DOE's participation in the Project and (B) no other final and non-appealable Governmental Order is in effect nor has any Change of Law occurred that, in either case, as applicable, (1) sets aside, enjoins or legally prohibits any Clean Line Entity's performance under the Participation Agreement or any other Transaction Document currently in effect or any Clean Line Entity's performance in any material respect under any Material Project Contract to which it is a party or (2) affects in any material respect the legality, validity or enforceability of any of the Transaction Documents, any Project Equity Commitment or any of the Material Project Contracts currently in effect; and
  - (vii) no Event of Default has occurred and is continuing.

It is expressly understood that this Officer's Certificate is being executed by the undersigned authorized signatory solely on behalf of the Company, and not in a personal capacity. Capitalized terms used but not defined herein shall have the meanings given to them in the Participation Agreement.

[Signature on following page]

<sup>&</sup>lt;sup>2</sup> Insert as appropriate.

IN WITNESS WHEREOF, the undersigned has executed this Officer's Certificate on behalf of the Company as of the date first above written.	
	Name: Title:

#### Officer's Certificate Regarding Acquisitions by Condemnation

#### [DATE]

The undersigned, [NAME], [TITLE] of Plains and Eastern Clean Line Holdings LLC (the "Company") does hereby certify on behalf of the Company, pursuant to Section 6.3(c) of the Participation Agreement dated as of [DATE] (the "Participation Agreement"), by and among the Company, Arkansas Clean Line LLC, Plains and Eastern Clean Line Oklahoma LLC, Oklahoma Land Acquisition Company LLC, Plains and Eastern Clean Line LLC, and the United States Department of Energy, as follows:

- (i) the Clean Line Entities have complied with all of the requirements and procedures set forth in Schedule 1 to the Participation Agreement with respect to the DOE Delegated Real Estate Rights to be acquired;
- (ii) Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments in respect of not less than 2,000 MW of the Electrical Capacity in the aggregate (calculated as the sum of (A) with respect to Acceptable Permitted Project Investments, the sum of each portion of the Electrical Capacity transferred (and for which the Clean Line Entities have received payment or will receive payment within three (3) years after the date of such Permitted Project Investment) pursuant to each Acceptable Permitted Project Investment Commitment and (B) with respect to the Acceptable Transmission Services Agreements, the sum of the average Electrical Capacity committed in the initial five (5) years of the term for each such Acceptable Transmission Services Agreement) are in full force and effect and no event has occurred and is continuing (whether as a result of a default or the failure of a condition precedent or otherwise) that gives the Project Participant party thereto the right to terminate such Acceptable Transmission Services Agreement or such Acceptable Permitted Project Investment Commitment;
- (iii) [the Converter Station Real Estate Rights Agreements delivered pursuant to the foregoing conditions precedent are in full force and effect and neither any Clean Line Entity nor any other Person party thereto is in default thereunder][the Clean Line Entities own in fee free and clear of Liens other than Permitted Liens all Real Estate Rights necessary for the construction of the Converter Station Facility and the Intermediate Converter Station]<sup>1</sup>;
- (iv) the Interconnection Agreements delivered by the Company pursuant to Section 6.3(a)(vi) of the Participation Agreement are in full force and effect and neither any Clean Line Entity nor any other Person or party thereto is in default thereunder;
  - (v) the Financing Condition is satisfied;

<sup>&</sup>lt;sup>1</sup> Insert as appropriate.

- (vi) (A) no Governmental Order is in effect nor has any Change of Law occurred that, in either case, as applicable, sets aside, enjoins or legally prohibits (1) DOE's performance under the Participation Agreement or any other Transaction Document currently in effect or (2) DOE's participation in the Project and (B) no other final and non-appealable Governmental Order is in effect nor has any Change of Law occurred that, in either case, as applicable, (1) sets aside, enjoins or legally prohibits any Clean Line Entity's performance under the Participation Agreement or any other Transaction Document currently in effect or any Clean Line Entity's performance in any material respect under any Material Project Contract to which it is a party or (2) affects in any material respect the legality, validity or enforceability of any of the Transaction Documents, any Project Equity Commitment or any of the Material Project Contracts currently in effect; and
  - (vii) no Event of Default has occurred and is continuing.

It is expressly understood that this Officer's Certificate is being executed by the undersigned authorized signatory solely on behalf of the Company, and not in a personal capacity. Capitalized terms used but not defined herein shall have the meanings given to them in the Participation Agreement.

[Signature on following page]

IN WITNESS WHEREOF, the undersigned has executed this Officer's Certificate on behalf of the Company as of the date first above written.	
	Name: Title:

# Schedule 8 to Participation Agreement

## Reserved

# Schedule 9 to Participation Agreement

## Litigation

None

# Schedule 10 to Participation Agreement

### **Environmental Matters**

None

# Schedule 11 to Participation Agreement

## **Governmental Approvals**

1. Section 1222 Decision

# Schedule 12 to Participation Agreement

## **Code of Conduct for Acquisitions of Project Real Estate Rights**

[Attached]

## PLAINS & EASTERN

CLEAN LINE

# PLAINS AND EASTERN CLEAN LINE OKLAHOMA LLC CODE OF CONDUCT FOR EMPLOYEES, RIGHT-OF-WAY AGENTS AND SUBCONTRACTOR EMPLOYEES

This Code of Conduct applies to all communications and interactions with property owners and occupants of property by all employees, right-of-way agents and subcontractor employees representing Plains and Eastern Clean Line Oklahoma LLC ("Plains & Eastern") in the negotiation of right-of-way and the performance of surveying, environmental assessments and the other activities for the Plains & Eastern project (the "Project") on property not owned by Plains & Eastern.

- I. All communications with property owners and occupants must be factually correct and made in good faith.
  - a. Do provide maps and documents necessary to keep the landowner properly informed.
  - b. Do not make false or misleading statements.
  - c. Do not purposely or intentionally misrepresent any fact.
  - d. If you do not know the answer to a question, do not speculate about the answer. Advise the property owner that you will investigate the question and provide an answer later.
  - e. Follow-up in a timely manner on all commitments to provide additional information.
  - f. Do not send written communications suggesting an agreement has been reached when, in fact, an agreement has not been reached.
  - g. If information provided is subsequently determined to be incorrect, follow up with the landowner as soon as practical to provide the corrected information.
  - h. Do provide the landowner with appropriate contact information should additional contacts be necessary.
- II. All communications and interactions with property owners and occupants of property must be respectful and reflect fair dealing.
  - a. When contacting a property owner in person, promptly identify yourself as representing Plains & Eastern.
  - b. When contacting a property owner by telephone, promptly identify yourself as representing Plains & Eastern.
  - c. Do not engage in behavior that may be considered harassing, coercive, manipulative, intimidating or causing undue pressure.
  - d. All communications by a property owner, whether in person, by telephone or in writing, in which the property owner indicates that he or she does not want to negotiate or does not want to give permission for surveying or other work on his or her property, must be respected and politely accepted without argument. Unless specifically authorized by Plains & Eastern, do not contact the property owner again regarding negotiations or requests for permission.
  - e. When asked to leave the property, promptly leave and do not return unless specifically authorized by Plains & Eastern.
  - f. If discussions with the property owner become acrimonious, politely discontinue the discussion and withdraw from the situation.
  - g. Obtain unequivocal permission to enter the property for purposes of surveying or conducting environmental assessments or other activities. Clearly explain to the property owner the scope of the work to be conducted based on the permission given.



- Attempt to notify the occupant of the property each time you enter the property based on this permission.
- h. Do not represent that a relative, neighbor and/or friend have signed a document or reached an agreement with Plains & Eastern.
- i. Do not ask a relative, neighbor and/or friend of a property owner to convince the property owner to take any action.
- j. Do not represent that a relative, neighbor and/or friend supports or opposes the Project.
- k. Do not suggest that any person should be ashamed of or embarrassed by his or her opposition to the Project or that such opposition is inappropriate.
- I. Do not suggest that an offer is "take it or leave it."
- m. Do not argue with property owners about the merits of the Project.
- n. Do not threaten to call law enforcement officers or obtain court orders.
- o. Avoid discussing a property owner's failure to note an existing easement when purchasing the property and other comments about the property owner's acquisition of the property.
- p. Do not threaten the use of eminent domain.
- III. All communications and interactions with property owners and occupants of property must respect the privacy of property owners and other persons.
  - a. Discussions with property owners and occupants are to remain confidential.
  - b. Do not discuss your negotiations or interactions with other property owners or other persons unaffiliated with Plains & Eastern.
  - c. Do not ask relatives, neighbors and/or friends to influence the property owner or any other person.



## PLAINS & EASTERN

CLEAN LINE

# ARKANSAS CLEAN LINE LLC CODE OF CONDUCT FOR EMPLOYEES, RIGHT-OF-WAY AGENTS AND SUBCONTRACTOR EMPLOYEES

This Code of Conduct applies to all communications and interactions with property owners and occupants of property by all employees, right-of-way agents and subcontractor employees representing Arkansas Clean Line LLC ("Plains & Eastern") in the negotiation of right-of-way and the performance of surveying, environmental assessments and the other activities for the Plains & Eastern project (the "Project") on property not owned by Plains & Eastern.

- I. All communications with property owners and occupants must be factually correct and made in good faith.
  - a. Do provide maps and documents necessary to keep the landowner properly informed.
  - b. Do not make false or misleading statements.
  - c. Do not purposely or intentionally misrepresent any fact.
  - d. If you do not know the answer to a question, do not speculate about the answer. Advise the property owner that you will investigate the question and provide an answer later.
  - e. Follow-up in a timely manner on all commitments to provide additional information.
  - f. Do not send written communications suggesting an agreement has been reached when, in fact, an agreement has not been reached.
  - g. If information provided is subsequently determined to be incorrect, follow up with the landowner as soon as practical to provide the corrected information.
  - h. Do provide the landowner with appropriate contact information should additional contacts be necessary.
- II. All communications and interactions with property owners and occupants of property must be respectful and reflect fair dealing.
  - a. When contacting a property owner in person, promptly identify yourself as representing Plains & Eastern.
  - b. When contacting a property owner by telephone, promptly identify yourself as representing Plains & Eastern.
  - c. Do not engage in behavior that may be considered harassing, coercive, manipulative, intimidating or causing undue pressure.
  - d. All communications by a property owner, whether in person, by telephone or in writing, in which the property owner indicates that he or she does not want to negotiate or does not want to give permission for surveying or other work on his or her property, must be respected and politely accepted without argument. Unless specifically authorized by Plains & Eastern, do not contact the property owner again regarding negotiations or requests for permission.
  - e. When asked to leave the property, promptly leave and do not return unless specifically authorized by Plains & Eastern.
  - f. If discussions with the property owner become acrimonious, politely discontinue the discussion and withdraw from the situation.
  - g. Obtain unequivocal permission to enter the property for purposes of surveying or conducting environmental assessments or other activities. Clearly explain to the property owner the scope of the work to be conducted based on the permission given.



- Attempt to notify the occupant of the property each time you enter the property based on this permission.
- h. Do not represent that a relative, neighbor and/or friend have signed a document or reached an agreement with Plains & Eastern.
- i. Do not ask a relative, neighbor and/or friend of a property owner to convince the property owner to take any action.
- j. Do not represent that a relative, neighbor and/or friend supports or opposes the Project.
- k. Do not suggest that any person should be ashamed of or embarrassed by his or her opposition to the Project or that such opposition is inappropriate.
- I. Do not suggest that an offer is "take it or leave it."
- m. Do not argue with property owners about the merits of the Project.
- n. Do not threaten to call law enforcement officers or obtain court orders.
- o. Avoid discussing a property owner's failure to note an existing easement when purchasing the property and other comments about the property owner's acquisition of the property.
- p. Do not threaten the use of eminent domain.
- III. All communications and interactions with property owners and occupants of property must respect the privacy of property owners and other persons.
  - a. Discussions with property owners and occupants are to remain confidential.
  - b. Do not discuss your negotiations or interactions with other property owners or other persons unaffiliated with Plains & Eastern.
  - c. Do not ask relatives, neighbors and/or friends to influence the property owner or any other person.



#### Permitted Project Investments Representations and Warranties and Covenants

Each Person that makes a Permitted Project Investment in the Project (such Person, an "Investor") shall be required to make the following representations and warranties and agree to the following covenants for the benefit of DOE in the connection with such Permitted Project Investment:

#### **Representations and Warranties:**

<u>Prohibited Persons</u>. (a) None of the Investor, any Controlling Person of such Investor or any Principal Person of such Investor or any Principal Person of such Controlling Person of such Investor is a Prohibited Person and (b) to such Investor's knowledge, no event has occurred and no condition exists that is likely to result in such Investor, any Controlling Person of such Investor, any Principal Person of such Investor or any Principal Person of such Controlling Person of such Investor becoming a Prohibited Person.

Anti-Corrupt Practices Laws, Etc. (a) The Investor, each Controlling Person of such Investor and each Principal Person, employee and agent of such Investor and each Principal Person of each Controlling Person of such Investor have complied with all AM Laws, Anti-Corruption Laws and Sanctions and (b) such Investor has implemented and maintains in effect policies and procedures reasonably designed to ensure compliance by such Investor and its Principal Persons with AM Law, Anti-Corruption Laws and Sanctions.

#### **Covenants:**

#### Prohibited Persons.

- (a) The Investor shall provide immediate written notice (including a brief description relating thereto) to DOE if, at any time, it learns that the representations made with respect to Prohibited Persons (including in respect of the Debarment Regulations) were erroneous when made or have become erroneous by reason of changed circumstances.
- (b) If any Person that controls the Investor or any of their respective Principal Persons becomes a Prohibited Person, the Investor shall, within sixty (60) days of knowing that such Person has become a Prohibited Person, engage and continue to engage in constructive discussions with DOE regarding the removal or replacement of such Person or, if such removal or replacement is not reasonably feasible, the implementation of other mitigation matters.

#### AM Laws, Anti-Corruption Laws Etc.

The Investor shall and shall cause its Principal Persons, employees and agents to (a) comply with all applicable AM Laws and Anti-Corruption Laws, (b) conduct its business in compliance with all applicable AM Laws and Anti-Corruption Laws and (c) maintain internal management and accounting practices and controls that are adequate to ensure the Investor's compliance with all applicable AM Laws and Anti-Corruption Laws.

#### Definitions:

"Affiliate" means with respect to any Person, any other Person that directly or indirectly Controls, or is under common Control with, or is Controlled by, such Person and, if such Person is an individual, any member of the immediate family of such individual and any trust whose principal beneficiary is such individual or one or more members of such immediate family and any Person who is Controlled by any such member or trust.

"AM Laws" means, with respect to any Person, all applicable laws concerning or relating to anti-money laundering.

"Anti-Corruption Laws" means, with respect to any Person, all Applicable Laws concerning or relating to bribery or corruption, including, the Foreign Corrupt Practices Act of 1977 (Pub. L. No. 95 213, §§101-104).

"Control" means (including, with its correlative meanings, "Controlled by" and "under common Control with") as used with respect to any Person, possession, directly or indirectly, of the power to direct or cause the direction of management or policies of such Person (whether through ownership of voting securities or partnership or other ownership interests, by contract or otherwise); provided, that any Person that owns directly or indirectly ten percent (10%) or more of the equity interests having ordinary voting powers for the election of directors or other applicable governing body of another Person (but excluding limited partnership or similar types of ownership interests and tax equity investors) shall be deemed to Control such other Person.

"Controlling Person" means, with respect to any Person, any other Person that, directly or indirectly Controls such Person.

"<u>Debarment Regulations</u>" means (a) the Government-wide Debarment and Suspension (Non procurement) regulations (Common Rule), 53 Fed. Reg. 19204 (May 26, 1988), (b) Subpart 9.4 (Debarment, Suspension, and Ineligibility) of the Federal Acquisition Regulations, 48 C.F.R. §§ 9.400 – 9.409 and (c) the revised Government-wide Debarment and Suspension (Non-procurement) regulations (Common Rule), 60 Fed. Reg. 33037 (June 26, 1995).

"Governmental Authority" means any federal, state, county, municipal, or regional authority, or any other entity of a similar nature, exercising any executive, legislative, judicial (including any court of competent jurisdiction), regulatory, or administrative function of government.

"Governmental Order" means with respect to any Person, any judgment, order, decision, or decree, or any action of a similar nature, of or by a Governmental Authority having competent jurisdiction over such Person or any of its properties.

"OFAC" means the Office of Foreign Assets Control, an agency of the U.S. Department of the Treasury under the auspices of the Under Secretary of the Treasury for Terrorism and Financial Intelligence.

"OFAC-Listed Person" has the meaning set forth in clause (a) of the definition of Prohibited Person.

"OFAC Sanctions Program" means any economic or trade sanction that OFAC is responsible for administering and enforcing. A list of OFAC Sanctions Programs may be found at http://www.treasury.gov/resource-center/sanctions/Programs/Pages/Programs.aspx.

"Person" means any individual, entity, firm, corporation, company, voluntary association, partnership, limited liability company, joint venture, trust, unincorporated organization, Governmental Authority, committee, department, authority or any other body, incorporated or unincorporated, whether having distinct legal personality or not.

"<u>Principal Person</u>" means, with respect to any Person, any officer, director, owner, key employee or other Person with primary management or supervisory responsibilities with respect to such Person or any other Person (whether or not an employee) who has critical influence on or substantive control over such Person.

"<u>Prohibited Person</u>" means any Person (or any Person that is an Affiliate of a Person) that is:

- (a) named, identified or described on the list of "Specially Designated Nationals and Blocked Persons" as published by OFAC (an "OFAC-Listed Person");
- (b) is an agent, department or instrumentality of, or is otherwise beneficially owned by, Controlled by or acting on behalf of, directly or indirectly, (i) an OFAC-Listed Person or (ii) any Person, organization, foreign country or regime that is subject to any Sanctions;
- (c) is debarred, suspended, proposed for debarment with a final determination still pending, declared ineligible or voluntarily excluded (as such terms are defined in the Debarment Regulations) from contracting with any United States federal government department or any agency or instrumentality thereof or otherwise participating in procurement or non-procurement transactions with any United States federal government or agency pursuant to any of the Debarment Regulations;
- (d) indicted, convicted or had a final and non-appealable Governmental Order rendered against it for any of the offenses listed in any of the Debarment Regulations; or

(e) otherwise blocked, subject to sanctions under or engaged in any activity in violation of other United States economic sanctions, including but not limited to, the Trading with the Enemy Act, the International Emergency Economic Powers Act, the Comprehensive Iran Sanctions, Accountability and Divestment Act or any similar law or regulation with respect to Iran or any other country, the Sudan Accountability and Divestment Act, any OFAC Sanctions Program, or any economic sanctions regulations administered and enforced by the United States or any enabling legislation or executive order relating to any of the foregoing.

"Sanctions" means economic or financial sanctions or trade embargoes or restrictive measures enacted, imposed, administered or enforced from time to time by (a) the United States government, including those administered by OFAC, the U.S. Department of State or the U.S. Department of Commerce, (b) the United Nations Security Council, (c) the European Union or any of its member states or (d) any other applicable Governmental Authority and including, for the avoidance of doubt, the Trading with the Enemy Act, the International Emergency Economic Powers Act, the Comprehensive Iran Sanctions, Accountability and Divestment Act and the Sudan Accountability and Divestment Act.

#### Conditions in Acceptable Transmission Services Agreements and Acceptable Permitted Project Investment Commitments

#### Part A

#### **Acceptable Transmission Services Agreements**

Acceptable Transmission Services Agreements delivered pursuant to Sections 6.2(a)(iv), 6.2(b)(ii), 6.3(a)(v), 6.3(b)(ii) and 6.4(a)(ii) of the Agreement may be subject to conditions to their effectiveness or commencement of transmission service thereunder similar in substance to the following:

- 1. <u>FERC Approval</u>. FERC has issued one or more orders under Section 205 of the FPA accepting the Acceptable Transmission Services Agreement for filing without substantial condition or modification that is materially adverse to either the Clean Line Parties or the customer under the Acceptable Transmission Services Agreement.
- 2. <u>State Regulatory Approval</u>. Any necessary approvals from state public utility regulatory bodies have been obtained, made or given authorizing all transactions and payments contemplated in the Acceptable Transmission Services Agreement, including the approval of power purchase agreements, as applicable, on terms that are acceptable to the customer and the Clean Line Parties.

#### 3. <u>Applicable Federal Regulatory Approvals.</u>

- (a) If and to the extent that the customer under an Acceptable Transmission Services Agreement is a federal entity that must comply with certain federal environmental or regulatory reviews and approvals prior to the effectiveness of an Acceptable Transmission Services Agreement, the effectiveness of such Acceptable Transmission Services Agreement may be conditioned upon completion of such federal or environmental regulatory reviews and approvals and such reviews and approvals not being subject to judicial review.
- (b) If the Acceptable Transmission Services Agreement will involve a source of energy that must receive Governmental Approvals prior to the initiation of commercial operation and generation of electricity for delivery to, and transmittal over, the Project; such Acceptable Transmission Services Agreement may be conditioned upon Governmental Approvals having been duly obtained, in full force and effect and not subject to any pending or threatened challenges or judicial review.
- 4. <u>OATT</u>. FERC has issued an order under Section 205 of the FPA accepting the rate schedule or OATT filed by the Clean Line Parties without substantial condition or modification that is materially adverse to either the Clean Line Parties or the customer under the Acceptable Transmission Services Agreement.

- 5. <u>Interconnection Agreements</u>. All necessary interconnection agreements involving the interconnection of the generation facilities under contract with or owned or operated by the customer under the Acceptable Transmission Services Agreement to the Project have been executed and are in full force and effect.
- 6. <u>Governmental Approvals</u>. All non-ministerial Governmental Approvals required for the construction and operation of the Project have been duly obtained and are in full force and effect and shall not be subject to any pending or threatened challenge or judicial review.
- 7. <u>Material Contracts</u>. All material contracts needed to construct and operate the Project, including interconnection agreements involving the interconnection of the Project to the transmission systems under the operational control of MISO, SPP and TVA have been executed and are in full force and effect.
- 8. <u>Notice to Proceed</u>. A full notice to proceed has been issued under each material construction contract with respect to the Project.
- 9. <u>Project Financial Close</u>. Project Financial Close has occurred or shall occur simultaneously with the effectiveness of the Acceptable Transmission Services Agreement.
- 10. Representations and Warranties. Each of the applicable representations and warranties of the Clean Line Parties made pursuant to the Acceptable Transmission Services Agreement or any related agreement is true and correct in all material respects, except to the extent that any such representation or warranty is expressly made only as of another date, in which case such representation and warranty shall be true and correct in all material respects as of such other date.
- 11. <u>Performance of Obligations</u>. Each Clean Line Party shall have performed in all material respects all obligations required to be performed by such party under the Acceptable Transmission Services Agreement (and any related agreements) and under any applicable transaction documents pertaining to the Project or the Other Facilities.
- 12. <u>Security</u>. Any necessary security or collateral among and between the Clean Line Parties and the customer under the Acceptable Transmission Services Agreement is effective.
- 13. <u>General Open Access Terms and Conditions</u>. The Acceptable Transmission Services Agreement will include, either directly or by reference, general terms and conditions consistent with FERC's open access transmission rules and policies as applied to firm point-to-point transmission service under the rate schedule or OATT filed by the Clean Line Parties (e.g., sale or assignment of transmission service).
- 14. <u>Insurance</u>. The Clean Line Parties have delivered to the customer certificates evidencing all insurance required to be obtained by the Clean Line Parties under the Acceptable Transmission Services Agreement.
- 15. <u>Commercial Operation</u>. The Project (a) has commenced commercial operation and has satisfied the requirements for "substantial completion" (or term of similar import) as defined in and in accordance with all Construction Contracts and the initial Electrical Capacity

(as specified in the definition thereof) of the Project has been certified by an Independent Engineer, (b) has safely and reliably energized and energy may be delivered across the Project Facilities to SPP's, MISO's and TVA's transmission systems in accordance with the Interconnection Agreements and (c) otherwise complies with the design criteria, system performance and testing requirements and operating standards set forth in the Acceptable Transmission Services Agreement and the Clean Line Parties have delivered to the customer a certificate certifying, or the customer has approved, the same.

16. <u>Legal Opinion</u>. Delivery by the Clean Line Parties to the customer of an opinion of legal counsel that all Governmental Approvals required for the Clean Line Parties to own, construct and operate the Project and to perform their respective obligations under the Acceptable Transmission Services Agreement have been obtained.

#### **Acceptable Permitted Project Investment Commitments**

Acceptable Permitted Project Investment Commitments delivered pursuant to Sections 6.2(a)(iv), 6.2(b)(ii), 6.3(a)(v), 6.3(b)(ii) and 6.4(a)(ii) of the Agreement may be subject to conditions to their effectiveness thereunder similar in substance to the following:

- 1. <u>FERC Approval</u>. FERC has issued one or more orders approving the Permitted Project Investment without substantial condition or modification that is materially adverse to either the Clean Line Parties or Permitted Project Investment counterparty.
- 2. <u>State Regulatory Approval</u>. Any necessary approvals from state public utility regulatory bodies have been obtained, made or given authorizing all transactions and payments contemplated in Permitted Project Investment on terms that are acceptable to the Permitted Project Investment counterparty and the Clean Line Parties.
- 3. <u>Interconnection Agreements</u>. All necessary interconnection agreements involving the interconnection of the generation facilities under contract with or owned or operated by the Permitted Project Investment counterparty to the Project have been executed and are in full force and effect.
- 4. <u>Governmental Approvals</u>. All non-ministerial Governmental Approvals required for the construction and operation of the Project have been duly obtained and are in full force and effect and shall not be subject to any pending or threatened challenge or judicial review.
- 5. <u>Material Contracts</u>. All material contracts needed to construct and operate the Project, including interconnection agreements involving the interconnection of the Project to the transmission systems under the operational control of MISO, SPP and TVA have been executed and are in full force and effect.
- 6. <u>Notice to Proceed</u>. A full notice to proceed has been issued under each material construction contract with respect to the Project.

- 7. <u>Project Financial Close</u>. Project Financial Close has occurred or shall occur simultaneously with the effectiveness of the Permitted Project Investment.
- 8. Representations and Warranties. Each of the applicable representations and warranties of the Clean Line Parties made pursuant to the Permitted Project Investment or any related agreement is true and correct in all material respects, except to the extent that any such representation or warranty is expressly made only as of another date, in which case such representation and warranty shall be true and correct in all material respects as of such other date.
- 9. <u>Performance of Obligations</u>. Each Clean Line Party shall have performed in all material respects all obligations required to be performed by such party under the Permitted Project Investment (and any related agreements) and under any applicable transaction documents pertaining to the Project or the Other Facilities.
- 10. <u>Security</u>. Any necessary security or collateral among and between the Clean Line Parties and the Permitted Project Investment counterparty under the Permitted Project Investment (and any related agreements) is effective.
- 11. <u>Insurance</u>. The Clean Line Parties have delivered to the Permitted Project Investment counterparty certificates evidencing all insurance required to be obtained by the Clean Line Parties under the Permitted Project Investment (and any related agreements).
- 12. <u>Commercial Operation</u>. The Project (a) has commenced commercial operation and has satisfied the requirements for "substantial completion" (or term of similar import) as defined in and in accordance with all Construction Contracts and the initial Electrical Capacity (as specified in the definition thereof) of the Project has been certified by an Independent Engineer, (b) has safely and reliably energized and energy may be delivered across the Project Facilities to SPP's, MISO's and TVA's transmission systems in accordance with the Interconnection Agreements and (c) otherwise complies with the design criteria, system performance and testing requirements and operating standards set forth in the Permitted Project Investment and the Clean Line Parties have delivered to the Permitted Project Investment counterparty a certificate certifying, or the Permitted Project Investment counterparty has approved, the same.
- 13. <u>Legal Opinion</u>. Delivery by the Clean Line Parties to the Permitted Project Investment counterparty of an opinion of legal counsel that all Governmental Approvals required for the Clean Line Parties to own, construct and operate the Project and to perform their respective obligations under the Permitted Project Investment (and any related agreements) have been obtained.

#### Part B

Acceptable Transmission Services Agreements delivered pursuant to Sections 6.2(a)(iv), 6.2(b)(ii), 6.3(a)(v), 6.3(b)(ii) and 6.4(a)(ii) of the Agreement shall not be terminated before the end of the term except as set forth below:

1. <u>Failure to Perform</u>. The Acceptable Counterparty shall have the right to terminate in the event another party to the Acceptable Transmission Services Agreement has failed to perform a material obligation under the Acceptable Transmission Services Agreement and the other party has not cured such failure within a reasonable period of time.

#### **DAVIS-BACON ACT REQUIREMENTS**

#### SECTION 1.

#### (a) <u>Minimum wages</u>.

(i) All laborers and mechanics employed or working upon the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor which is attached hereto and made a part hereof, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of Section 1(a)(iv) of this Schedule 15; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in Section 1(d) below. Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein; provided, that the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under Section 1(a)(ii) of this Schedule 15) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

(ii) (A) The contracting officer shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The contracting officer shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:

- (1) The work to be performed by the classification requested is not performed by a classification in the wage determination:
- (2) The classification is utilized in the area by the construction industry; and
- (3) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.
- (B) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the contracting officer agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the contracting officer to the Administrator of the Wage and Hour Division, Employment Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.
- (C) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the contracting officer do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the contracting officer shall refer the questions, including the views of all interested parties and the recommendation of the contracting officer, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.
- (D) The wage rate (including fringe benefits where appropriate) determined pursuant to Sections 1(a)(ii)(B) or (C) of this Schedule 15, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.
- (iii) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.
- (iv) If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program; <u>provided</u>, that the Secretary of Labor has found, upon the written

request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.

(b) <u>Withholding</u>. The Department of Energy shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld from the contractor under this contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), all or part of the wages required by the contract, the Department of Energy may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.

#### (c) Payrolls and basic records.

- Payrolls and basic records relating thereto shall be maintained by (i) the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work (or under the United States Housing Act of 1937, or under the Housing Act of 1949, in the construction or development of the project). Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected, and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.
- (ii) (A) The contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to the (write in name of appropriate federal agency) if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit the payrolls to the applicant, sponsor, or owner, as the

case may be, for transmission to the Department of Energy. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under Section 1(c)(i) of this Schedule 15, except that full social security numbers and home addresses shall not be included on weekly transmittals. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH–347 is available for this purpose from the Wage and Hour Division Web http://www.dol.gov/esa/whd/forms/wh347instr.htm or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to the (write in name of appropriate federal agency) if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit them to the applicant, sponsor, or owner, as the case may be, for transmission to the Department of Energy, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of this Section 1(c)(ii)(A) of this Schedule 15 for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the sponsoring government agency (or the applicant, sponsor, or owner).

- (B) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:
  - (1) That the payroll for the payroll period contains the information required to be provided under  $\underline{Section\ 1(c)(ii)}$  of this Schedule 15, the appropriate information is being maintained under  $\underline{Section\ 1(c)(i)}$  of this Schedule 15, and that such information is correct and complete;
  - (2) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in Regulations, 29 CFR part 3;
  - (3) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.

- (C) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH– 347 shall satisfy the requirement for submission of the "Statement of Compliance" required by Section 1(c)(ii)(B) of this Schedule 15.
- (D) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under section 1001 of title 18 and section 231 of title 31 of the United States Code.
- (iii) The contractor or subcontractor shall make the records required under Section 1(c)(i) of this Schedule 15 available for inspection, copying, or transcription by authorized representatives of the Department of Energy or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the Department of Energy may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

#### (d) Apprentices and trainees.

Apprentices. Apprentices will be permitted to work at less than the (i) predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

- Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program, the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (iii) <u>Equal employment opportunity</u>. The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR part 30.

- (e) <u>Compliance with Copeland Act requirements</u>. The contractor shall comply with the requirements of 29 CFR part 3, which are incorporated by reference in this contract.
- (f) <u>Subcontracts</u>. The contractor or subcontractor shall insert in any subcontracts the clauses contained in <u>Sections 1(a)</u> through (j) of this Schedule 15 and such other clauses as the Department of Energy may by appropriate instructions require, and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in 29 CFR 5.5.
- (g) <u>Contract termination: debarment</u>. A breach of the contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in 29 CFR 5.12.
- (h) <u>Compliance with Davis-Bacon and Related Act requirements</u>. All rulings and interpretations of the Davis-Bacon and Related Acts contained in 29 CFR parts 1, 3, and 5 are herein incorporated by reference in this contract.
- (i) <u>Disputes concerning labor standards</u>. Disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and the contracting agency, the U.S. Department of Labor, or the employees or their representatives.

#### (j) <u>Certification of eligibility</u>.

- (i) By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (ii) No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (iii) The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

#### SECTION 2.

(a) <u>Overtime Requirements</u>. No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such

laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek.

- (b) <u>Violation; liability for unpaid wages; liquidated damages</u>. In the event of any violation of the clause set forth in <u>Section 2(a)</u> of this Schedule 15 the contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in <u>Section 2(a)</u> of this Schedule 15, in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in <u>Section 2(a)</u> of this Schedule 15.
- (c) Withholding for unpaid wages and liquidated damages. The Department of Energy shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in Section 2(b) of this Schedule 15.
- (d) <u>Subcontracts</u>. The contractor or subcontractor shall insert in any subcontracts the clauses set forth in <u>Sections 2(a)</u> through <u>(d)</u> of this Schedule 15 and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in <u>Sections 2(a)</u> through <u>(d)</u> of this Schedule 15.

# Schedule 16 to Participation Agreement

## **Required Approvals**

1. Section 1222 Decision

#### CLEAN LINE UNIFORM ACT EXECUTION PLAN

The following materials set out the guidelines and procedures that the Clean Line Entities will follow for acquisition of Project Real Estate Rights in a manner that meets the requirements of the Uniform Act.

#### **INDEX**

- I. Market Data Studies and Determination of Average Fair Market Per Acre Value
- II. Notice to Landowners
- III. Appraisal Waiver Valuation Review
- IV. Determining Settlement Offers for Waiver Parcels
- V. Appraisals
- VI. Review Appraisal Process
- VII. Landowner Negotiations
  - A. Landowner Negotiations—Waiver Parcels
  - B. Landowner Negotiations—Appraisal Parcels
- VIII. Relocations
  - IX. Exhibits
    - Exhibit A Land Offer Summary (example)
    - Exhibit B Valuation Memorandum (form)
    - Exhibit C Technical Review Report (example)
    - Exhibit D Technical Review Report (example)
    - Exhibit E Easement Calculation Worksheet (form)

#### I. Market Data Studies and Determination of Average Fair Market Per Acre Value

Market data studies, from a qualified real estate appraisal firm, either have been or will be ordered during the course of the Project to assist in determining the current market value of the land along the proposed route for the Project.

- 1. The Clean Line Entities will engage a real estate appraisal firm with experience in linear infrastructure projects (the "Appraisal Firm") to perform county-wide market data studies, appraisals and other related tasks, all consistent with the standards set forth in the Uniform Standards of Professional Appraisal Practice ("USPAP"). The Clean Line Entities will provide the Appraisal Firm with the list of counties traversed by the Project Area ROW (as defined below). The Appraisal Firm will provide property sales data within such counties to establish fair market value of various land types for the parcels on which the Clean Line Entities would like to pursue easement acquisition (the "Project Area ROW").
- 2. The Appraisal Firm will review and compile all of the relevant recent property sales within the county, for each county traversed by the Project Area ROW.
- 3. In addition to its review and compilation of the sales data for each county, the Appraisal Firm will analyze the sales data to determine property value trends. For example, in Texas County, Oklahoma, the Appraisal Firm determined that land value trends were based not just on the land use, but also on land parcel size.
- 4. Based on the sales data collection and analysis conducted by the Appraisal Firm, the Clean Line Entities and a second qualified party (the "Market Data Analyst") will review all of the data provided and determine the average per acre value for specific land types within each county. The Market Data Analyst will have sufficient understanding of real estate valuation in general, knowledge of the real estate market within the geographic areas where the Project Area ROW is located in particular, and experience with easement acquisition under the Uniform Act. Generally the Market Data Analyst will review only the sales data for the most recent 12 months, unless there is insufficient data for that period, in which case the Market Data Analyst will review the sales data for

-

<sup>&</sup>lt;sup>1</sup> The original market analysis was performed by Contract Land Staff, a national ROW acquisition firm with experience in acquiring right-of-way ("ROW") for linear infrastructure projects throughout the United States and Canada, including several projects that involved ROW acquisition under the Uniform Act. Clean Line may engage Contract Land Staff, or other experienced acquisition firms to manage the ROW acquisition on the Project under the direction of Clean Line's Vice President of Land. Other Market Data Analysts may be used from time to time to provide updated analysis of the market data, provided, however, in all events all such Market Data Analysts must be similarly qualified and will be supervised by the Clean Line Entities.

<sup>&</sup>lt;sup>2</sup> Land types may differ in each county (depending on the terrain and typical uses of land within the county) but some examples of typical land use types encountered are crop, pasture, timber, residential, hobby farm, etc.

the most recent three years to determine the average historical per acre value.<sup>3</sup> Once an average historical per acre value is determined for each land use type in the county, those per acre values will be increased by ten percent (10%). The resulting per acre value will be used as the average fair market value for each land use type within each county (the "Average Fair Market Per Acre Value").

- 5. The Average Fair Market Per Acre Value for each land use type within in each county will be used as part of the process of determining which parcels crossed by the Project Area ROW may qualify as Waiver Parcels, as described in Section III below.
- 6. The Clean Line Entities will analyze, and update if necessary, the sales data and market data analysis from time to time as land values increase or decrease in order to determine if the Average Fair Market Value for each land type should change.

<sup>3</sup> When reviewing sales data for both the previous 12 months and 36 months, the Market Data Analyst will compare the averages of sales within each such period and will use the average value that is higher.

-

#### II. Initial Notice to Landowners

Consistent with § 24.102(b) of the Uniform Act, the Clean Line Entities will notify potentially affected landowners in writing of their interest in acquiring an easement.<sup>4</sup>

- 1. Prior to the initiation of formal negotiations with landowners, the Clean Line Entities will provide notice of their intent to acquire an easement (the "Formal Notice Letter"); such notification will also include a summary of the basic protections provided to the landowner under the Uniform Act (the "Landowner Brochure"). The Formal Notice and Landowner Brochure will either be hand delivered to the landowner by land agents of the ROW acquisition company (the "Land Agent") or employees of the Clean Line Entities or mailed via certified mail, return receipt requested, or mailed via registered mail. Once delivered, a copy of the Formal Notice Letter will be placed in the office file for the landowner tract.
- 2. The Formal Notice Letter will be written in English in plain, understandable language and will include the name and telephone number of a person who may be contacted for answers to questions or if additional assistance is needed.
- 3. As part of each Formal Notice Letter, the Clean Line Entities will offer foreign-speaking landowners and any other landowners requiring special assistance, appropriate resources to enable the landowner to read and understand the Formal Notice Letter, as well as any subsequent communications and proposed easement terms. If a Land Agent determines upon first contact with a landowner that the landowner does not speak or read English, or requires any other form of special assistance, the Land Agent will notify the Clean Line Entities, and the Clean Line Entities will ensure that a trained agent or employee is available to assist the landowner as needed.

\_

<sup>&</sup>lt;sup>4</sup> The Clean Line Entities will also take appropriate steps, consistent with the Uniform Act, to notify and engage with tenants where acquisition of the easement would affect any tenant rights or tenant-owned property.

### III. Appraisal Waiver - Valuation Review

The Uniform Act provides that an appraisal is not required for parcels that: (i) have an anticipated easement acquisition cost of \$10,000 or less, and (ii) for which the valuation analysis is uncomplicated. If both criteria are met, the parcel will be deemed a "Waiver Parcel".

Clean Line will review the parcels within the Project Area ROW to determine if any such parcels meet the criteria for waiving an appraisal ("Waiver Valuation Review"). The Waiver Valuation Review will be performed by the Market Data Analyst or other qualified party designated by the Clean Line Entities (the "Valuation Reviewer"). The Valuation Reviewer will have sufficient understanding of real estate valuation in general, knowledge of the real estate market within the geographic areas where the Project Area ROW is located in particular, and experience with waiver valuation under the Uniform Act. The Waiver Valuation Review process and criteria for designation of a Waiver Parcel is further described below.

- 1. Determination of easement compensation of \$10,000 or less.
  - (a) The Valuation Reviewer will first determine whether contiguous individual tax parcels within the Project Area ROW should be combined or merged into a single parcel for purposes of the Waiver Valuation Review. This process will be performed by utilizing the merge function from the land tracking software (the "Land Database")<sup>5</sup>. The merge function groups multiple contiguous tax parcels owned by the same person (by name and address) within the same county into one larger single tract.
  - (b) Next, the Valuation Reviewer will review aerial imagery maps to determine if multiple tax parcels in common ownership are also in common use (i.e., being farmed as one contiguous parcel). If the Valuation Reviewer has determined that multiple tax parcels are in both common ownership and common use, then the multiple tax parcels will be combined into one parcel for the purpose of evaluating whether such parcel meets the \$10,000 threshold for a Waiver Parcel.
  - (c) Finally, the Valuation Reviewer will multiply the total acreage of the easement sought over the newly combined parcel by the county Average Fair Market Per Acre Value for the parcel's land type to determine if the total fair market value for the easement over such parcel is \$10,000 or less. If the value is \$10,000 or less, then the parcel will have met the first of the two requirements to be treated as a Waiver Parcel.

<sup>&</sup>lt;sup>5</sup> The Land Database is used to track, among many other things, names, addresses, tax parcel numbers, contacts, statuses, activity notes, etc., for each of the parcels within the Project Area ROW.

- 2. Determination if the valuation analysis is uncomplicated. The Valuation Reviewer will consider the following criteria to determine if the valuation analysis is uncomplicated:
  - (a) Is the acquisition of the parcel simple (i.e., a fee purchase vs. an easement purchase)?
  - (b) What are the damages, if any, to the remainder of the landowner's property?
  - (c) Are there any buildings, structures or improvements located in the easement area?
  - (d) Will the acquisition involve any relocation?

If the valuation analysis is determined to be uncomplicated, then the parcel has met the second requirement to be treated as a Waiver Parcel.

- Once the Valuation Reviewer determines that a parcel has met the two requirements for a
  Waiver Parcel, the Valuation Reviewer will indicate that the parcel is qualified as a
  Waiver Parcel on the Land Offer Summary spreadsheet, which is described in further
  detail in Section IV below.
- 4. In accordance with § 24.102(c)(2)(ii)(C) of the Uniform Act, the Clean Line Entities may approve exceeding the \$10,000 threshold for a Waiver Parcel, up to a maximum of \$25,000, provided that the Clean Line Entities offer the landowner the option of having the Clean Line Entities obtain an Appraisal (as defined in Section V below) for the parcel. If a Landowner requests an Appraisal for any parcel where the easement is valued between \$10,000 and \$25,000, then the Clean Line Entities shall obtain an Appraisal.

#### IV. Determining Settlement Offer Amounts for Waiver Parcels

- 1. The Clean Line Entities and the Valuation Reviewer will create a spreadsheet, titled the "Land Offer Summary", for each county that provides the following information for each parcel or the merged/combined parcels:
  - (a) Tract Name
  - (b) Tax ID Numbers
  - (c) Owner Name
  - (d) Width and Length of the Easement Area
  - (e) Total Acres within the Easement Area
  - (f) Average Fair Market Per Acre Value for the applicable land use type (100%—representing the fair market value for fee title)
  - (g) Average Fair Market Per Acre Value for the applicable land use type (75%—representing the fair market value for an easement)
  - (h) Settlement Offer (see below)
  - (i) Notation as to whether the parcel qualifies as a Waiver Parcel
  - (j) Notation as to whether an Appraisal is required

The "Settlement Offer" for each parcel or merged/combined parcel will be derived by multiplying the total acreage of the Easement Area by the applicable Average Fair Market Per Acre Value. Because many parcels are irregular in size and shape (i.e. not a perfect square or rectangle), the total acreage of each parcel shall be calculated by using appropriate software. An example of a Land Offer Summary for Texas County, Oklahoma is attached hereto as Exhibit A.

- 2. The Valuation Reviewer will prepare a memorandum to file for each county (the "<u>Valuation Memorandum</u>") certifying that the Valuation Reviewer has reviewed the Land Offer Summary and all other relevant background data. The Valuation Memorandum will include at a minimum the following documents:
  - (a) A description of the Project as it pertains to the specific county
  - (b) The market data study for the county
  - (c) Land Offer Summary

-

 $<sup>^6</sup>$  The Clean Line Entities have elected to base a Settlement Offer on 100% of fee market value for the subject parcel, even though the acquisition of easement rights is traditionally valued by appraisers between 40 - 90% of fee.

For each Waiver Parcel, the Valuation Reviewer will certify in the Valuation Memorandum that, based on his or her review of the relevant data, that (i) the proposed Settlement Offer represents just compensation for the subject easement, and (ii) an appraisal is unnecessary because the valuation is uncomplicated and the anticipated value of the proposed acquisition is estimated at \$10,000 or less (or if applicable, up to \$25,000), based on a review of available data. See Exhibit B attached hereto for the form of Valuation Memorandum.

- 3. The Vice President of Land for the Clean Line Entities will review the findings as determined by the Valuation Reviewer and will sign the Valuation Memorandum to confirm his or her approval of such findings. Upon approval, a status of "Meets Waiver Requirements" will be entered for each Waiver Parcel in the Land Database.<sup>7</sup>
- 4. An Appraisal will be ordered for any parcel which does not qualify as a Waiver Parcel, as discussed in Section V below.

<sup>&</sup>lt;sup>7</sup> The Land Database is used for the digital record keeping and tracking of parcels within the Project Area ROW. The Clean Line Entities have created a list of "statuses" within the Land Database in order to easily identify or track parcels that may fall within the same type of category (i.e., "Survey Permission Granted", "Meets Waiver Requirements", "Easement Signed", etc.).

#### V. Appraisals

Parcels that do not qualify as a Waiver Parcel will require a full appraisal by a state licensed/certified independent real estate appraiser (the "Appraiser"), qualified to conduct appraisals in accordance with the requirements of the Uniform Act. The Appraiser shall not have an interest, direct or indirect, in the property being evaluated.

- 1. An appraisal will be ordered for each parcel or the merged/combined parcel that does not qualify as a Waiver Parcel (an "Appraisal"). Once an Appraisal is ordered, a status of "Appraisal Ordered" will be entered into the Land Database for the relevant parcel.
- 2. The Clean Line Entities or the Land Agent will contact the landowner to determine if the landowner would like to accompany the Appraiser during the site inspection of the property. The landowner will be given the opportunity to present information and material for consideration by the Appraiser that the landowner believes is relevant to determining the value of the easement property. All such information and material received from the landowner by the Clean Line Entities, the Land Agent or any other contract employees will be provided to the Appraiser for consideration.
- 3. The Appraiser will be informed by an employee or representative of the Clean Line Entities (the "<u>Appraisal Coordinator</u>") as to whether the landowner wants to be present during the Appraiser's site inspection of the property.
- 4. The Appraiser will prepare Appraisals according to and consistent with the requirements of USPAP and relevant state and local requirements. The Appraiser will be provided with the following information before beginning the appraisal process for each parcel or merged/combined parcel:
  - (a) Name, address and phone number(s) of the landowner
    - i. The Appraiser will contact the landowner if it was determined that the landowner wants to be present during the inspection.
    - ii. In the event the landowner cannot be reached via phone or via mail, the Appraiser will contact the Appraisal Coordinator, and the Appraisal Coordinator will communicate with the landowner to determine if the landowner wishes to be present. The Appraisal Coordinator will communicate its findings to the Appraiser in an expeditious manner.
  - (b) Vesting deeds or title report, if available

- (c) Sketch of the proposed easement area
- (d) Sample Easement Agreement
- 5. Upon completion of the Appraisal, an electronic copy of the Appraisal will be delivered to the Clean Line Entities, the Appraisal Coordinator, and the Review Appraiser (as defined in Section VI below) and a status will be entered into the Land Database of "Appraisal Received."

#### VI. Review Appraisal Process

A review appraiser ("<u>Review Appraiser</u>") is an appraiser who examines the reports of other appraisers to ascertain whether their conclusions are consistent with the data reports and with other generally known information about the parcel. The Review Appraiser will review and analyze the relevant facts assembled by the Appraiser using reason, judgment, and a review of supporting documentation and drawings in order to form an opinion or conclusion with respect to the findings contained in the Appraisal.

- 1. The Clean Line Entities will hire a qualified appraiser to act as the Review Appraiser for the Project. The Review Appraiser will (a) be a state-certified real estate appraiser who has past experience and knowledge of appraisals and USPAP guidelines, (b) be familiar with the Project, appraisal reports and the real estate market for the area, and (c) not have any interest, direct or indirect, in the property being evaluated for the easement. The Review Appraiser will do a desk review property inspection of the property covered by the Appraisal.
- 2. At a minimum, in the evaluation of the Appraisal, the Review Appraiser will:
  - (a) Read the Appraisal in its entirety, taking notes on items which may require further evaluation
  - (b) Review the current alignment of the subject parcel and legal description
  - (c) Review and analyze the appraised value in light of comparable sales data used in the analysis
  - (d) Review aerial maps of the property
  - (e) Check calculations in the report for accuracy
  - (f) Evaluate appraisal principle application and techniques
  - (g) Determine if the facts cited in the Appraisal are correct and the approaches and sales data that were used to determine value are reasonable
  - (h) Determine if the Appraiser appropriately applied the tests of highest and best use,
  - (i) Ensure that the Appraisal follows the requirements set forth in the Uniform Act
  - (j) Understand and ensure that any special valuation peculiarities are identified and that they are justified and reasonable
  - (k) Ensure compliance with the Clean Line Entities' policies and requirements
- 3. Upon conclusion of the detailed review, the Review Appraiser will sign a statement certifying that (a) he or she made a thorough and detailed analysis of the Appraisal, (b) he or she either agrees or disagrees with the content and facts, and (c) the Appraisal is in compliance with USPAP and other applicable standards. If the Review Appraiser

requires corrections or revisions, they will be outlined in the Technical Review Report (as defined below). Finally, the Review Appraiser will either accept the contents and comments of the Appraisal or will disapprove the Appraisal. The two possible conclusions of the Review Appraiser are:

- (a) Approval the Appraiser approves the Appraisal as written.
- (b) Disapproval the Appraisal does not meet with the acceptable standards for a specific reason(s) such as content, valuation or other conditions as delineated in a Technical Review Report.
- 4. The Review Appraiser will prepare a "Technical Review Report" and document the validity and findings of the Appraisal. Examples of Technical Review Reports are attached hereto as Exhibit C and Exhibit D.
- 5. A status in the Land Database will be entered of either "Review Appraisal Approved" or "Review Appraisal Denied".
- 6. In the event that the Review Appraiser rejects the Appraisal, either:
  - (a) The Appraisal will be sent back to the original Appraiser for revisions based on the appraisal review and then resubmitted through the review process as outlined above; or
  - (b) A meeting will be held between the Appraiser and Review Appraiser to gather more facts regarding the subject parcel to formalize a joint appraisal analysis.

#### VII. Landowner Negotiations

Landowners will be treated fairly and consistently across the Project when negotiating for easement rights that affect their property.

#### A. Landowner Negotiations—Waiver Parcels

- 1. Employees of the Clean Line Entities or Land Agents<sup>8</sup> will personally contact landowners whenever possible to discuss the Project and how it may impact their property. If any landowner cannot be contacted personally, the Land Agent will deliver the information via First Class Mail (and with respect to the Formal Notice Letter and any final Settlement Offer, via Certified Mail or registered first-class mail—return receipt requested).
- 2. Following delivery of the Formal Notice Letter, as described in Section II above, Land Agents will contact the landowner to provide the following information:
  - (a) The proposed form of Easement Agreement
  - (b) A sketch of the easement area on the landowner's property
  - (c) A Construction Questionnaire, which is a document designed to obtain information about the property, such as land uses, irrigation, utilities, structures, gates and fences, etc. The Clean Line Entities endeavor to obtain this information early in the development process so that it can be taken into consideration during construction planning.
  - (d) A Survey Permission Form that allows the Clean Line Entities to perform surveys (if the landowner had not previously granted the Clean Line Entities survey access rights)
  - (e) A compensation worksheet, which provides (i) a description of the size of the easement on the property, (ii) the Settlement Offer, and (iii) how such Settlement Offer was calculated (the "Easement Calculation Worksheet"). The form of Easement Calculation Worksheet is attached hereto as Exhibit E.
  - (f) A Structure and Damages Calculation Worksheet
  - (g) A copy of the Clean Line Entites' Code of Conduct
  - (h) In Texas, the Landowner Bill of Rights
  - (i) In Oklahoma, a full and complete copy of (i) the Private Rights Settlement Agreement dated January 14, 2011, and (ii) the Oklahoma Corporation Commission's October 28, 2011 order approving Plains and Eastern Clean Line Oklahoma LLC's application to conduct business as a public utility in Oklahoma

-

<sup>&</sup>lt;sup>8</sup> In some instances contact or negotiations with the landowner may be performed by Clean Line employees, rather than Land Agents; as used hereinafter, the term "<u>Land Agent</u>" shall be deemed to include, when applicable, employees of Clean Line.

- 3. When meeting with a landowner, the Land Agent will make every reasonable effort to:

  (a) discuss the Settlement Offer, including explanations as to the basis for the Settlement Offer, (b) explain the Project, (c) explain the Clean Line Entities' policies and procedures (including payment of incidental expenses when applicable), and (d) generally be available to answer any questions or concerns expressed by the landowner. The landowner will be given reasonable opportunity to consider the Settlement Offer and present material which the landowner believes relevant to determining the value of the easement property and to suggest modifications in the proposed terms and conditions of the easement. Land Agents and Clean Line will give full and fair consideration to landowner's comments and suggestions. Land Agents will not use coercive action to induce an agreement on price or terms. Land Agents will exhaust all reasonable negotiations with landowners and will strive to come to voluntary agreement with all landowners.
- 4. In the event that the Clean Line Entities or the Land Agent determines that there is a tenant on the property, the Land Agent will contact the tenant to discuss tenant-related issues and will ensure that the tenant is compensated for crops or other tenant-owned property as required under the Uniform Act.
- 5. When the landowner accepts the Settlement Offer, the landowner and the Clean Line Entities will execute the following documents:
  - (a) Easement Agreement
  - (b) Easement Calculation Worksheet
  - (c) Structure and Damages Calculation Worksheet
- 6. Land Agents will document in the Land Database a summary of all contacts and interactions made with landowners, tenants and other interested parties with respect to each parcel or merged/combined parcel of land within the Project Area ROW.
- 7. Statuses will be entered in the Land Database to track the following information:
  - (a) Date the offer was made to the landowner
  - (b) Amount of the offer
  - (c) Any landowner counter offers
  - (d) Date the Easement Agreement was signed by the landowner
  - (e) Amount of the check written
  - (f) Amount of the balance payment due, if any
  - (g) Date that the balance payment is due, if applicable

#### B. Landowner Negotiations—Appraisal Parcels

- 1. Land Agents will personally contact landowners whenever possible to discuss the Project and how it may impact their property. If any landowner cannot be contacted personally, the Land Agent will deliver the information via First Class Mail (and with respect to the Formal Notice Letter and any final Settlement Offer, via Certified Mail or registered first-class mail—return receipt requested).
- 2. Following delivery of the Formal Notice Letter, as described in Section II above, Land Agents will contact the landowner to provide the following information:
  - (a) The proposed form of Easement Agreement
  - (b) A sketch of the easement area on the landowner's property
  - (c) A Construction Questionnaire, which is a document designed to obtain information about the property, such as land uses, irrigation, utilities, structures, gates and fences, etc. The Clean Line Entites endeavor to obtain this information early in the development process so that it can be taken into consideration during construction planning.
  - (d) A Survey Permission Form that allows the Clean Line Entities to perform surveys (if the landowner had not previously granted the Clean Line Entities survey access rights)
  - (e) A Structure and Damages Calculation Worksheet
  - (f) A copy of Clean Line's Code of Conduct
  - (g) In Texas, the Landowner Bill of Rights
  - (h) In Oklahoma, a full and complete copy of (i) the Private Rights Settlement Agreement dated January 14, 2011, and (ii) the Oklahoma Corporation Commission's October 28, 2011 order approving Plains and Eastern Clean Line Oklahoma LLC's application to conduct business as a public utility in Oklahoma.
- 3. The Land Agent will ask the landowner if they want to be present during any on-site inspections of the property with the Appraiser. The Land Agent will document in the Land Database the requirement of either "Wishes to Accompany the Appraiser" or "Does Not Wish to be Present for Appraisal On-site Inspections". If the landowner does not wish to be present for any on-site inspections, the Land Agent will request that the landowner sign another Survey Permission Form that acknowledges that they have waived this right to accompany the Appraiser.
- 4. In the event the landowner wishes to be present during the Appraisal, the Appraiser will notify the landowner of the date and time of the site inspection. The Appraisal is performed on the property by a state certified/licensed Appraiser (refer to Section V above for more detail).

- 5. When the Appraisal is completed, the Land Agent will meet with the landowner and present the following documents:
  - (a) Copy of the Appraisal
  - (b) The Easement Calculation Worksheet.
- 6. When meeting with a landowner, the Land Agent will make every reasonable effort to: (a) discuss the Settlement Offer, including explanations as to the basis for the Settlement Offer of just compensation, (b) explain the Project, (c) explain the Clean Line Entities' policies and procedures (including payment of incidental expenses when applicable) and (d) generally be available to answer any questions or concerns expressed by the landowner. The landowner will be given reasonable opportunity to consider the Settlement Offer and present material which the landowner believes relevant to determining the value of the easement property and to suggest modifications in the proposed terms and conditions of the easement. Land Agents and the Clean Line Entities will give full and fair consideration to landowner's comments and suggestions. Land Agents will not use coercive action to induce an agreement on price or terms. Land Agents will exhaust all reasonable negotiations with landowners and will strive to come to voluntary agreement with all landowners.
- 7. In the event that the Clean Line Entities or the Land Agent determines that there is a tenant on the property, the Land Agent will contact the tenant to discuss any crops or other tenant-owned property and will ensure that the tenant is compensated for crops or other tenant-owned property as required under the Uniform Act.
- 8. When the landowner accepts the Settlement Offer, the landowner and Clean Line will execute the following documents:
  - (a) Easement Agreement
  - (b) Easement Calculation Worksheet
  - (c) Structure and Damages Calculation Worksheet
- 9. Land Agents will document, in the Land Database, a summary of all contacts and interactions made with landowners, tenants and other interested parties with respect to each parcel or merged/combined parcel of land within the Project Area ROW.
- 10. Statuses will be entered in the Land Database to track the following information:
  - (a) Date the offer was made to the landowner
  - (b) Amount of the offer
  - (c) Any landowner counter offers

- (d) Date the Easement Agreement was signed
- (e) Amount of the check written
- (f) Amount of the balance payment due, if any
- (g) Date that the balance payment is due, if applicable

#### VIII. RELOCATIONS

At this time the Clean Line Entities do not anticipate that any residences or persons will be relocated as a result of the Project. In the event circumstances change and relocation is required, the Clean Line Entities will draft policies and procedures that follow the Uniform Act for this process.

#### IX. EXHIBITS

Exhibit A – Land Offer Summary (example)

Exhibit B – Valuation Memorandum (form)

Exhibit C – Technical Review Report (example)

Exhibit D – Technical Review Report (example)

Exhibit E – Easement Calculation Worksheet- (form)

OK-TE-031.000 OK-TE-023.000 OK-TE-021.000 OK-TE-017.000 OK-TE-013.000 OK-TE-030.00C OK-TE-029.000 OK-TE-028.000 OK-TE-026.000 OK-TE-025.000 OK-TE-019.000 OK-TE-016.000 OK-TE-015.000 OK-TE-014.000 OK-TE-011.000 OK-TE-009.000 OK-TE-006.000 OK-TE-005.000 OK-TE-004.000 OK-TE-003.000 OK-TE-002.000 TractName Actual Width (150ft corridor) 146 150 150 150 150 150 150 150 150 150 150 150 150 150 150 13 150 144 14 150 150 Length 5224 2630 5205 5085 1832 2615 2624 5251 2632 2614 2883 2608 5189 2615 1856 2624 2701 2627 2608 4931 295 **Easement Acres** 9 18 9 18 9 9 9 18 Ь 9 17 18 9 9 9 10 18 9 Ь 9 Fee Price Per Acre \$1,760 \$1,760 \$1,760 \$2,500 \$2,500 \$2,500 \$2,500 \$2,500 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 Settlement Price Per Acre \$1,760 \$2,500 \$1,760 \$1,760 \$2,500 \$2,500 \$2,500 \$2,500 \$2,500 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 \$840 75% Easement \$11,291 \$11,031 \$11,409 \$11,885 \$11,930 \$16,945 \$33,731 \$1,856 \$17,435 \$31,830 \$33,903 \$11,258 \$5,709 \$5,706 \$5,670 \$6,255 \$5,658 \$5,658 \$5,699 \$355 \$365 Offer Settlement Offer \$44,975 \$14,708 \$15,847 \$23,247 \$45,204 \$15,054 \$15,212 \$22,594 \$42,440 \$7,612 \$7,598 \$7,608 \$7,561 \$8,340 \$7,544 \$15,011 \$15,906 \$2,475 \$7,544 \$473 \$487 Qualified for Waiver Yes Yes Yes No No S No S Yes Yes S No 8 S No S S O S S 8 S Appraisal Per CLE Yes S 8 Yes S No S **Date Last Reviewed** for Waivers 4/16/2015

OK-TE Land Offer Summary 11/5/2015

OK-TE Land Offer Summary 11/5/2015

TractName	Actual Width (150ft corridor)	Length	Easement Acres	Fee Price Per Acre	Per Acre	Offer	Offer	Qualified for Waiver	Appraisal Per CLE	Date Last Reviewed for Waivers
OK-TE-032.000	150	2574	9	\$840	\$840	\$5,584	\$7,445	Yes	N <sub>0</sub>	4/16/2015
ОК-ТЕ-033.000	150	2615	9	\$840	\$840	\$5,674	\$7,565	No	Yes	4/16/2015
OK-TE-034.000	150	1318	5	\$840	\$840	\$2,859	\$3,812	No	Yes	4/16/2015
ОК-ТЕ-035.000	150	1318	5	\$840	\$840	\$2,859	\$3,812	No	Yes	4/16/2015
OK-TE-036.000	150	2636	9	\$840	\$840	\$5,718	\$7,624	No	Yes	4/16/2015
ОК-ТЕ-037.000	150	2608	9	\$840	\$840	\$5,659	\$7,545	No	Yes	4/16/2015
OK-TE-038.000	150	2608	9	\$840	\$840	\$5,659	\$7,545	Yes	Yes	4/16/2015
OK-TE-039.000	150	5208	18	\$840	\$840	\$11,298	\$15,064	No	Yes	4/16/2015
OK-TE-040.000	150	2615	9	\$840	\$840	\$5,674	\$7,565	No	Yes	4/16/2015
OK-TE-041.000	150	2615	9	\$840	\$840	\$5,674	\$7,565	No	Yes	4/16/2015
OK-TE-042.000	150	2609	9	\$840	\$840	\$5,661	\$7,548	No	Yes	4/16/2015
OK-TE-043.000	150	2609	9	\$840	\$840	\$5,661	\$7,548	No	Yes	4/16/2015
OK-TE-044.000	150	2634	9	\$840	\$840	\$5,714	\$7,618	Yes	No	4/16/2015
OK-TE-045.000	150	1308	5	\$840	\$840	\$2,837	\$3,782	Yes	Yes	4/16/2015
OK-TE-046.000	150	1308	5	\$840	\$840	\$2,837	\$3,782	Yes	Yes	4/16/2015
OK-TE-047.000	150	1314	5	\$840	\$840	\$2,850	\$3,799	Yes	No	4/16/2015
OK-TE-048.000	150	2627	9	\$840	\$840	\$5,699	\$7,599	Yes	No	4/16/2015
ОК-ТЕ-049.000	150	1314	5	\$840	\$840	\$2,850	\$3,799	Yes	Yes	4/16/2015
OK-TE-050.000	150	2624	9	\$840	\$840	\$5,693	\$7,591	No	Yes	4/16/2015
ОК-ТЕ-051.000	150	2624	9	\$840	\$840	\$5,693	\$7,591	No	Yes	4/16/2015
OK-TE-052.000	150	3929	14	\$840	\$840	\$8,523	\$11,364	N <sub>o</sub>	Yes	4/16/2015

OK-TE-057.000 OK-TE-056.000 OK-TE-055.000 OK-TE-054.000 OK-TE-053.000 TractName Actual Width (150ft corridor) 150 150 150 150 150 Length 2599 1310 2608 3916 1305 **Easement Acres** 5 9 9 13 Fee Price Per \$1,760 Acre \$840 \$840 \$840 \$840 Settlement Price Per Acre \$1,760 \$840 \$840 \$840 \$840 \$11,856 \$5,638 \$2,841 Offer \$8,495 \$2,832 Settlement Offer \$15,808 \$11,327 \$3,788 \$7,517 \$3,776 Qualified for Waiver Yes Yes S No Yes Appraisal Per CLE Yes No Yes No S O Date Last Reviewed for Waivers 4/16/2015 4/16/2015 4/16/2015 4/16/2015 4/16/2015

OK-TE Land Offer Summary 11/5/2015

## Exhibit "B"

## PLAINS & EASTERN

## **VALUATION MEMORANDUM**

State	of		County
	· .		

Waiver Valuation Analysis

Го: Deann Lanz, Vice President of Land	
rom: [Name] [Title]	
Date:, 20	
RE: Compensation Valuation Review of Parcels in [County], [State]	
The Plains and Eastern project is a linear DC electric transmission line that will cross the countemproximately miles. The project will seek 150 - 200 foot wide easements in which construct, operate and maintain the proposed transmission system. A desktop review of aer magery and other available geo-referenced data available by public sources along with the Mark Data Study, prepared by, was utilized in the evaluation of determining and establishing Settlement Offer compensation.	to ial et
Attached to this document are the following documents to establish and document the methodology and logic of the Settlement Offers.	he
Market Data Study of Comparable Sales	
Land Offer Summary	
hereby certify that based on my review of the data, the proposed Settlement Offer for Waiv Parcels set forth in the Land Offer Summary is fair and just compensation and recommend the appraisal be required for such Waiver Parcels.	
Valuation Reviewer:	
By:	
Title:	
Date:	
Approved:	
By: Deann Lanz	
Title: Vice President of Land	
Date:	

# Exhibit "C" TECHNICAL REVIEW REPORT

Tract #:	
Tax ID #:	
Owners of Record:	
I am in receipt of that certain appraisal report dated	(the "Appraisal"), prepared
by of Integra Realty Resources (	the "Appraiser") for the property located
in [Section/Township/Range] (the "Property"), as sub	stantially shown as Exhibit in the
Appraisal. The Appraisal was prepared for and on b	ehalf of the Plains and Eastern Clean Line
LLC ("Clean Line") to utilize and rely on for purp	oses of negotiating with landowners for
easements of DC electric transmission lines.	

#### Appraisal Summary:

- Size of the taking for the easement area
- Highest and best uses and the before and after taking
- Any improvements
- Date of the valuation and the valuation
- Value of the total property or larger parcel and include major items such as timber, improvements and damages

#### Scope of Review:

• I have made a thorough review of the Appraisal and my opinions are based on the materials submitted in the Appraisal, discussions with the Appraiser and discussions with Clean Line (and any other individuals that are pertinent to the review) and my personal knowledge of the local real estate markets. As the Review Appraiser I performed a desk review only of the Appraisal.

#### Property Data Summary:

• Brief description of the size, location of the easement and anything that has influences on the value of the easement. State the current use of the Property and summarize the adequacy of the highest and best use analysis.

#### Area Appraised:

• Define the easement and easement area to be taken

#### Valuation:

• Include approaches to value, last sale of the subject Property, number of sales, factors that influence value, Appraiser's analysis and value opinions.

#### Comments and Recommendations:

• Comments on overall quality of the Appraisal and market support for conclusions. Cite high and low points, if applicable. Recommend/approve the opinion of value, or if appropriate, disapprove or provide a different valuation and your basis of the change.

#### Certification:

• Include a signed certification in compliance with the standards under which the appraisal review report was prepared.

#### Conclusion:

Date

<ul> <li>A short section on what your action</li> </ul>	s were in regards to the Appraisal reviewed.
	-
Review Appraiser	
Appraisal Certification #	-

# EXHIBIT "D" TECHNICAL REVIEW REPORT

1.	IDENTIFICATION:		
	Report reviewed: By	, MAI, CCIM, Ok	dahoma Certified General
	Appraiser No.	and Mr. Oklahon	na Certified General
	Appraiser No.	both appraisers employed with	advisors/
	Real estate and real pro-	perty interest being appraised: a contigu	uous tract owned by
	wit	h 435,144sf or 9.99 ac. gross or 413,36	54sf or 9.49 ac. net being
	appraised at full fee val	ue. The Legal Description is NE/4NE/4	4NE/4 of Sec.
	Township 19 N. R. 1 E.	County, Oklahoma. The proper	rty address is
		OK 74074 and Identified as Parcel#A	-001. The property interest
	appraised is partial fee	value for the utility and temporary ease	ments.
	Effective date of Repor	t: September 12, 2014 signed by	on
	October 3, 2014.	-	

Effective date for review: November 21, 2014.

Intended use and purpose of the review: To express an opinion as to the appropriateness and validity of the appraisers' reports, including their techniques, analysis and conclusions.

#### 2. EXTENT OF THE REVIEW PROCESS:

The reviewer conducted a desk and field review of the appraisal report. The appraisal is being reviewed for its completeness of content, supporting data and analysis to sufficiently support the appraisers' values and conclusions and appropriateness of the techniques used by the appraisers. The report is also being reviewed for its compliance with the Uniform Standards of Professional Appraisal Practice (USPAP) and the Scope of Services provided by the client, which is essentially the same as required by the regulations of the Uniform Relocation Act as embodies in 49CFR 24 and Titles 17 and 69 of the Oklahoma State Statutes regarding valuation for eminent domain.

#### 3. ASSUMPTIONS AND LIMITING CONDITIONS:

The reviewer did not make an independent search of applicable market, cost and income data and assumes that the data provided by the appraiser is a true, accurate and complete representation of the data available for the valuation of the subject property under this review.

The review performed is a desk review. A personal inspection of the subject was not

performed. A comparable sales or independent verification of the cost and market data was not performed. This review will only accept or reconcile the appraisers' final valuation to recommend compensation based on the appraisals.

The reviewer will assume that the title and legal description provided by the appraisers are accurate.

The reviewer assumes that all pictorial images of the subject and the comparable sales are accurate.

The appraisal report is of a partial taking of utility easements rather than in full fee title. Therefore, the reviewer will assume that all aspects of the compensation will be considered as damages except for those items that cannot be relocated or replaced.

## 4. ADEQUACY AND RELEVANCE OF THE DATA AND APPROPRIATENESS OF THE ADJUSTMENTS:

The appraisers have sufficient data with 4 vacant land sales. The sales are all located west of Stillwater with frontage on or near SH-51. The sales therefore, bear a locational similarity to the subject and are relevant to use in the subject's valuation. The sales are moderately inferior and superior to the subject so that the appraisers made only small adjustments to arrive at a value between the extremes of the comparable sales. To bracket the subject with sales that are inferior and superior is an appropriate and relevant technique. The appraisers logically adjusted their high sales downward and the low sales upward. The improvements were valued by using national data source of improvement values that do not usually sell in the open market. Given the age and condition of the buildings, the use of was appropriate. As for the fencing, the reviewer would have preferred a quote from a local fencing contractor, but is adequate. Ideally, the appraiser would have cited the Section and Page used from when valuing the improvements. However, work-up in the addenda section of the report, the given the detail of the use of the source had to be legitimate. While the land data is of sufficient quantity and highly relevant with appropriate adjustments, the improvement data is only adequate in the absence of market and local contractors. Therefore, the market of the appraisers is adequate and appropriately adjusted.

#### 5. COMPLETENESS OF THE REPORTS:

To a sufficient extent, the appraisal report contains a sufficient degree of completeness to meet the summary requirements under USPAP Standard 2-2(b). The appraisal report has comparable sales sheets completed with deed and verification data. The Master Addenda has the locator maps and the comp sale photos so that the report with the master addenda is complete as regards the presentation of the data. The report clearly defines the subject being appraised, the rights to be appraised and the definition of the value to be appraised. The report has a relevant scope of work, a description of the subject and the subject

neighborhood and an analysis and proper conclusion to the highest and best use of the subject. Both appraisers accompanied their presentation of their data with an analysis before arriving at a conclusion of value. The report has sufficient photos of the subject in both the take area and affected improvements. There are sketches of the improvements affected, but not an overall site sketch. Finally there are the required signed certificates and addenda sections that complete the documentation of both reports. Therefore, the report has a sufficient degree of completeness.

#### 6. APPROPRIATENESS OF APPRAISAL TECHNIQUES AND METHODS:

Both appraisers used a conventional and totally appropriate method to value the subject. The sales selected were on the basis of similarity and direct comparison with the subject. In addition, the appraisers included sales that are slightly inferior and superior to the subject to allow for some bracketing of the sales with the subject also. Bracketing is an appropriate technique, especially if very similar sales for the subject cannot be found. In this case the bracketing is in support of the similar sales that the appraiser were able to make a direct comparison. The use of local contractors is preferable to the use of and only for the landscaping/tree was a local contractor used. This method is considered appropriate though; the accuracy of this method diminishes if the quality of the improvements require a large adjustment for depreciation. Given the nature of the improvements that do not sell on the open market, the appraisers had no choice but to use a cost service with a large depreciation factor. It is somewhat surprising that the fencing could not get a local contractor bid. However, it may be possible that no local fencing contractor was available to provide a timely bid within the project time frame. The comparable sales are appropriate by the time frame, location and similarity in features and use to the subject for bracketing or direct comparison. The photos and exhibits also have a sufficient degree of appropriateness, quantity and quality.

#### 7. VALIDITY OF ANALYSIS, OPINIONS AND CONCLUSIONS OF VALUE:

From the above, the reviewer has established that both appraisers have obtained sufficient data and used it appropriately to value the subject. The correct application of appropriate and sufficient data will be reflected in the analysis of the appraisers. The appraisers used a detailed point-by point comparative analysis section supported by a detailed grid showing adjustments to the sales. The appraisers decided on a value towards the upper end of their comparable sales. The basis for this analysis is that the subject has SH-51 frontage with a corner onto Country Club Road. Therefore, the opinion of value logically flows from this analysis and the conclusion of value is valid. The valuation of the improvements is well documented and accepted. The contractor has revised the easement to avoid the shed and residence as well as reduce the area of taking. The appraiser's compensation will be reduced significantly as a result of this revision. Therefore, the recommended value will be set on the following page:

### RECOMMEND VALUE AS ACCEPTED MOSTLY IN THE REPORT:

Indicated Value of Subject:	=\$1	1,306,170
Damages:		
Land; Utility E'smnt, 21,056sf. @ \$3.12/sf.X60%	=\$	39,417
Temporary Easement, 10,756sf.		
@ \$3.12/sf. X 10%	=\$	3,359
Improvements; Replace Metal Gate,	=\$	500
<b>Replace Fence, 785lf.</b> @ \$5.13/lf.	<u>=\$</u>	4,028
<b>Sub-Total Damages:</b>	=\$	47,304
Non- Damages (items acquired)		
Pecan tree	=\$	300
Barn	<u>=\$</u>	5,097
<b>Sub-Total Non-Damages:</b>	=\$	5,397
<b>Total Compensation:</b>	=\$	52,701
Say:	=\$	52,700

The effective date of the appraisal review of the subject property is November 21, 2014.

Review Appraiser Owner,

OREAB#

#### REVIEW APPRAISER'S STANDARD CERTIFICATION

I certify that, to the best of my knowledge and belief:

- \_\_ the facts and data reported by the reviewer and used in the review process are true and correct.
- the analysis, opinions and conclusions in this review report are limited only by the assumptions and limiting conditions stated in this review report and are my personal, impartial and unbiased professional analysis, opinions and conclusions.
- I have no (or the specified) present or prospective interest in the property that is the subject of the work under review and no (or the specified) personal interest with respect to the parties involved
- \_ I have not performed a previous appraisal or review of the subject property.
- \_ I have no bias with respect to the property that is the subject of the work under review or to the parties involved with this assignment.
- \_ my engagement in this assignment was not contingent upon developing or reporting predetermined results.
- \_\_ my compensation is not contingent on an action or event resulting from the analysis, opinions or conclusions in this review or from its use.
- my analyses, opinions and conclusions were developed and this review report was prepared in conformity with the *Uniform Standards of Professional Appraisal Practice*.
- \_ I have not made a personal inspection of the subject property of the work under review.
- \_ no one provided significant appraisal, appraisal review, or consulting assistance to the person signing this certification.

Signed and dated this 21st day of November 2014.

Review Appraiser OREAB

### Exhibit "E"

## Plains and Eastern Clean Line Arkansas LLC EASEMENT CALCULATION SHEET

This Easement Calculation Sheet is made a part of that certain Transmission Line Easement Agreement ("Easement Agreement") between Landowner and Plains and Eastern Clean Line Arkansas LLC ("Plains and Eastern").

				Date
Tract Numl	ber:	-		
Landowner	Name:			
Permanent	Easement	15	0 ft. (+/-)	
Total		ft. (+/-)		
			<del>_</del>	
Land Use F	_			Φ0.00
	0.239000	(+/- acres) X		\$0.00
0	0.000000	(+/- acres) X		\$0.00 " \$0.00
			"Total Easement Consideration	φυ.υυ
The Total I	Easement Consider	ration shall he no	aid as follows:	
<u>The Total E</u>	tasemeni Consider	anon snan ve po	uu as jouows.	
(A) Initial l	Payment (30% of t	he Total Easeme	ent Consideration)	\$0.00
AND				
(R) Ralance	e <b>Due</b> prior to the e	earlier of		
	•		erty to install structures or wires,	
		-	ded pursuant to the Easement	
Agreement	Extension, the "E	asement Compe	nsation Deadline'')	\$0.00
T				
Easement A	Agreement Extensi	<u>on</u>		
	•	•	nded for two additional one-year	
-			ideration due by 12-31-2017 for ne second extension) (" <i>Extension</i>	
	•		credited towards the Balance Due	e
Δ Initial Payme	ent is paid at time of	grant of the Easer	nent Agreement	\$0.00
•		-	ned that the Permanent Easement w	idth is greater or less than 150'
and/or the lin	near footage is great	er or less than as .	shown above, Plains and Eastern sh	all adjust the Balance Due such
	l Easement Consider his Easement Calcul		actual footage and width and calcul	ated using the same formulas as
sei jorin on i	nis Easemeni Caicai	anon sneer.		
A Landowner	acknowledges and a	aroos that Plains a	nd Eastern is under no obligation to	nay the Ralance Due portion of
			s and Eastern fails to do so on or be	
			nent Agreement, the Easement Agree	•
			ial Payment and any Extension Payı er liability to Landowner.	nent (if applicable), and Plains
	J			
Δ Plains and E	Eastern has the right	to extend the Ease	ment Compensation Deadline for tw	o additional one-vear periods by
payment of th	he Extension Paymer	nt to Landowner p	rior to the Easement Compensation	Deadline. All sums paid by
	Eastern for such exter Balance Due.	ision shall be reta	ined by Landowner and are non-refi	ındable, and will not be credited
iowaras ine I	<b>Б</b> ишис <b>Дис.</b>			
Acceptance				
LANDOWNI	ER:			DATE:
				<u> </u>
Plains and Ea	astern:			DATE: